FORM 10-Q

SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

	washington, b.c. 2	20349
[X]	QUARTERLY REPORT PURSUANT TO SECTION : EXCHANGE ACT OF 1934	13 OR 15 (d) OF THE SECURITIES AND
	For the quarterly period ende	ed June 29, 1996
	OR	
[]	TRANSITION REPORT PURSUANT TO SECTION EXCHANGE ACT OF 1934	13 OR 15 (d) OF THE SECURITIES
	For the transition period from	to
	Commission File Number	er 0-1222
	DUCOMMUN INCORPOR	RATED
	(Exact name of registrant as spec	ified in its charter)
	Delaware	95-0693330
	(State or other jurisdiction of incorporation or organization)	I.R.S. Employer Identification No.
	23301 South Wilmington Avenue, Can	rson, California 90745
	(Address of principal executive	offices) (Zip Code)
	(310) 513-7200	9
	(Registrant's telephone number, :	including area code)
	N/A	
(F	ormer name, former address and former fiso report)	
to be	ate by check mark whether the registrant of filed by Section 13 or 15 (d) of the Section 12 months (or for such shorter period	urities Act of 1934 during the

to file such reports), and (2) has been subject to such filing requirements for the past 90 days.

Yes X No

Indicate the number of shares outstanding of each of the issuer's classes of common stock, as of the latest practicable date. As of June 29, 1996, there were outstanding 7,295,237 shares of common stock.

DUCOMMUN INCORPORATED FORM 10-Q INDEX

		Page
Part I. Financial Info	ormation	
Item 1.	Financial Statements	
	Consolidated Balance Sheets at June 29, 1996 and December 31, 1995	3
	Consolidated Statements of Income for Three Months Ended June 29, 1996 and July 1, 1995	4
	Consolidated Statements of Income for Six Months Ended June 29, 1996 and July 1, 1995	5
	Consolidated Statements of Cash Flows for Six Months Ended June 29, 1996 and July 1, 1995	6
	Notes to Consolidated Financial Statements	7 - 14
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	15 - 18
Part II. Other Informat	cion	
Item 4.	Submission of Matters to a Vote of Security Holders	19
Item 6.	Exhibits and Reports on Form 8-K	19
Signatures		20

Item 1. Financial Statements

DUCOMMUN INCORPORATED AND SUBSIDIARIES CONSOLIDATED BALANCE SHEETS (In thousands, except share amounts)

	June 29, 1996	
ASSETS Current Assets:	4 104	
Cash and cash equivalents Accounts receivable (less allowance for doubtful	\$ 104	\$ 371
accounts of \$246 and \$366)	18,889	13,828
Inventories	21, 285	
Deferred income taxes (Note 6)	5,439	
Other current assets	1,448	1,151
Tabal Oursest Assats	47.405	
Total Current Assets Property and Equipment Not	47,165 25,334	33,802
Property and Equipment, Net Deferred Income Taxes (Note 6)	25,334 6,112	23,011 6,451
Excess of Cost Over Net Assets Acquired (Net of Accumulated	0,112	0,431
Amortization of \$2,873 and \$2,323)	17,905	16,697
Other Assets	603	
	\$ 97,119	
	=======	======
LIABILITIES AND SHAREHOLDERS' EQUITY Current Liabilities:		
Current portion of long-term debt (Note 5)	\$ 929	,
Accounts payable	10,403	4,917
Accrued liabilities	13,745	13,728
Total Current Liabilities	25,077	
Long-Term Debt (Note 5)	18,543	
Convertible Subordinated Debentures (Note 5)		
Other Long-Term Liabilities	1,288	633
Total Liabilities	44,908	
Commitments and Contingencies (Notes 7) Shareholders' Equity:		
Common stock		
Common stock \$.01 par value; authorized 12,500,000 shares; issued and outstanding 7,295,237 shares in 1996 and 4,852,281 in 1995	73	49
Additional paid-in capital	59,072	
Accumulated deficit		(10,450)
Total Shareholders' Equity	52,211	24,588
	* 07 110	* 90 07 <i>1</i>
	\$ 97,119 ======	\$ 80,974 ======

DUCOMMUN INCORPORATED AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF INCOME (In thousands, except per share amounts)

	For Three Months Ended		
	June 29, 1996	July 1, 1995	
Net Sales	\$ 28,869	\$ 23,201	
Operating Costs and Expenses: Cost of goods sold Selling, general and administrative expenses	•	15,869 4,997	
Total Operating Costs and Expenses	25,253	20,866	
Operating Income Interest Expense	3,616 (275)	2,335 (988)	
Income Before Taxes Income Tax Expense (Note 6)	3,341 (935)	1,347 (377)	
Net Income	\$ 2,406	\$ 970 ======	
Earnings Per Share: Primary Fully Diluted	\$.35 .31		
Weighted Average Number of Common and Common Equivalent Shares Outstanding for Computation of Earnings Per Share: Primary Fully Diluted	6,922 7,820	4,773 7,627	

DUCOMMUN INCORPORATED AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF INCOME (In thousands, except per share amounts)

	For Six Months Ended	
	June 29, 1996	
Net Sales	\$ 52,661	\$ 43,823
Operating Costs and Expenses: Cost of goods sold Selling, general and administrative expenses	35,038 12,043	9,436
Total Operating Costs and Expenses	47,081	39,752
Operating Income Interest Expense		4,071 (1,869)
Income Before Taxes Income Tax Expense (Note 6)		2,202 (617)
Net Income	\$ 3,516 ======	\$ 1,585 ======
Earnings Per Share: Primary Fully Diluted	\$.55 .51	\$.34 .31
Weighted Average Number of Common and Common Equivalent Shares Outstanding for Computation of Earnings Per Share: Primary Fully Diluted	6,370 7,839	4,735 7,626

DUCOMMUN INCORPORATED AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF CASH FLOWS (In thousands, except per share amounts)

	For Six Months Ended	
	June 29, 1996	July 1, 1995
Cash Flows from Operating Activities: Net Income Adjustments to Reconcile Net Income to Net Cash Provided by Operating Activities:	\$ 3,516	\$ 1,585
Depreciation and amortization Deferred income tax provision Changes in Assets and Liabilities, Net of Effects from Acquisitions:	2,132 655	2,203 365
Accounts receivable Inventories Other assets Accounts payable Accrued and other liabilities	(2,260) (2,892) (281) 4,600 (955)	(1,816) (857) 85 (1,371) 761
Net Cash Provided by Operating Activities	4,515	955
Cash Flows from Investing Activities: Purchase of Property and Equipment Acquisition	(2,783) (8,000)	(1,078) (4,427)
Net Cash Used in Investing Activities	(10,783)	(5,505)
Cash Flows from Financing Activities: Net Borrowings (Repayments) of Long-Term Debt Cash Premium for Conversion of Convertible Subordinated Debentures Other	6,627 (609) (17)	(3,884)
Net Cash Provided by (Used in) Financing Activities	6,001 	(3,892)
Net Decrease in Cash and Cash Equivalents Cash and Cash Equivalents at Beginning of Period	(267) 371	(8,442) 8,483
Cash and Cash Equivalents at End of Period	\$ 104 ======	\$ 41 ======
Supplemental Disclosures of Cash Flow Information:		
Interest Expense Paid Income Taxes Paid	\$ 1,108 \$ 1,018	\$ 1,659 \$ 125

Supplementary Information for Non-Cash Financing Activities:

During the first six months of 1996, the Company issued 2,417,205 new shares of common stock upon conversion of \$24,263,000 of its outstanding 7.75% convertible subordinated debentures.

DUCOMMUN INCORPORATED AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

- Note 1. The consolidated balance sheets, consolidated statements of income and consolidated statements of cash flows are unaudited as of and for the three months and six months ended June 29, 1996 and July 1, 1995. The financial information included in the quarterly report should be read in conjunction with the Company's consolidated financial statements and the related notes thereto included in its annual report to shareholders for the year ended December 31, 1995.
- Note 2. Certain amounts and disclosures included in the consolidated financial statements required management to make estimates which could differ from actual results.
- Note 3. Earnings per common share computations are based on the weighted average number of common and common equivalent shares outstanding in each period. Common equivalent shares represent the number of shares which would be issued assuming the exercise of dilutive stock options, reduced by the number of shares which would be purchased with the proceeds from the exercise of such options. For 1996 and 1995, shares associated with convertible securities have been included in the weighted average number of shares outstanding. The computations of earnings per share are as follows:

DUCOMMUN INCORPORATED AND SUBSIDIARIES COMPUTATION OF EARNINGS PER COMMON AND COMMON EQUIVALENT SHARES (In thousands, except per share amounts)

	For Three Months Ended	
	June 29, 1996	July 1, 1995
Income for Computation of Primary Earnings Per Share Interest, Net of Income Taxes, Relating to 7.75% Convertible	\$2,406	\$ 970
Subordinated Debentures Net Income for Computation of	1	390
Primary Earnings Per Share	2,406	970
Net Income for Computation of Fully Diluted Earnings Per Share	2,407	1,360
Applicable Shares: Weighted Average Common Shares Outstanding for Computation of Primary Earnings Per Share Weighted Average Common Equivalent Shares Arising From:	6,438	4,468
7.75% convertible subordinated debentures	888	2,806
Stock Options: Primary Fully diluted	484 494	305 353
Weighted Average Common and Common Equivalent Shares Outstanding for Computation of Fully Diluted Earnings Per Share	7,820	7,627
Earnings Per Share: Primary Fully diluted	\$.35 .31	\$.20 .18

DUCOMMUN INCORPORATED AND SUBSIDIARIES COMPUTATION OF EARNINGS PER COMMON AND COMMON EQUIVALENT SHARES (In thousands, except per share amounts)

	For Six Months Ended	
	June 29, 1996	July 1, 1995
Income for Computation of Primary Earnings Per Share Interest, Net of Income Taxes, Relating to 7.75% Convertible	\$3,516	\$1,585
Subordinated Debentures Net Income for Computation of	443	779
Primary Earnings Per Share Net Income for Computation of	3,516	1,585
Fully Diluted Earnings Per Share	3,959	2,364
Applicable Shares: Weighted Average Common Shares Outstanding for Computation of Primary Earnings Per Share Weighted Average Common Equivalent Shares Arising From: 7.75% convertible subordinated	5,903	4,467
debentures Stock Options:	1,439	2,806
Primary Fully diluted	467 497	268 353
Weighted Average Common and Common Equivalent Shares Outstanding for Computation of Fully Diluted Earnings Per Share	7,839	7,626
Earnings Per Share: Primary Fully diluted	\$.55 .51	\$.34 .31

Note 4. Acquisition

In January 1995, Ducommun acquired the capital stock of 3dbm, Inc. ("3dbm") for \$4,780,000 in cash (of which \$353,000 was withheld with respect to certain assets and potential liabilities of 3dbm) and \$400,000 in notes. 3dbm supplies high-power expanders, microcells and other wireless communications hardware used in cellular telephone networks, and microwave components and subsystems to both military and commercial customers.

On June 28, 1996, Ducommun acquired substantially all of the assets and assumed certain liabilities of MechTronics of Arizona, Inc. ("MechTronics") for \$8,000,000 in cash and a \$750,000 note. Ducommun may be required to make additional payments for the period June 28, 1996 to December 31, 1996, and each of the calendar years ending December 31, 1997, 1998, and 1999, based on the future financial performance of the business of MechTronics. MechTronics is one of the United States' leading manufacturers of high quality and high reliability mechanical and electromechanical enclosure products for the defense electronics, commercial aviation and communications markets.

The acquisition of MechTronics was accounted for under the purchase method of accounting. However, the consolidated statements of income do not include any operating results for MechTronics, since the acquisition was not completed until June 28, 1996. The cost of the acquisition has been preliminarily allocated on the basis of the estimated fair value of assets acquired and liabilities assumed. This resulted in approximately \$2,203,000 of cost in excess of net assets acquired. Such excess (which will increase for any future contingent payments) is being amortized on a straight line basis over fifteen years.

Note 5. Long-term debt and convertible subordinated debentures are summarized as follows:

	(In thousands)	
	June 29, 1996	December 31, 1995
Bank credit agreement	\$14,500	\$ 8,100
Term and real estate loans	3,789	3,559
Promissory notes related to acquisitions	1,183	1,186
Total debt	19,472	12,845
Less current portion	929	3,910
·		
Total long-term debt	\$18,543	\$ 8,935
· ·	======	======
7.75% Convertible subordinated		
debentures due 2011	\$	\$24,263
	======	======

During the first six months of 1996, the Company issued 2,417,205 new shares of common stock upon conversion of \$24,263,000 of its outstanding 7.75% convertible subordinated debentures. The Company paid cash of \$609,000 for the conversions. The conversions reduced interest expense by approximately \$324,000 in the first six months of 1996.

In May 1996, the Company and its bank amended the Company's credit agreement. The amended credit agreement provides for a \$24,000,000 line of credit at May 16, 1996. The line of credit has an expiration date of July 1, 1998. Interest is payable monthly on the outstanding borrowings based on the bank's prime rate (8.25% at June 29, 1996). A Eurodollar pricing option is also available to the Company for terms of up to six months at the Eurodollar rate plus a spread based on the leverage ratio of the Company calculated at the end of each fiscal quarter (1.50% to 1.75% at June 29, 1996). At June 29, 1996, the Company had \$9,158,000 of unused lines of credit, after deducting \$14,500,000 of loans outstanding and \$342,000 for an outstanding standby letter of credit which supports the estimated post-closure maintenance cost for a former surface impoundment.

Borrowings under the credit agreement are secured by most of the assets of the Company and its subsidiaries. The credit agreement includes minimum leverage and fixed charge coverage ratios, earnings covenants, and limitations on capital expenditures, acquisitions, future dividend payments and outside indebtedness.

The carrying amount of long-term debt approximates fair value based on the terms of the related debt and estimates using interest rates currently available to the Company for debt with similar terms and remaining maturities.

Note 6. Income Taxes

The provision for income tax expense consists of the following:

	(in thousands)		
	June 29, 1996	July 1, 1995	
Current tax expense:			
Federal State	\$ 127 585	\$ 52 200	
	712	252	
Deferred tax expense:			
Federal	715	346	
State	(60)	19	
	655	365	
Income Tax Expense	\$ 1,367 ======	\$ 617 =====	

Deferred tax assets (liabilities) consist of the following:

	(in thousands)		
	June 29, 1996	Dec. 31, 1995	
Federal and state NOLs Credit carryforwards Employment-related reserves Inventory reserves Other	\$ 10,402 1,324 1,348 751 1,956	\$ 11,538 1,197 1,691 748 1,352	
Depreciation	15,781 (2,462)	16,526 (2,552)	
Net deferred tax assets before valuation allowance Deferred tax assets valuation allowance	13,319 (1,768)	13,974 (2,433)	
Net deferred tax asset	\$ 11,551 ======	\$ 11,541 ======	

The decrease in the valuation allowance is due to the Company's reevaluation of the realizability of income tax benefits from future operations related to the acquisition of MechTronics consummated in June 1996. As a result, the carrying value of the net deferred tax benefit was increased by \$665,000 and goodwill associated with the MechTronics acquisition was reduced by the same amount.

(in thousands)

At June 29, 1996, the Company had federal tax NOLs totalling \$27 million which expire in the years 1999 through 2004. SFAS 109 requires that the tax benefit of such NOLs be recorded, measured by enacted tax rates, as an asset to the extent management assesses the utilization of such NOLs to be "more likely than not." Management has determined that the income of the Company will, more likely than not, be sufficient to realize the recorded deferred tax asset prior to the ultimate expiration of the NOLs. Realization of the future tax benefits of NOLs is dependent on the Company's ability to generate sufficient taxable income within the carryforward period. In assessing the likelihood of utilization of existing NOLs, management considered the historical results of operations of its operating subsidiaries, including acquired operations, and the current economic environment in which the Company operates. Management does not consider any material future changes in trends or the relationship between reported pretax income and federal and state taxable income or material asset sales or similar non-routine transactions in assessing the likelihood of realization of the recorded deferred tax asset.

Future levels of pretax income are dependent upon the extent of defense spending and other government budgetary pressures, the level of new aircraft orders by commercial airlines, production rate requirement for the Space Shuttle program, general economic conditions,

interest rates, competitive pressures on sales and margins, price levels and other factors beyond the Company's control.

The ability of the Company to utilize the NOLs could be subject to significant limitation in the event of a "change in ownership" as defined in the Internal Revenue Code. A "change of ownership" could be caused by purchases or sales of the Company's securities owned by persons or groups now or in the future having ownership of 5% or more of the Company's outstanding common stock or issuance by the Company of common stock.

Note 7. Contingencies

Ducommun's subsidiary, Aerochem, Inc. ("Aerochem"), is a major supplier of chemical milling services for the aerospace industry. Aerochem has been directed by California environmental agencies to investigate and take corrective action for groundwater contamination at its El Mirage, California facility. Based upon currently available information, the Company has established a provision for the cost of such investigation and corrective action.

Aerochem has been notified by the United States Environmental Protection Agency that Aerochem and other generators of hazardous waste disposed at the Casmalia Resources Hazardous Waste Facility in California (the "Casmalia Site"), an inactive hazardous waste treatment, storage and disposal facility, may be responsible for certain costs associated with the cleanup and closure of the Casmalia Site. Aerochem contributed less than 1/4 of 1% of the total waste disposed of at the Casmalia Site and many other substantially larger companies and governmental entities are involved at the Casmalia Site. The Company has established a provision, based on currently available information, for Aerochem's share of the estimated cost of cleanup and closure of the Casmalia Site.

In the normal course of business, Ducommun and its subsidiaries are defendants in certain other litigation, claims and inquiries, including matters relating to environmental laws. In addition, the Company makes various commitments and incurs contingent liabilities. While it is not feasible to predict the outcome of these matters, the Company does not presently expect that any sum it may be required to pay in connection with these matters would have a material adverse effect on its consolidated financial position or results of operations.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

FINANCIAL STATEMENT PRESENTATION

The interim financial statements reflect all adjustments, consisting only of normal recurring adjustments, which are, in the opinion of the Company, necessary for a fair presentation of the results for the interim periods presented.

ACQUISITION

On June 28, 1996, Ducommun acquired substantially all of the assets and assumed certain liabilities of MechTronics of Arizona, Inc. ("MechTronics") for \$8,000,000 in cash and \$750,000 in a note. Ducommun may be required to make additional payments for the period June 28, 1996 to December 31, 1996, and each of the calendar years ending December 31, 1997, 1998, and 1999, based on the future financial performance of the business of MechTronics. MechTronics is one of the United States' leading manufacturers of high quality and high reliability mechanical and electromechanical enclosure products for the defense electronics, commercial aviation and communications markets.

This acquisition was funded from internally generated cash, a note payable to the seller and borrowings under the Company's credit agreement with its bank (see Financial Condition for additional information).

RESULTS OF OPERATIONS AND EFFECTS OF INFLATION

Second Quarter 1996 Compared to Second Quarter 1995

Net sales increased 24% to \$28,869,000 in the second quarter of 1996. The increase was primarily due to increased off-load work for aircraft structural components from prime contractors and major subcontractors and increased build rates for certain models of commercial jet aircraft.

The Company had substantial sales to Lockheed Martin, Northrop Grumman, McDonnell Douglas and Boeing. During the second quarter of 1996 and 1995, sales to Lockheed Martin were approximately \$2,329,000 and \$2,298,000, respectively; sales to Northrop Grumman were approximately \$2,490,000 and \$2,221,000, respectively; sales to McDonnell Douglas were approximately \$2,832,000 and \$2,841,000, respectively; and sales to Boeing were approximately \$4,572,000 and \$1,542,000, respectively. The sales to Lockheed Martin are primarily related to the Space Shuttle program. The sales relating to Northrop Grumman, McDonnell Douglas and Boeing are diversified over a number of different commercial and military programs.

Gross profit, as a percentage of sales, was 32.6% for the second quarter of 1996 compared to 31.6% in 1995. This increase was primarily the result of change in sales mix, economies of scale resulting from sales increases and improvements in production efficiencies.

Selling, general and administrative expenses, as a percentage of sales, decreased to 20.1% in 1996, compared to 21.5% of sales for 1995. This decrease in expenses as a percentage of sales was primarily related to higher sales volume offset by increased period costs related to the higher sales volume, \$244,000 of debt conversion expense related to the conversion of \$15,837,000 of convertible subordinated debentures and acquisition related expenses of \$130,000 during the second quarter of 1996.

Interest expense decreased 72% to \$275,000 in 1996 primarily due to the conversion of \$28,000,000 of convertible subordinated debentures and lower debt levels.

The increase in income tax expense was primarily due to the increase in income before taxes. The Company continues to use its federal net operating loss carryforward to offset taxable income for income tax purposes. Cash expended to pay income taxes was \$618,000 in 1996, compared to \$50,000 in 1995. For further discussion relating to the income taxes, see Note 6 to the consolidated financial statements.

Net income for the second quarter of 1996 was \$2,406,000, or \$0.31 per share, compared to \$970,000, or \$0.18 per share, in 1995.

Six Months of 1996 Compared to Six Months of 1995

Net sales increased 20% to \$52,661,000 in the first six months of 1996. The increase was primarily due to increased off-load work for aircraft structural components from prime contractors and major subcontractors and increased build rates for certain models of commercial jet aircraft.

The Company had substantial sales to Lockheed Martin, Northrop Grumman, McDonnell Douglas and Boeing. During the first six months of 1996 and 1995, sales to Lockheed Martin were approximately \$4,792,000 and \$4,546,000, respectively; sales to Northrop Grumman were approximately \$4,368,000 and \$4,613,000, respectively; sales to McDonnell Douglas were approximately \$5,532,000 and \$5,147,000, respectively; and sales to Boeing were approximately \$6,028,000 and \$2,767,000, respectively. The sales to Lockheed Martin are primarily related to the Space Shuttle program. The sales relating to Northrop Grumman, McDonnell Douglas and Boeing are diversified over a number of different commercial and military programs.

At June 29, 1996, backlog believed to be firm was approximately \$117,400,000, including \$24,257,000 for space-related business, compared to \$93,100,000 at July 1, 1995 and \$92,600,000 at December 31, 1995. Approximately \$47,000,000 of the total backlog is expected to be delivered during 1996.

Gross profit, as a percentage of sales, was 33.5% for the first six months of 1996 compared to 30.8% in 1995. This increase was primarily the result of change in sales mix, economies of scale resulting from sales increases and improvements in production efficiencies.

Selling, general and administrative expenses increased to \$12,043,000, or 22.9% of sales in 1996, compared to 21.5% of sales for 1995. The increase in expenses as a percentage of sales was primarily the result of period costs related to higher sales volume and \$811,000 of debt conversion expense related to the conversion of \$24,263,000 of convertible subordinated debentures and acquisition related costs.

Interest expense decreased 63% to \$697,000 in 1996 primarily due to the conversion of \$24,263,000 of convertible subordinated debentures and lower debt levels.

The increase in income tax expense was primarily due to the increase in income before taxes. The Company continues to use its federal net operating loss carryforwards to offset taxable income for income tax purposes. Cash expended to pay income taxes was \$1,018,000 in 1996, compared to \$125,000 in 1995. For further discussion relating to the income taxes, see Note 6 to the consolidated financial statements.

Net income for 1996 was \$3,516,000, or \$0.51 per share, compared to \$1,585,000, or \$0.31 per share, in 1995.

FINANCIAL CONDITION

Liquidity and Capital Resources

Cash flow from operating activities for the six months ended June 29, 1996 was \$4,515,000, of which \$2,783,000 was used to purchase property and equipment. During the first six months of 1996, the Company borrowed \$8,000,000 to fund the acquisition of MechTronics. The Company also issued 2,417,205 new shares of common stock upon conversion of \$24,263,000 of its outstanding 7.75% convertible subordinated debentures.

The Company continues to depend on operating cash flow and the availability of its bank line of credit to provide short-term liquidity. Cash from operations and bank borrowing capacity are expected to provide sufficient liquidity to meet the Company's obligations during 1996.

In May 1996, the Company and its bank amended the Company's credit agreement. The amended credit agreement provides for a \$24,000,000 line of credit at May 16, 1996. The line of credit has an expiration date of July 1, 1998. Interest is payable monthly on the outstanding borrowings based on the bank's prime rate (8.25% at June 29, 1996). A Eurodollar pricing option is also available to the Company for terms of up to six months at the Eurodollar rate plus a spread based on the leverage ratio of the Company calculated at the end of each fiscal quarter (1.50% to 1.75% at June 29, 1996). At June 29, 1996, the Company had \$9,158,000 of unused lines of credit, after deducting \$14,500,000 of loans outstanding and \$342,000 for an outstanding standby letter of credit which supports the estimated post-closure maintenance cost for a former surface impoundment.

The Company spent \$2,783,000 on capital expenditures during the first six months of 1996 and expects to spend less than \$6,000,000 for capital expenditures in 1996.

Ducommun's subsidiary, Aerochem, Inc. ("Aerochem"), is a major supplier of chemical milling services for the aerospace industry. Aerochem has been directed by California environmental agencies to investigate and take corrective action for groundwater contamination at its El Mirage, California facility. Based upon currently available information, the Company has established a provision for the cost of such investigation and corrective action.

Aerochem has been notified by the United States Environmental Protection Agency that Aerochem and other generators of hazardous waste disposed at the Casmalia Resources Hazardous Waste Facility in California (the "Casmalia Site"), an inactive hazardous waste treatment, storage and disposal facility, may be responsible for certain costs associated with the cleanup and closure of the Casmalia Site. Aerochem contributed less than 1/4 of 1% of the total waste disposed of at the Casmalia Site and many other substantially larger companies and governmental entities are involved at the Casmalia Site. The Company has established a provision, based on currently available information, for Aerochem's share of the estimated cost of cleanup and closure of the Casmalia Site.

In the normal course of business, Ducommun and its subsidiaries are defendants in certain other litigation, claims and inquiries, including matters relating to environmental laws. In addition, the Company makes various commitments and incurs contingent liabilities. While it is not feasible to predict the outcome of these matters, the Company does not presently expect that any sum it may be required to pay in connection with these matters would have a material adverse effect on its consolidated financial position or results of operations.

PART II - OTHER INFORMATION

Item 4. Submission of Matters to a Vote of Security Holders

The 1996 annual meeting of the Company was held on May 1, 1996. At the meeting, Robert C. Ducommun and Thomas P. Mullaney were elected as directors of the Company to serve for three-year terms expiring at the annual meeting in 1999. In the election of directors, the shareholder vote was as follows: Robert C. Ducommun, For - 5,249,074, Abstain - 2,740; Thomas P. Mullaney - For - 5,249,074, Abstain - 2,740. The directors whose terms of office continued after the annual meeting are: Norman A. Barkeley, H. Frederick Christie, Kevin S. Moore, Richard J. Pearson and Arthur W. Schmutz.

Item 6. Exhibits and Reports on Form 8-K

- (a) The following exhibits are filed with this report:
 - 10.1 Fourth Amended and Restated Loan Agreement between Ducommun Incorporated as Borrower and Bank of America National Trust and Savings Association as Bank, dated May 16, 1996
 - 10.2 First Amendment to Fourth Amended and Restated Loan Agreement between Ducommun Incorporated as Borrower and Bank of America National Trust and Savings Association as Bank, dated June 27, 1996
 - 27 Financial Data Schedule
- (b) No reports on Form 8-K were filed during the quarter for which this report is filed.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

DUCOMMUN INCORPORATED

(Registrant)

By: /s/ James S. Heiser

James S. Heiser

Vice President, Chief Financial Officer

and General Counsel

(Duly Authorized Officer of the Registrant)

By: /s/ Samuel D. Williams

Samuel D. Williams

Vice President and Controller

(Chief Accounting Officer of the Registrant)

Date: July 23, 1996

FOURTH AMENDED AND RESTATED LOAN AGREEMENT

Dated as of May 16, 1996,

between

Ducommun Incorporated,
a Delaware corporation,

as Borrower,

and

Bank of America National Trust And Savings Association, as Bank.

TABLE OF CONTENTS

		Page
ARTICLE 1. DEF	FINITIONS AND ACCOUNTING TERMS	. 1
1.1 1.2 1.3 1.4 1.5	Defined Terms Use of Defined Terms Accounting Terms Exhibits and Schedules References to "Borrower and/or its Subsidiaries	. 20 . 20 . 20
ARTICLE 2. LOA	ANS AND STANDBY LETTERS OF CREDIT	. 21
2.1 2.2 2.3 2.4 2.5 2.6 2.7 2.8 2.9	Line A Loans. Line B Loans. Procedure for Borrowing. Conversion and Continuation Elections. Standby Letters of Credit. Voluntary Reduction of the Commitments. Automatic Reduction of the Line A Commitment. Automatic Increase of the Line A Commitment. Automatic Termination of the Line B Commitment.	. 21 . 21 . 22 . 24 . 27 . 27
ARTICLE 3. PA	YMENTS AND FEES	. 29
3.11 3.12 3.13	Principal and Interest Facility Fee Commitment Fee and Special Commitment Costs. Standby Letter of Credit Fees and Costs. IBOR Fees and Costs Default Rate/Late Payments Computation of Interest and Fees. Non-Banking Days. Manner and Treatment of Payments. Funding Sources. Failure to Charge Not Subsequent Waiver. Survivability. Direct Debit (Pre-Billing).	. 30 . 31 . 32 . 32 . 35 . 35 . 35 . 36 . 36
ARTICLE 4. REF	PRESENTATIONS AND WARRANTIES	. 38
4.1 4.2 4.3 4.4 4.5 4.6	Existence and Qualification; Power; Compliance With Laws	. 38 . 39 . 39
4.7	Changes Title to and Location of Property Intangible Assets	. 41

		Filing of Financing Statements 4	
		Public Utility Holding Company Act 4	
		Litigation 4	
		Binding Obligations 4	
		No Default 4	
		ERISA Plans 4	.2
	4.15	Regulations G, T, U and X; Investment Company	
		Act 4	
		Disclosure 4	
		Tax Liability 4	
		Fiscal Year 4	
		Employee Matters 4	
		Environmental Matters 4	
		Intercompany Indebtedness 4	
		Promissory Notes 4	
		Projections 4	
	4.24	Teledyne Acquisition 4	-6
ARTICLE !	5. AFF	FIRMATIVE COVENANTS (OTHER THAN INFORMATION	
AND REPOR	RTING	REQUIREMENTS)4	.7
	5.1	Payment of Taxes and Other Potential Liens 4	
	5.2	Preservation of Existence 4	
	5.3	Maintenance of Properties 4	
	5.4	Maintenance of Insurance 4	
	5.5	Compliance With Laws 4	
	5.6	Inspection Rights 4	
	5.7	Keeping of Records and Books of Account 4	8
	5.8	Compliance With Agreements, Duties and	
		Obligations 4	
	5.9	Use of Proceeds 4	
		Collateral and Collateral Documents 4	
		Subsidiary Guaranty 5	
		Bank Audit 5	
		Notice of Location Change 5	
		Interest Rate Protection Agreement 5	
	5.15	Teledyne Acquisition 5	,Θ
ARTICLE (6. NEC	GATIVE COVENANTS 5	,2
	6.1	Disposition of Property 5	
	6.2	Mergers 5	
	6.3	Redemption, Dividends and Distributions 5	
	6.4	ERISA 5	
	6.5	Change in Nature of Business	
	6.6	Indebtedness, Guaranties and Liens 5	
	6.7	Transactions with Affiliates 5	
	6.8	Change in Fiscal Year 5	6
	6.9	Payment or Prepayment of Subordinated	
		Obligations 5	6
	6.10	Intercompany Indebtedness 5	6
	6.11	Leverage Ratio 5	6

4			
7	6.12	Fixed Charge Coverage Ratio	58
		Losses/Earnings	
		Capital Expenditures	
		Acquisitions	
ARTICLE	7. INF	ORMATION AND REPORTING REQUIREMENTS	59
		Financial and Business Information	
		Compliance Certificates	
	7.3	Revisions or Updates to Schedules	63
ADTTOLE	0 001	DITIONS	C 4
AKIICLE	o. CUN	DI110N2	04
	8.1	Initial Loan, Etc	64
	8.2	Any Increasing Loan, Etc	66
		Any Loan	
	0.0	7y 20a	•
		NTS OF DEFAULT AND REMEDIES UPON EVENT OF	
DEFAULT.			67
		Events of Default	
	9.2	Remedies Upon Event of Default	70
ARTICLE	10. MI	SCELLANEOUS	72
	10.1	Cumulative Remedies; No Waiver	72
	10.1	Amendments; Consents	72
	10.2	Costs, Expenses and Taxes	
	10.3	Survival of Representations and Warranties	
	10.4	Notices	
	10.5	Execution of Loan Documents	
	10.7	Binding Effect; Assignment	
	10.7	Lien on Deposits and Property in Possession of	14
	10.0	any Bank	74
	10.9	Indemnity by Borrower	
		Environmental Indemnity And Other Matters	
		Nonliability of the Bank	
		No Third Parties Benefited	
		Confidentiality	
		Further Assurances	
		Integration	
		Governing Law	
		Severability of Provisions	
		Headings	
		Time of the Essence	
		Arbitration	
		Personal Liability of Responsible Officials	
		Prior Agreement Superseded	
		Teledyne Indemnity	

FOURTH AMENDED AND RESTATED LOAN AGREEMENT

Dated as of May 16, 1996

This FOURTH AMENDED AND RESTATED LOAN AGREEMENT ("Agreement") is entered into by and between Ducommun Incorporated, a Delaware corporation, as Borrower, and Bank of America National Trust and Savings Association, as Bank.

This Agreement is an amendment and restatement in full of that certain Third Amended and Restated Loan Agreement, dated as of January 20, 1995, by and among Borrower and Bank, (as heretofore amended, the "Original Loan Agreement"). It is the intention of the parties hereto to effect certain amendments to the Original Loan Agreement as of the Restatement Date as expressly reflected herein. The Original Loan Agreement otherwise continues in full force and effect from the date thereof to the Restatement Date and, at all times on and after the Restatement Date, as restated hereby.

WHEREFORE, in consideration of the mutual covenants and agreements herein contained, the parties hereto covenant and agree as follows:

ARTICLE 1. DEFINITIONS AND ACCOUNTING TERMS

1.1 Defined Terms. As used in this Agreement, the following terms shall have the meanings set forth respectively after each:

"Accrued Amount" shall have the meaning given such term in Section $3.13(\mbox{c})$.

"Acquisition" means any transaction, or any series of related transactions, by which Borrower and/or any of its Subsidiaries directly or indirectly (i) acquires any going business or all or substantially all of the assets of any firm, partnership, joint venture, corporation or division thereof, whether through purchase of assets, merger or otherwise, or (ii) acquires (in one transaction or as the most recent transaction in a series of transactions) control of at least a majority in ordinary voting power of the securities of a corporation which have ordinary voting power for the election of directors, or (iii) acquires control of a 50% or more ownership interest in any partnership or joint venture.

"Accounts" means, as of any date of determination, any right of Borrower or any of its Subsidiaries to

payment for goods sold or leased, or to be sold or leased, or for services rendered or to be rendered, or for amounts due from vendors, no matter how evidenced, including accounts receivable, chattel paper, contract rights, purchase orders, notes, instruments, drafts, acceptances and other terms of obligations and receivables.

"Affiliate" means, as to any Person, any other Person which directly or indirectly controls, or is under common control with, or is controlled by, such Person. As used in this definition, "control" (and its correlative meanings, "controlled by" and "under common control with") shall mean possession, directly or indirectly, of power to direct or cause the direction of management or policies (whether through ownership of securities or partnership or other ownership interests, by contract or otherwise), provided that, in any event, any Person that owns, directly or indirectly, 50% or more of the securities having ordinary voting power for the election of directors or other governing body of a corporation (other than securities having such power only by reason of the happening of a contingency), or 50% or more of the partnership or other ownership interests of any other Person (other than as a limited partner of such other Person), will be deemed to control such corporation or other Person.

"Agreement" means this Amended and Restated Loan Agreement, either as originally executed or as it may from time to time be further supplemented, modified, amended, restated or extended.

"Bank" means Bank of America National Trust and Savings Association.

"Banking Day" means any Monday, Tuesday, Wednesday, Thursday or Friday on which Bank is open for business at its address for notice designated as provided herein.

"Bank's Office" means Bank's address as set forth on the signature pages of this Agreement, or such other address as Bank hereafter may designate by written notice to Borrower.

"Billed Amount" shall have the meaning given such term in Section 3.13(b).

"Borrower" means Ducommun Incorporated, a Delaware corporation.

"Borrowing Date" means any date on which a Loan is made under Section 2.3. $\,$

"Brice" means Brice Manufacturing Company, a California corporation, formerly known as J. Nelson Hoffman Manufacturing, Inc.

"Brice Agreement" means the Stock Purchase and Sale Agreement dated as of December 6, 1994 by and between Borrower and the shareholders of Brice.

"Capital Expenditure" means any expenditure (including any capitalized lease expenditure) that is considered a capital expenditure under generally accepted accounting principles, consistently applied, including, without limitation, any amount that is required to be treated as a capitalized asset pursuant to Financial Accounting Standards Board Statement No. 13.

"Cash" means, when used in connection with any Person, all monetary and non-monetary items belonging to such Person that are treated as cash in accordance with generally accepted accounting principles, consistently applied.

"Cash Collateralize" means to pledge and deposit with or deliver to Bank, for the benefit of Bank, as additional collateral for the Obligations, cash or deposit account balances pursuant to documentation in form and substance satisfactory to Bank. Derivatives of such term shall have corresponding meaning. Borrower hereby grants Bank a security interest in all such cash and deposit account balances. Cash Collateral shall be maintained in blocked, non-interest bearing deposit accounts at Bank.

"Cash Equivalents" means, when used in connection with any Person, such Person's Investments in:

- (a) Government Securities due within one year after the date of the making of the Investment;
- (b) certificates of deposit issued by, bank deposits in, bankers' acceptances of, and repurchase agreements covering Government Securities executed by, any Bank or other bank doing business in and incorporated under the Laws of the United States of America or any state thereof and having on the date of such Investment combined capital, surplus and undivided profits of at least \$500,000,000, in each case due within one year after the date of the Investment; and/or
- (c) readily marketable commercial paper of corporations doing business in and incorporated under

the Laws of the United States of America or any state thereof given on the date of such Investment the highest credit rating by NCO/Moody's Commercial Paper Division of Moody's Investors Service, Inc. or Standard & Poor's Corporation, in each case due within six months after the date of the making of the Investment.

"Cash Flow" means for any fiscal period of Borrower and its Subsidiaries, in each case determined on a consolidated basis in accordance with generally accepted accounting principles, EBITDA for that fiscal period, (i) minus any increase (and plus any decrease) in Working Capital over that fiscal period (excluding Cash and outstanding Revolving Loans), (ii) plus any loss (and minus any gain) on the sale of fixed assets during that fiscal period, and (iii) plus non-cash extraordinary losses, (iv) minus non-cash extraordinary income, (v) minus taxes payable in cash during that fiscal period, (vi) plus interest income earned for that fiscal period, (vii) plus the Net Cash Proceeds of the sale of any fixed asset of Borrower or its Subsidiaries to the extent that such Net Cash Proceeds have not already been paid to the Bank pursuant to the provisions of this Agreement or used to repay purchase money indebtedness (or any indebtedness that represents a refinance of such purchase money indebtedness) related to the fixed asset sold or used to purchase assets that will replace the fixed asset sold.

"CERCLA" means the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended, 42 U.S.C. Section 9601 et seg.

"Certificate of a Responsible Official" means a certificate signed by a Responsible Official of the Person providing the certificate.

"Code" means the Internal Revenue Code of 1986, as amended.

"Collateral" means, collectively, all Property on or in which Bank has a Lien pursuant to this Agreement or any other Loan Document.

"Collateral Documents" means all present and future security agreements (including, without limitation, the Security Agreement), deeds of trust, mortgages, assignments, pledge agreements, financing statements, fixture filings, landlord waivers, consents and other documents granting Liens to Bank, or perfecting, effecting, facilitating, consenting to, providing notice of or otherwise evidencing such Liens.

"Commitments" means the Line A Commitment and the Line B Commitment.

"Contingent Obligation" means, as to any Person, any (a) direct or indirect guarantee of Indebtedness of, or other obligation performable by, any other Person, including any endorsement (other than for collection or deposit in the ordinary course of business), co-making or sale with recourse of the obligations of any other Person or (b) assurance given to an obligee with respect to the performance of an obligation by, or the financial condition of, any other Person, whether direct, indirect or contingent, including any purchase or repurchase agreement covering such obligation or any collateral security therefor, any agreement to provide funds (by means of loans, capital contributions or otherwise) to such other Person, any agreement to support the solvency or level of any balance sheet item to such other Person, or any "keep-well", "take-or-pay", "through put" or other arrangement of whatever nature having the effect of assuring or holding harmless any obligee against loss with respect to any obligation of such other Person. The amount of any Contingent Obligation shall be deemed to be an amount equal to the stated or determinable amount of the related primary obligation (unless the Contingent Obligation is limited by its terms to a lesser amount, in which case to the extent of such amount) or, if not stated or determinable, the maximum reasonably anticipated liability in respect thereof as determined by the Person in good faith.

"3DBM Acquisition" means the Acquisition pursuant to which Borrower acquired all of the issued and outstanding capital stock of 3DBM, Inc., a California corporation.

"3DBM Agreement" means the Stock Purchase and Sale Agreement dated as on January 20, 1995, by and between Borrower and each of Herb Berger, Robert McDonough, James A. Tranovich and Uri Yulzari.

"Default" means any Event of Default and/or any event that, with the giving of notice or passage of time or both, would be an Event of Default.

"Designated Deposit Account" means a deposit account to be maintained by Borrower at the Bank's Office, as from time to time designated by Borrower by written notification to Bank.

"Disposition" means the sale, transfer or other disposition in any single transaction or series of related

transactions of (i) any Subsidiary of Borrower, (ii) all or substantially all of the assets of any Subsidiary of Borrower, (iii) all or substantially all of the assets of a division or comparable business segment of Borrower, or (iv) any other asset, or group of related assets, of Borrower or any of its Subsidiaries that has or have at the date of the Disposition a book value or fair market value of \$1,000,000 or more, other than inventory sold, returned to vendors or otherwise disposed of in the ordinary course of business, and other than any such sale, transfer or other disposition for which the proceeds thereof are reinvested in Borrower or any Subsidiary or the replacement of assets of Borrower or any Subsidiary.

"dollars" or "\$" means United States dollars.

"Due Date" shall have the meaning given such term in Section 3.13(a).

"Dynatech Lease" means the Standard Office Lease dated October 3, 1986 between Malibu Canyon Business Park and Dynatech/UZ, Inc., as amended from time to time with the consent of Bank.

"Earnings Before Interest, Taxes Depreciation and Amortization or "EBITDA"" means, for any fiscal period of Borrower and its Subsidiaries, on a consolidated basis, Net Income for such fiscal period, plus the related Interest Expense for such fiscal period, plus the provisions for taxes accrued in that fiscal period, plus depreciation and amortization for such fiscal period, minus interest income earned for such fiscal period.

"Effective Tangible Net Worth" means, as of any date of determination thereof, the consolidated total assets of Borrower and its Subsidiaries (excluding Intangible Assets) minus Total Senior Liabilities of Borrower and its Subsidiaries, determined in accordance with generally accepted accounting principles, consistently applied.

"Environmental Laws" means any and all Laws, statutes, ordinances, rules, regulations, orders, or determinations of any Governmental Agency pertaining to health or the environment in effect in any and all jurisdictions in which Borrower is or at any time may be doing business, or where the real Property of Borrower is located, including without limitation the Clean Air Act, as amended, CERCLA, the Federal Water Pollution Control Act Amendments, the Occupational Safety and Health Act of 1970, as amended, the Resource Conservation and Recovery Act of 1976, as amended, the Safe Drinking Water Act, as amended, California Health and Safety Code Sections 25100

et seq., California Water Code Section 313000 et seq., and the Toxic Substances Control Act, as amended.

"ERISA" means the Employee Retirement Income Security Act of 1974, and any regulations issued pursuant thereto, as amended or replaced and as in effect from time to time.

"Event of Default" shall have the meaning provided in Section 9.1. $\,$

"Excess Cash Flow" means, for any Fiscal Year of Borrower and its Subsidiaries, in each case determined on a consolidated basis in accordance with generally accepted accounting principles, Cash Flow minus (i) capital expenditures made during that Fiscal Year net of any purchase money financing incurred in connection therewith, minus (ii) scheduled payments for that Fiscal Year in respect of capital leases and scheduled payments under the Dynatech Lease, minus (iii) scheduled principal payments for that Fiscal Year in respect of all indebtedness, excluding payments attributable to the termination of the Line B Commitment, minus (iv) without duplication of the amounts in clause (iii), any decrease in Total Line B Outstandings for that Fiscal Year that does not result in a corresponding increase in Total Line A Outstandings, and any permanent decrease in the Line A Commitment during that Fiscal Year that is accompanied by a payment or prepayment of Line A Loans in the amount of such decrease minus (v) earn-out payments payable for that Fiscal Year in connection with the acquisition of Brice and the 3DBM Acquisition, minus (vi) interest expense for that Fiscal Year, minus (vii) any reduction in Borrower's environmental contingencies liability account for that Fiscal Year resulting from a cash payment, (viii) plus any increase in Borrower's environmental contingencies liability account for that Fiscal Year, (ix) minus scheduled non-compete payments for that Fiscal Year as required by the Brice Agreement, and minus (x) without duplication of the foregoing, Premium Payments paid with respect to Borrower's redemption of any Subordinated Obligation.

"Excluded Equipment" means the equipment described on the attached Schedule 5.10, which is incorporated by this reference, to the extent that, and so long as, either CIT Group/Equipment Financing Inc. or General Electric Credit Corporation holds a first priority security interest therein.

"Fiscal Year" shall have the meaning given such term in Section 4.18. $\,$

"Fixed Charge Coverage Ratio" means, for any fiscal period, the ratio of: $\ensuremath{\mathsf{C}}$

(a) EBITDA for that fiscal period: plus (i) any loss (and minus any gain) on the sale of fixed assets during that fiscal period; minus (ii) any extraordinary non-operating income during that fiscal period; plus (iii) any non-cash extraordinary loss attributable to the write-down in the book value of any fixed asset during that period and any other non-cash extraordinary loss that the Bank determines, in its reasonable discretion, does not or will not have an adverse effect on the cash flow of Borrower and its Subsidiaries while any portion of the Commitment remains outstanding; minus (iv) the provision for taxes accrued in that fiscal period; minus (v) capital expenditures (net of purchase money financing) made during that fiscal period; plus (vi) interest income during that fiscal period; minus (vii) any reduction in Borrower's environmental contingencies liability account during that fiscal period resulting from a cash payment; plus (viii) any increase in Borrower's environmental contingencies liability account during that fiscal period; minus (ix) scheduled payments under the Dynatech Lease; minus (x) earn-out payments earned for that fiscal period in connection with the acquisition of Brice and the 3DBM Acquisition, if such earnout payments are capitalized; minus (xi) without duplication of the foregoing, and only to the extent not expensed in calculating EBITDA, Premium Payments paid with respect to Borrower's redemption of any Subordinated Obligation;

(b) to:

(i) interest charges payable in cash for that fiscal period; plus (ii) scheduled principal payments with respect to indebtedness for that fiscal period; plus (iii) scheduled payments for that fiscal period in respect of capital leases; plus (iv) scheduled non-compete payments for that fiscal period due under the Brice Agreement; plus (v) without duplication of the amounts in clause (ii), \$750,000 for each fiscal quarter in Fiscal Year 1996, \$1,000,000 for each fiscal quarter in Fiscal Year 1997, and \$1,250,000 for each fiscal quarter in Fiscal Year 1998;

provided, however for the fiscal quarters ending December 31, 1996, March 31, 1997 and June 30, 1997, the numerator specified in clause (a) above shall be decreased by any payments made by Borrower in excess of \$10,000,000 to (A) purchase or redeem the

Subordinated Obligations (exclusive of any Premium Payment) and (B) consummate the transaction contemplated by the Teledyne Acquisition Agreement.

"Government Securities" means readily marketable direct obligations of the United States of America or obligations fully guarantied by the United States of America.

"Governmental Agency" means (a) any international, foreign, federal, state, county or municipal government, or political subdivision thereof, (b) any governmental or quasi-governmental agency, authority, board, bureau, commission, department, instrumentality or public body, or (c) any court, administrative tribunal or public utility.

"Grand Cayman Rate" means the interest rate (rounded upward to the nearest 1/16th of one percent) at which Bank's Grand Cayman Branch, Grand Cayman, British West Indies, would offer U.S. dollar deposits for the applicable interest period to other major banks in the offshore dollar inter-bank market.

"Guaranty" means the Continuing Guaranty dated as of January 20, 1995, executed and delivered by the Subsidiaries of Borrower to Bank, either as originally executed or as it may from time to time be supplemented, modified, amended, restated or extended.

"IBOR Lending Office" means the office, branch or affiliate identified on the signature page hereof as Bank's IBOR Lending Office or such other office, branch or affiliate as Bank may hereafter designate as its IBOR Lending Office by notice to Borrower.

"IBOR Loan" means a Loan that bears interest determined in relation to the IBOR Rate. $\,$

"IBOR Market" means the Grand Cayman, British West Indies market established by and among banks for the solicitation, offer and acceptance of dollar deposits in such banks.

"IBOR Market Day" means any Banking Day on which commercial banks are open for international business (including dealing in dollar deposits) in Grand Cayman, British West Islands.

"IBOR Rate" means the interest rate determined by the following formula, rounded upward to the nearest 1/100 of one percent (all amounts in the calculation to be

"IBOR Rate Spread" means, with respect to any Line B Loan, 2.25%, and with respect to any Revolving Loan, 1.50% for the period from the Restated Closing Date through August 31, 1996, and at all times thereafter, the IBOR Rate Spread shall be equal to the percentage set forth below opposite Borrower's Leverage Ratio as of the last day of the fiscal quarter most recently ended for the related Spread Period:

Percentage	Leverage Ratio
1.50%	Less than 1.21 to 1.00
1.75%	Equal to or greater than 1.21 to 1.00 but less than 1.51 to 1.00
2.00%	Equal to or greater than 1.51 to 1.00 but less than 1.81 to 1.00
2.25%	Equal to or greater than 1.81 to 1.00

"Indebtedness" means, as to any Person, (a) all indebtedness of such Person for borrowed money, (b) that portion of the obligations of such Person under capital leases which is properly recorded as a liability on a balance sheet of that Person prepared in accordance with generally accepted accounting principles, (c) any obligation of such Person that is evidenced by a promissory note or other instrument representing an extension of credit to such Person, whether or not for borrowed money, (d) any obligation of such Person for the deferred purchase price of Property or services (other than trade or other accounts payable in the ordinary course of business in accordance with customary industry terms), (e) any obligation of such Person that is secured by a Lien on assets of such Person, whether or not such obligation is non-recourse to the credit of such Person, but only to the extent of the fair market value of the assets so subject to the Lien, (f) obligations of such Person arising under acceptance facilities or under facilities for the discount of accounts receivable of such Person, (g) any direct or contingent obligations of such Person under Letters of Credit issued for the account of such Person and (h) any obligations of such Person under a

swap agreement or similar interest rate protection arrangement.

"Intangible Assets" means assets that are considered intangible assets under generally accepting accounting principles, consistently applied, including, without limitation, goodwill, organization expense, patents, trademarks, trade names, copyrights and other intangible Property.

"Interest Expense" means, with respect to any fiscal period of Borrower and its Subsidiaries, on a consolidated basis, the aggregate amount of interest, fees, charges and related expenses paid or payable to a lender in connection with borrowed money that is treated as interest (other than accretion of original issue discount on long-term debt) in accordance with generally accepted accounting principles, consistently applied, and that is charged against Net Income for that fiscal period.

"Interest Period" means, as to any IBOR Loan, the period commencing on the Borrowing Date of such Loan or on the Redesignation Date on which the Loan is converted into or continued as an IBOR Loan, and ending on the date one, two, three or six months thereafter as selected by Borrower in its Request for Loan or Request for Redesignation; provided that:

- (i) if any Interest Period would otherwise end on a day that is not a Banking Day, that Interest Period shall be extended to the following Banking Day unless the result of such extension would be to carry such Interest Period into another calendar month, in which event such Interest Period shall end on the preceding Banking Day;
- (ii) any Interest Period that begins on the last Banking Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period) shall end on the last Banking Day of the calendar month at the end of such Interest Period;
- (iii) no Interest Period for any Line A Loan shall extend beyond the date of any reduction in the Line A Commitment required by Section 2.8, unless the aggregate outstanding principal amount of all Line A Loans which are IBOR Loans having Interest Periods ending after that date, plus the Outstanding Standby Letters of Credit (to the extent that the expiration

date thereof is after that date) is not greater than the amount of the Line A Commitment (after giving effect to any reductions thereto scheduled to be made on that date); and

(iv) no Interest Period for any Line A Loan shall extend beyond the Line A Termination Date and no Interest Period for any Line B Loan shall extend beyond the Line B Termination Date.

"Investment" means, when used in connection with any Person, any investment by or of that Person, whether by means of purchase or other acquisition of stock or other securities or by means of loan, advance, capital contribution, guaranty or other debt or equity participation or interest in any other Person, or otherwise, and includes, without limitation, any partnership and joint venture interests of such Person.

"Laws" means, collectively, all international, foreign, federal, state and local statutes, treaties, rules, regulations, ordinances, codes and administrative or judicial precedents.

"Letter of Credit Obligations" means at any time the sum of (a) the Outstanding Standby Letter of Credit, plus (b) the amount of all unreimbursed drawings under all Standby Letters of Credit.

"Leverage Ratio" means, for any fiscal period, the rate of Total Senior Liabilities for such period to Effective Tangible Net Worth for such period, in each instance of Borrower and its Subsidiaries on a consolidated basis.

"Lien" means any mortgage, deed of trust, pledge, hypothecation, security interest, encumbrance, lien or charge of any kind, whether voluntarily incurred or arising by operation of Law or otherwise, affecting any Property, including any agreement to give any of the foregoing, any conditional sale or other title retention agreement, any lease in the nature thereof, and/or the filing of or agreement to give any financing statement under the Uniform Commercial Code or comparable Law of any jurisdiction.

"Line A Commitment" means \$14,000,000, subject to any adjustment thereto pursuant to Sections 2.6, 2.7 and 2.8.

"Line B Commitment" means \$10,000,000, subject to any adjustment thereto pursuant to Sections 2.6 and 2.9.

"Line B Loan" means any Loan made by Bank under the Line B Commitment. $\ensuremath{\mathsf{E}}$

"Line B Note" means a promissory note made by Borrower to the order of Bank in an original principal amount equal to the amount of the Line B Commitment (and any promissory note that may be issued in substitution, renewal, extension, replacement or exchange therefor), substantially in the form of Exhibit A, with all blanks properly completed, either as originally executed or as the same may from time to time be supplemented, modified, amended, renewed, extended or refinanced.

"Line B Termination Date" means August 31, 1996.

"Loan" or "Loans" means the Loans to be made by Bank to Borrower pursuant to this Agreement, including the Revolving Loans and the Line B Loans.

"Loan Documents" means, collectively, this Agreement, the Notes, the Security Agreement, the Guaranty, the Standby Letters of Credit, the Collateral Documents and any other certificates, documents or agreements of any type or nature heretofore or hereafter executed or delivered by Borrower and/or any one or more of its Subsidiaries or Affiliates to Bank in any way relating to or in furtherance of this Agreement, in each case either as originally executed or as the same may from time to time be supplemented, modified, amended, restated or extended.

"Maximum Line A Loan Amount" means, as of any date of determination thereof, the Line A Commitment minus the Outstanding Standby Letters of Credit.

"Multiemployer Plan" means any employee benefit plan of the type described in Section 4001(a)(3) of ERISA.

"Net Cash Proceeds" means, with respect to any sale or Disposition, the gross Cash consideration (a) paid to or for the account of Borrower or one of its Subsidiaries by reason of that sale or Disposition, (b) paid as a principal installment on or to retire the principal of any evidence of indebtedness which forms all or a portion of the consideration paid by reason of that sale or Disposition, or (c) paid in consideration of the sale or assignment by Borrower or one of its Subsidiaries of any Property other than Cash forming all or a portion of the consideration paid by reason of that sale or Disposition minus (i) any amount paid or payable by the transferor to retire existing obligations of or relating to the Property sold or to retire existing Liens or Rights of Others on

the property sold to the extent that the transferor is contractually obligated to do so, and (ii) the reasonable out-of-pocket expenses of the transferor incident to that sale or Disposition, including brokerage commissions, investment banking fees, attorneys' fees and other similar expenses.

"Net Income" means, with respect to any fiscal period of Borrower and its Subsidiaries, on a consolidated basis, the consolidated net income of Borrower and its Subsidiaries for that period, determined in accordance with generally accepted accounting principles, consistently applied.

"Note" means the Revolving Line Note and/or the Line B Note.

"Obligations" means all present and/or future obligations of every kind or nature of Borrower or any Party at any time and/or from time to time owed to Bank under any one or more of the Loan Documents, whether due or to become due, matured or unmatured, liquidated or unliquidated, or contingent or noncontingent, including obligations of performance as well as obligations of payment, and including interest that accrues after the commencement of any bankruptcy or insolvency proceeding by or against Borrower or any Subsidiary or Affiliate of Borrower.

"Outstanding Standby Letters of Credit" means, as of any date of determination thereof, the aggregate face amount of all Standby Letters of Credit outstanding on such date minus the aggregate amount, if any, paid in Cash by Bank under such Standby Letters of Credit that has been reimbursed by Borrower.

"Party" means any Person (including Borrower and/or any Subsidiaries or Affiliates of Borrower), other than Bank, which now or hereafter is a party to any of the Loan Documents.

"PBGC" means the Pension Benefit Guaranty Corporation or any successor thereof established under ERISA.

"Person" means any entity, whether an individual, trustee, corporation, general partnership, limited partnership, joint stock company, trust, unincorporated organization, bank, business association, firm, joint venture, Governmental Agency, or otherwise.

"Plan" means any employee benefit plan subject to ERISA and maintained by Borrower and/or any Subsidiary of

Borrower or to which Borrower and/or any Subsidiary of Borrower is required to contribute on behalf of its employees.

"Premium Payments" means the added consideration paid by Borrower to the holder of a Subordinated Obligation to induce such holder to convert such Subordinated Obligation to equity.

"Property" means any interest in any kind of property or asset, whether real, personal or mixed, or tangible or intangible.

"RCRA" means the Resources Conservation and Recovery Act, 42 U.S.C. Section 6901 et seq. and the regulations thereto, 40 CFR Part 261.

"Redesignation Date" means any date on which, under Section 2.4, Borrower (a) converts Loans of one Type to another Type, or (b) continues as Loans of the same Type, but with a new Interest Period, Loans having Interest Periods expiring on such date.

"Reference Rate" means, for any day, the rate of interest in effect for such day as publicly announced from time to time by Bank in San Francisco, California, as its "reference rate." (The "reference rate" is a rate set by Bank based upon various factors including Bank's costs and desired return, general economic conditions and other factors, and is used as a reference point for pricing some loans, which may be priced at, above, or below such announced rate.) Any change in the reference rate announced by Bank shall take effect at the opening of business on the day specified in the public announcement of such change.

"Reference Rate Loan" means a Loan that bears interest based on the Reference Rate. $\,$

"Reference Rate Spread" means, with respect to any Line B Loan, 0.75%, and with respect to any Revolving Loan, 0%.

"Regulation D" means Regulation D, as at any time amended, of the Board of Governors of the Federal Reserve System, or any other regulation in substance substituted therefor.

"Regulatory Development" means (a) any change in the Laws, (b) change in the application of any existing Laws or the interpretation thereof by any Governmental Agency or central bank or comparable authority (whether or not

having the force of Law), or (c) compliance by any Bank with any request or directive (whether or not having the force of Law) of any Governmental Agency or central bank or comparable authority.

"Request for Loan" means a written request for a Revolving Loan substantially in the form of Exhibit C, signed by a Responsible Official of Borrower and properly completed to provide all information required to be included therein.

"Request for Standby Letter of Credit" means a written request for the issuance of a Standby Letter of Credit substantially in the form of Exhibit E, signed by a Responsible Official of Borrower and properly completed to provide all information required to be included therein.

"Reserve Percentage" means the total of the maximum reserve percentages for determining the reserves to be maintained by member banks of the Federal Reserve System for Eurocurrency Liabilities, as defined in the Federal Reserve Board Regulation D, rounded upward to the nearest 1/100th of one percent. The percentage will be expressed as a decimal, and will include, but not be limited to, marginal, emergency, supplemental, special and other reserve percentages.

"Responsible Official" means:

- (a) When used with reference to any Person, other than an individual, any corporate officer of such Person, general partner of such Person, corporate officer of a corporate general partner of such Person, or corporate officer of a corporate general partner of a partnership that is a general partner of such Person, or any other responsible official thereof duly acting on behalf thereof; and
- (b) When used with reference to a Person who is an individual, such Person.

Except as otherwise specifically provided herein, any requirement that any document or certificate be signed or executed by any Person requires that such document or certificate be signed or executed by a Responsible Official of such Person, and that the Responsible Official signing or executing such document or certificate on behalf of such Person shall be authorized to do so by all necessary corporate, partnership and/or other action.

"Restated Closing Date" means the time and Banking Day on which the conditions set forth in Section 8.1 are satisfied or waived pursuant to Section 10.2.

"Revolver Termination Date" means July 1, 1998.

"Revolving Line Note" means a promissory note made by Borrower to the order of Bank evidencing the obligation of Borrower to repay Revolving Loans made to it by Bank (and any promissory note that may be issued in substitution, renewal, extension, replacement or exchange therefor), substantially in the form of Exhibit B, with all blanks properly filled in, and executed by Borrower in favor of Bank, either as originally executed or as the same may from time to time be supplemented, modified, amended, renewed, extended or refinanced.

"Revolving Loan" means any Loan made by Bank under the Line A Commitment.

"Right of Others" means, as to any Property in which a Person has an interest, any legal or equitable claim, right, title or other interest (other than a Lien) in or with respect to that Property held by any other Person, and any option or right held by any other Person to acquire any such claim, right, title or other interest, including any option or right to acquire a Lien.

"Security Agreement" means the Amended and Restated Security Agreement dated as of January 20, 1995, executed and delivered by Borrower and its Subsidiaries to Bank, either as originally executed or as it may from time to time be supplemented, modified, amended, restated or extended.

"Spread Period" means, with respect to the last day of any fiscal quarter, the period consisting of the three months that commences on the date that is forty-five (45) calendar days after the last day of such fiscal quarter.

"Standby Letter of Credit" means any standby letter of credit issued by Bank pursuant to Section 2.5, in the standard form for standby letters of credit of Bank, either as originally issued or as the same may from time to time be supplemented, modified, amended, renewed or extended.

"Subordinated Obligations" means (i) the 7-3/4% Convertible Subordinated Debentures due 2011 of Borrower, and (ii) such other indebtedness of Borrower as is fully subordinated to the Obligations pursuant to written provisions approved in writing by Bank.

"Subsidiary" means, as of any date of determination thereof and with respect to any Person, any corporation, partnership or joint venture, whether now existing or hereafter organized or acquired: (a) in the case of a corporation, of which a majority of the securities having ordinary voting power for the election of directors or other governing body (other than securities having such power only by reason of the happening of a contingency) are at the time owned by such Person and/or one or more Subsidiaries of such Person, or (b) in the case of a partnership or joint venture, of which such Person or a Subsidiary of such Person is a general partner or joint venturer or of which a majority of the partnership or other ownership interests are at the time owned by such Person and/or one or more of its Subsidiaries.

"Surety Instruments" means all letters of credit (including standby and commercial), banker's acceptances, bank guaranties, shipside bonds, surety bonds and similar instruments.

"Teledyne" means Teledyne Industries, Inc.

"Teledyne Acquisition" means the purchase by Jay-El Products, Inc. of the microwave switch product line of Teledyne.

"Teledyne Acquisition Agreement" means the purchase agreement pursuant to which the Teledyne Acquisition will be consummated.

"Teledyne Acquisition Price" means the total value of the consideration paid by Borrower and its Subsidiaries in connection with the Teledyne Acquisition.

"to the best knowledge of" means, when modifying a representation, warranty or other statement of any Person, that the fact or situation described therein is known by the Person (or, in the case of a Person other than a natural Person, known by a Responsible Official of that Person) making the representation, warranty or other statement, or with the exercise of reasonable due diligence under the circumstances (in accordance with the standard of what a reasonable Person in similar circumstances would have done and consistent with Borrower's normal business practices) should have been known by the Person (or, in the case of a Person other than a natural Person, should have been known by a Responsible Official of that Person).

"Total Liabilities" means, as of any date of determination thereof, with respect to Borrower and its Subsidiaries, on a consolidated basis, the sum of (a) all liabilities that should be reflected as a liability in a consolidated balance sheet of Borrower and its Subsidiaries on such date prepared in accordance with generally accepted accounting principles, consistently applied, plus (b) the amount of obligations of other Persons (other than Borrower or any Subsidiary of Borrower) that are guaran-tied by Borrower or any Subsidiary of Borrower or are the subject of any agreement by Borrower or any Subsidiary of Borrower in the nature of a guaranty, plus (c) the aggregate face amount of all outstanding Standby Letters of Credit and other letters of credit issued at the request of Borrower or any of Borrower's Subsidiaries; provided, however, that any amount described in clauses (b) and (c) shall be added only to the extent that the guaranty, agreement in the nature of a guaranty, or Standby Letter of Credit covers liabilities that would not be reflected in a consolidated balance sheet of Borrower and its Subsidiaries on such date.

"Total Line A Outstandings" means, as of any date of determination thereof, the sum of (a) all outstanding Revolving Loans evidenced by the Revolving Note on that date, and (b) Outstanding Standby Letters of Credit.

"Total Line B Outstandings" means, as of any date of determination thereof, all outstanding Loans evidenced by the Line B Note on that date.

"Total Outstandings" means, as of any date of determination thereof, the sum of (a) Total Line A Outstandings, and (b) Total Line B Outstandings.

"Total Senior Liabilities" means, as of any date of determination, Total Liabilities as of that date minus Subordinated Obligations as of that date.

"Type" means whether a Loan is an IBOR Loan or a Reference Rate Loan. $\ensuremath{\mathsf{Loan}}$

"Working Capital" means, for any fiscal period of the Borrower and its Subsidiaries, determined on a consolidated basis in accordance with generally accepted accounting principles, current assets (excluding cash and cash equivalents) minus current liabilities (excluding principal payments for that fiscal period in respect to all indebtedness, scheduled payments for that fiscal period in respect of capital leases and scheduled payments under the Dynatech Lease, Revolving Loans, and regularly

scheduled non-compete payments due under the Brice Agreement and the 3DBM Agreement).

- 1.2 Use of Defined Terms. Any defined term used in the plural shall refer to all members of the relevant class, and any defined term used in the singular shall refer to any one or more of the members of the relevant class.
- 1.3 Accounting Terms. All accounting terms not specifically defined in this Agreement shall be construed in conformity with, and all financial data required to be submitted by this Agreement shall be prepared in conformity with, generally accepted accounting principles applied on a consistent basis, as in effect on the date hereof, except as otherwise specifically prescribed herein. In the event that generally accepted accounting principles change during the term of this Agreement such that the financial covenants contained herein would then be calculated in a different manner or with different components, Borrower and Bank agree to amend this Agreement in such respects as are necessary to conform those covenants as criteria for evaluating Borrower's financial condition, or limiting the aggregate Loans hereunder, to substantially the same criteria as were effective prior to such change in generally accepted accounting principles. Without limiting the foregoing, and unless otherwise expressly stated to the contrary, all financial covenants set forth in this Agreement and all calculations required thereby shall be applied and calculated with respect to the performance of Borrower and its Subsidiaries on a consolidated basis.
- 1.4 Exhibits and Schedules. All exhibits and schedules to this Agreement, either as originally existing or as the same may from time to time be supplemented, modified or amended, are incorporated herein by this reference. A matter disclosed on any Schedule shall be deemed disclosed on all Schedules.
- 1.5 References to "Borrower and/or its Subsidiaries". Any reference herein to "Borrower and/or its Subsidiaries" or the like shall refer solely to Borrower during such times, if any, as Borrower shall have no Subsidiaries.

ARTICLE 2. LOANS AND STANDBY LETTERS OF CREDIT

- 2.1 Line A Loans. Subject to the terms and conditions set forth in this Agreement, at any time and from time to time from the Restated Closing Date through the Banking Day immediately preceding the Revolver Termination Date, Bank shall make Revolving Loans to Borrower in such amounts as Borrower may request that do not exceed in the aggregate at any one time outstanding the Line A Commitment; provided that except to the extent otherwise provided in Section 2.5(d), Bank shall not be obligated to make a Revolving Loan if, after giving effect to such Revolving Loan, Total Line A Outstandings would exceed the Line A Commitment. Subject to the limitations set forth herein, Borrower may borrow, repay and reborrow under the Line A Commitment without penalty or premium.
- 2.2 Line B Loans. Subject to the terms and conditions set forth in this Agreement, at any time and from time to time from the Restated Closing Date through the Banking Day immediately preceding the Line B Termination Date, Bank shall make Loans to Borrower in such amounts as Borrower may request that do not exceed in the aggregate at any one time outstanding the Line B Commitment. Once borrowed and repaid, a Line B Loan may not be reborrowed.
- 2.3 Procedure for Borrowing. (a) Each Loan shall be made upon Borrower's irrevocable written notice delivered to Bank in the form of a Request for Loan (which notice must be received by Bank prior to 11:00 a.m. California time) (i) three Banking Days prior to the requested Borrowing Date, in the case of IBOR Loans, and (ii) one Banking Day prior to the requested Borrowing Date, in the case of Reference Rate Loans, in each case specifying:
 - A. the amount of the requested Loan (with respect to any IBOR Loan, each of which shall be in an integral multiple of \$100,000 and not less than \$500,000);
 - B. the Commitment under which the Loan will be made.
 - $\hbox{\bf C. the requested Borrowing Date, which shall be a } \\ {\bf Banking Day;}$
 - D. the Type of Loan requested; and
 - E. the duration of the Interest Period applicable to such Loan included in such notice. If the Request for Loan fails to specify the duration of the $\,$

Interest Period for any Borrowing comprised of IBOR Loans, such Interest Period shall be one month.

- (a) Unless Bank has notified, in its sole and absolute discretion, Borrower to the contrary, a Loan may be requested by telephone, telecopier or telex by Responsible Official of Borrower, in which case Borrower shall confirm such request by promptly mailing a Request for Loan conforming to the above requirements to Bank.
- (b) Upon fulfillment of the applicable conditions set forth in Article 8, the proceeds of a Loan shall be credited in immediately available funds to the Designated Deposit Account.
- (c) The Revolving Loans made by Bank may be evidenced by notations on the Revolving Line Note or by other similar written records. The Line B Loans made by Bank may be evidenced by notations on the Line B Note or by other similar written records.
- (d) A Request for Loan shall be irrevocable upon receipt by ${\sf Bank}\,.$
- (e) Except as otherwise provided in Sections 2.5(d), the amount of each Revolving Loan may not be more than the then applicable Maximum Line A Loan Amount less the aggregate amount of all Revolving Loans previously made and outstanding. The amount of each Line B Loan may not be more than the then applicable Line B Commitment less the aggregate amount of all Line B Loans previously made and outstanding.
- (f) After giving effect to any Loan, there may not be more than ten different Interest Periods in effect with the respect to IBOR Loans.
- 2.4 Conversion and Continuation Elections. (a) Borrower may, upon irrevocable written notice to Bank in accordance with subsection 2.4(b):
 - (i) elect, as of any Banking Day, in the case of Reference Rate Loans, or as of the last day of the applicable Interest Period, in the case of any other Type of Loan, to convert any such Loan (or any part thereof which is an integral multiple of \$100,000 and which is not less than \$500,000) into a Loan of another Type; or
 - (ii) elect, as of the last day of the applicable Interest Period, to continue any Loan having an Interest Period expiring on such day (or any part thereof which is

in an integral multiple of \$100,000 and which is not less than \$500,000);

provided, that if at any time the aggregate amount of any IBOR Loan is reduced, by payment, prepayment, or conversion of part thereof, to be less than \$500,000, such IBOR Loan shall automatically convert into a Reference Rate Loan, and on and after such date the right of Borrower to continue such Loan as, and convert such Loan into, an IBOR Loan shall terminate.

- (b) Borrower shall deliver a Request for Redesignation in the form of the attached Exhibit D to be received by Bank not later than 9:00 a.m. (California time) at least (i) three Business Days in advance of the Redesignation Date, if the Loan is to be converted into or continued as an IBOR Loan bearing interest determined in relation to the IBOR Rate; and (ii) one Business Day in advance of the Redesignation Date, if the Loan is to be converted into a Reference Rate Loan, specifying:
 - A. the proposed Redesignation Date;
 - B. the aggregate amount of the Loan to be converted or continued;
 - C. the Type of Loan resulting from the proposed conversion or continuation; and
 - D. other than in the case of redesignation into a Reference Rate Loan, the duration of the requested Interest Period.
- (c) If, upon the expiration of any Interest Period for an IBOR Loan, Borrower has failed to select timely a new Interest Period for a refinancing IBOR Loan, or if any Default or Event of Default then exists, Borrower shall be deemed to have elected to convert such IBOR Loan into a Reference Rate Loan effective as of the expiration date of such Interest Period.
- (d) During the existence of a Default or Event of Default, Borrower may not elect to have a Loan converted into or continued as an IBOR Loan.
- (e) After giving effect to any redesignation of a Loan, there may not be more than ten different Interest Periods in effect with respect to IBOR Loans.

- (a) Subject to the terms and conditions hereof, at any time and from time to time from the Restated Closing Date through the Banking Day immediately preceding the Revolver Termination Date, Bank shall issue such Standby Letters of Credit as Borrower may request by a Request for Standby Letter of Credit; provided that, upon giving effect to such Standby Letter of Credit, (i) Total Line A Outstandings shall not exceed the Line A Commitment, and (ii) Outstanding Standby Letters of Credit shall not exceed \$1,000,000. Unless the Bank otherwise consents in writing, the term of any Standby Letter of Credit shall not exceed 18 months and shall in no event extend beyond the Revolver Termination Date. No Standby Letter of Credit shall be issued except to the extent reasonably necessary in the ordinary course of the business of Borrower or its Subsidiaries, and no Standby Letter of Credit shall be issued in any event to support any workers' compensation obligation of Borrower or its Subsidiaries. A Standby Letter of Credit may include a provision providing that the expiry date thereof will be automatically extended each year for an additional year unless Bank gives written notice to the contrary; provided, however, that each Standby Letter of Credit shall include a final expiry date which shall not be subject to automatic extension. Unless otherwise agreed to by Bank, the face amount of any Standby Letter of Credit shall not be less than \$100,000.
- (b) Each Request for Standby Letter of Credit shall be submitted to Bank not later than 11:00 a.m., Los Angeles time, at least ten (10) Banking Days prior to the date upon which the requested Standby Letter of Credit is to be issued.
- (c) Borrower agrees to pay to Bank, at its office designated as its address for notices pursuant to this Agreement, or at such other payment location as Bank shall have specified in writing to Borrower, with respect to each Standby Letter of Credit, within one (1) Banking Day after demand therefor, a principal amount equal to any payment made by the Bank under that Standby Letter of Credit, together with interest on such amount from the date of any payment made by Bank through the date of payment by Borrower at the rate provided for in Section 3.6. The principal amount of any such payment made by Borrower to Bank shall be used to reimburse Bank for the payment made by it under the Standby Letter of Credit.

- (d) Each time Borrower fails to make any payment required by Section 2.5(c), Bank may, but is not required to, without notice to or the consent of Borrower, make a Revolving Loan under the Line A Commitment in an aggregate amount equal to the amount paid by Bank on the relevant Standby Letter of Credit, whether or not the same would cause Total Line A Outstandings to exceed the Line A Commitment (without waiving the obligation of Borrower to reduce Total Line A Outstandings to an amount less than or equal to the Line A Commitment), and, for this purpose, the conditions precedent set forth in Article 8 and the amount limitations set forth in Section 2.3(a) shall not apply. The proceeds of such Revolving Loan shall be paid to Bank to reimburse it for the payment made by it under the Standby Letter of Credit.
- (e) The issuance of any supplement, modification, amendment, renewal or extension to or of any Standby Letter of Credit shall be treated in all respects the same as the issuance of a new Standby Letter of Credit.
- (f) The obligation of Borrower to pay to Bank the amount of any payment made by Bank under the Standby Letter of Credit shall be absolute, unconditional and irrevocable. Without limiting the foregoing, such obligation of Borrower shall not be affected by any of the following circumstances:
- (i) any lack of validity or enforceability of the Standby Letter of Credit, this Agreement, or any other agreement or instrument relating thereto;
 - (ii) any amendment or waiver of or any consent to departure from the Standby Letter of Credit, this Agreement, or any other agreement or instrument relating thereto;
 - (iii) the existence of any claim, setoff, defense or other rights which Borrower may have at any time against Bank, any beneficiary of the Standby Letter of Credit (or any Persons or entities for whom any such beneficiary may be acting) or any other Person, whether in connection with the Standby Letter of Credit, this Agreement or any other agreement or instrument relating thereto, or any unrelated transactions;
 - (iv) any demand, statement or any other document presented under the Standby Letter of Credit proving to be forged, fraudulent, invalid or insufficient in any respect or any statement therein being untrue or inaccurate in any respect whatsoever;

- (v) payment by Bank under the Standby Letter of Credit against presentation of a draft or any accompanying document which does not strictly comply with the terms of the Standby Letter of Credit, but which contains some ministerial discrepancy, the existence of which does not have a material adverse effect on the business, operations or condition (financial or otherwise) of Borrower and its Subsidiaries, taken as a whole;
- (vi) the solvency (or insolvency) or financial responsibility (or lack thereof) of any party issuing any documents in connection with the Standby Letter of Credit;
- (vii) any error in the transmission of any message relating to the Standby Letter of Credit not caused by Bank, or any delay or interruption in any such message; and/or
- (viii) any error, neglect or default of any correspondent of Bank in connection with the Standby Letter of Credit.
- (g) On and after the Restated Closing Date, Standby Letter of Credit Number 216411, as amended, issued by Bank and with a face amount as of the Restated Closing Date in the amount of \$341,512, shall be deemed for all purposes, including for purposes of the fees to be collected pursuant to Section 3.4, and reimbursement of costs and expenses to the extent provided herein, a Standby Letter of Credit outstanding under this Agreement and entitled to the benefits of this Agreement and the other Loan Documents, and shall be governed by the applications and agreements pertaining thereto and by this Agreement.
- (h) Upon (i) the request of Bank, (A) if Bank has honored any full or partial drawing request on any Standby Letter of Credit and such drawing has resulted in a Line A Loan hereunder, or (B) if, as of the Revolver Termination Date, any Standby Letter of Credit may for any reason remain outstanding and partially or wholly undrawn, or (ii) upon the occurrence of a Default or an Event of Default, then, Borrower shall immediately Cash Collateralize the Letter of Credit Obligations in an amount equal to such Letter of Credit Obligations.
- (i) Borrower agrees to sign Bank's form Application and Agreement for Standby Letter of Credit upon demand by Bank.

- (j) Subject to the next sentence, a Standby Letter of Credit may be requested pursuant to this Section 2.5 for the account of Borrower or for the account of any Subsidiary of Borrower. To the extent that a Subsidiary of Borrower is the account party under any Standby Letter of Credit, Borrower hereby guarantees the payment and performance of such Subsidiary with respect to any Obligation of such Subsidiary relating to such Standby Letter of Credit, and agrees to deliver to Bank, duly executed and in form and content acceptable to Bank, a duly executed continuing guaranty further evidencing the foregoing guaranty, together with a resolution or other evidence of the corporate authority of Borrower to execute, perform and deliver such continuing guaranty.
- 2.6 Voluntary Reduction of the Commitments. Borrower shall have the right, at any time and from time to time, without penalty or charge, upon at least two (2) Banking Days prior written notice to Bank, voluntarily to reduce, permanently and irrevocably, in aggregate principal amounts of an integral multiple of \$50,000 (but not less than \$250,000), or to terminate, the then undisbursed portion of the Commitments, provided that any such reduction or termination shall be accompanied by all accrued and unpaid commitment fees, if any, with respect to the portion of the Commitments being reduced or terminated, and provided further that the Line A Commitment shall not be reduced to an amount less than the amount of Total Line A Outstandings and the Line B Commitment shall not be reduced to an amount less than the amount of Total Line B Outstandings.
 - 2.7 Automatic Reduction of the Line A Commitment.
 - (a) The Line A Commitment shall automatically and permanently reduce on each December 31 in the amount set forth below opposite the year in which such date occurs:

Year	Annual Reduction To Line A Commitment
1996 1997 1998	\$ 3,000,000 \$ 4,000,000 \$ 3,000,000

(b) In addition to the foregoing, on or before March 30 of each Fiscal Year (commencing March 30, 1997), the Line A Commitment shall automatically and permanently reduce by an amount equal to 75% of the Excess Cash Flow of Borrower and its Subsidiaries for the immediately prior Fiscal Year; provided, however, that at such time as the

Line A Commitment is less than or equal to \$14,000,000, no further reductions thereof shall be required pursuant to this Section 2.7(b).

- (c) In addition to the foregoing, the Line A Commitment shall, concurrently with the receipt by the Borrower (or any Subsidiary) of the net cash proceeds from the issuance of equity securities, be reduced by the amount of such net cash proceeds. As used in the preceding sentence, "equity securities" shall not include (i) any security issued upon the conversion of the Borrower's Subordinated Obligations, or (ii) any security issued under the Borrower's employee stock plans to the extent that the net cash proceeds of the sale of such securities do not exceed \$100,000 in the aggregate in any Fiscal
- (d) In addition to the foregoing, the Line A Commitment shall automatically reduce upon any Disposition by an amount equal to 80% of the Net Cash Proceeds received by the Borrower or its Subsidiaries upon such Disposition.
- 2.8 Automatic Increase of the Line A Commitment. The Line A Commitment shall automatically increase on the Line B Termination Date by an amount equal to Total Line B Outstandings on such date, provided, however, in no event shall the Line A Commitment exceed \$24,000,000.
- 2.9 Automatic Termination of the Line B Commitment. The Line B Commitment will terminate on the Line B Termination Date.

ARTICLE 3. PAYMENTS AND FEES

3.1 Principal and Interest.

- (a) Interest shall be payable on the outstanding daily unpaid principal amount of each Loan from the date thereof until payment in full is made and shall accrue and be payable at the rates set forth herein both before and after default and before and after maturity and judgment, with interest on overdue interest to bear interest at the rate set forth in Section 3.6, to the fullest extent permitted by applicable Law. Upon prepayment or payment in full of all outstanding Loans, interest accrued through the date of such prepayment or payment shall be payable on such date.
- (b) Interest accrued on each Reference Rate Loan shall be payable monthly on or before the first Banking Day of each fiscal month of Borrower, commencing with the first such date to occur following the Restated Closing Date. Bank shall use its best efforts to notify Borrower of the amount of interest so payable prior to each interest payment date, but failure of Bank to do so shall not excuse payment of such interest when payable. Except as otherwise provided in Section 3.6, the unpaid principal amount of any Reference Rate Loan shall bear interest at a fluctuating rate per annum equal to the Reference Rate plus the applicable Reference Rate Spread. Each change in the interest rate shall take effect simultaneously with the corresponding change in the Reference Rate and/or the Reference Rate Spread. Each change in the Reference Rate shall be effective as of 12:01 a.m. on the Banking Day on which the change in the Reference Rate is announced, unless otherwise specified in such announcement, in which case the change shall be effective as so specified.
- (c) Interest accrued on each IBOR Loan which is for a term of three months or less shall be due and payable on the last day of the related Interest Period. Interest accrued on each other IBOR Rate Loan shall be due and payable on the date which is three months after the date such IBOR Loan was made and on the last day of the related Interest Period. Except as otherwise provided in Section 3.6, the unpaid principal amount of any IBOR Loan shall bear interest at a rate per annum equal to the IBOR Rate for that IBOR Loan plus the applicable IBOR Rate Spread.
- (d) If not sooner paid, the principal indebtedness evidenced by the Notes shall be payable as follows:

- (i) the amount, if any, by which the principal indebtedness evidenced by the Revolving Line Note at any time exceeds the Maximum Line A Loan Amount shall be payable immediately.
- (ii) the principal indebtedness evidenced by the Revolving Line Note shall be payable on the Revolver Termination Date.
- (iii) the principal indebtedness evidenced by the Line B Note shall be payable on the Line B Termination Date. Bank acknowledges that Borrower may use the proceeds of Revolving Loans (if available) to repay the outstanding Line B Loans on such date.
- (iv) the amount, if any, by which the principal indebtedness evidenced by the Line B Note at any time exceeds the Line B Commitment shall be payable immediately.
- (e) The Notes, or either of them, may, at any time and from time to time, be voluntarily paid or prepaid in whole or in part without premium or penalty; provided that: (i) any partial prepayment shall be in an amount not less than \$500,000 in the case of an IBOR Loan, (ii) Bank shall have received notice of any prepayment at least three Banking Days before the date of prepayment, in the case of an IBOR Loan, and at least one Banking Day before the date of a prepayment, in the case of a Reference Rate Loan, (iii) each prepayment of principal shall be accompanied by payment of interest accrued through the date of payment on the amount of principal paid, and (iv) in the case of any prepayment of any IBOR Loan, Borrower shall promptly upon demand reimburse Bank for any loss or cost directly or indirectly resulting from the prepayment, determined as set forth in Section 3.5.

3.2 Facility Fee.

- (a) Borrower shall pay to Bank a one time nonrefundable Line B facility fee in an aggregate amount equal to \$25,000 with respect to the Line B Commitment. Bank acknowledges that it has received such fee.
- (b) On or before he Restated Closing Date, Borrower shall pay to Bank a nonrefundable Line A facility fee in an amount equal to .25% of the Line A Commitment, less \$18,714. Thereafter, on each July 1, commencing July 1, 1997, Borrower shall pay to Bank an annual Line A facility fee in an amount equal to 0.25% of the Line A Commitment on such date. Such facility fee shall be fully earned as

of the date when due and shall be non-refundable under any circumstances. Such facility fee shall be due and payable on or before the Restated Closing Date.

- 3.3 Commitment Fee and Special Commitment Costs.
- (a) For the period from the Restated Closing Date through the Revolver Termination Date, Borrower shall pay to Bank a commitment fee equal to .25% per annum times the average daily difference between the Line A Commitment and the Total Line A Outstandings. Such commitment fee shall be payable quarterly in arrears within 5 Banking Days after the end of each calendar quarter, commencing with the quarter ending June 30, 1996. Bank shall use its best efforts to notify Borrower of the amount of the commitment fee so payable prior to each fee payment date, but failure of Bank to do so shall not excuse payment of such fee when payable.
- (b) On the Line B Termination Date, Borrower shall pay to Bank a commitment fee equal to .50% per annum times the average Total Line B Outstandings from the Restated Closing Date through the Line B Termination Date.
- (c) If Bank determines that compliance with any Law or regulation or with any guideline or request from any central bank or other Governmental Agency (whether or not having the force of Law) relating to the capital adequacy of banks or corporations in control of banks has or would have the effect of reducing the rate of return on the capital of Bank or any corporation controlling Bank as a consequence of, or with reference to, Bank's Commitments hereunder below the rate which Bank or such other corporation could have achieved but for such compliance (taking into account the policies of Bank or corporation with regard to capital adequacy), then Borrower shall from time to time, upon demand by Bank in accordance with this Section 3.3, within 15 days after demand, pay to Bank additional amounts sufficient to compensate Bank or such other corporation for such reduction.
 - (i) Bank may not seek compensation under this Section 3.3(c) for any reduction in rate of return in respect of any period which is more than eighteen months prior to the Bank's demand for such compensation.
 - (ii) A certificate as to any amounts for which Bank is seeking compensation under Section 3.3.(c), submitted to Borrower by Bank, shall be conclusive and binding for all purposes, absent manifest error. Bank shall calculate such amounts in a manner which

is consistent with the manner in which it makes calculations for comparable claims with respect to similarly situated borrowers from Bank and will not allocate to Borrower a proportionately greater amount of such compensation than it allocates to each of its other commitments to lend or other loans with respect to which it is entitled to demand comparable compensation. Bank agrees promptly to notify Borrower of any circumstances that would cause Borrower to pay additional amounts pursuant to this Section 3.3.(c), provided that the failure to give such notice shall not affect Borrower's obligation to pay such additional amounts hereunder.

- 3.4 Standby Letter of Credit Fees and Costs. Borrower shall pay a letter of credit fee equal to the greater of \$500 or 2.00% per annum of the outstanding undrawn amount of each Standby Letter of Credit issued under Section 2.5, payable at the time of issuance. Each Standby Letter of Credit fee is earned upon issuance of each Standby Letter of Credit and is nonrefundable. In addition, Borrower shall pay to Bank upon demand issuance and/or other fees the Bank notifies Borrower will be charged for issuing and processing Borrower's Letters of Credit.
 - 3.5 IBOR Fees and Costs.
 - (a) If the occurrence of any Regulatory Development after the Restated Closing Date:
 - (1) shall subject Bank or its IBOR Lending Office to any tax, duty or other charge or cost with respect to any IBOR Loan or its obligation to make IBOR Loans, or shall change the basis of taxation of payments to Bank of the principal of or interest on any IBOR Loan or any other amounts due under this Agreement in respect of any IBOR Loan or its obligation to make IBOR Loans (except for changes in any tax on the overall net income, gross income or gross receipts of Bank or its IBOR Lending Office);
 - (2) shall impose, modify or deem applicable any reserve (including, without limitation, any reserve imposed by the Board of Governors of the Federal Reserve System), special deposit or similar requirements (excluding any such requirement included in any applicable Reserve Percentage) against assets of, deposits with or for the account of, or credit extended by, Bank or its IBOR Lending Office; or
 - (3) shall impose on Bank or its IBOR Lending Office or the IBOR Market any other condition ${\sf Cond}({\sf Cond}($

affecting any IBOR Loan or its obligation to make IBOR Loans, or shall otherwise affect any of the same:

and the result of any of the foregoing, as determined by Bank, increases the cost to Bank or its IBOR Lending Office of making or maintaining any IBOR Loan or in respect of any IBOR Loan or its obligation to make IBOR Loans or reduces the amount of any sum received or receivable by Bank or its IBOR Lending Office with respect to any IBOR Loan or its obligation to make IBOR Loans (assuming Bank's IBOR Lending Office had funded 100% of its IBOR Loan in the IBOR Market), then, within 15 days after demand by Bank, Borrower shall pay to Bank such additional amount or amounts as will compensate Bank for such increased cost or reduction (determined as though Bank's IBOR Lending Office had funded 100% of its IBOR Loan in the IBOR Market); provided that Borrower shall not be liable to Bank for any such increased cost or reduction pursuant to this Section in respect of any period which is more than eighteen months prior to Bank's demand for such compensation. A statement of Bank claiming compensation under this subsection and setting forth the additional amount or amounts to be paid to it hereunder shall be conclusive in the absence of manifest error. Bank agrees to endeavor promptly to notify Borrower of any event of which it has actual knowledge which will entitle Bank to compensation pursuant to this Section, and agrees to designate a different IBOR Lending Office if such designation will avoid the need for or reduce the amount of such compensation and will not, in the judgment of Bank, otherwise be disadvantageous to Bank. If Bank claims compensation under this Section, Borrower may at any time, upon at least four (4) IBOR Market Days' prior notice to Bank and upon payment in full of the amounts provided for in this Section through the date of such payment plus any prepayment fee required by Section 3.8(d), pay in full the affected IBOR Loans of such Bank or request that such IBOR Loans be converted to Reference Rate Loans.

(b) If after the Restated Closing Date the occurrence of any Regulatory Development shall, in the opinion of Bank, make it unlawful or impossible for Bank or its IBOR Lending Office to make, maintain or fund its portion of any IBOR Loan, or to take deposits of dollars in the IBOR Market, or to determine or charge interest rates based upon the IBOR, then Bank's obligation to make IBOR Loans shall be suspended for the duration of such illegality or impossibility and Bank shall give notice thereof to Borrower. Before giving any notice pursuant to this Section, Bank shall designate a different Lending Office if such designation will avoid the need for giving

such notice and will not, in the judgment of Bank, be otherwise disadvantageous to Bank. Upon receipt of such notice, the outstanding principal amount of Bank's IBOR Loans, together with accrued interest thereon, automatically shall be converted to Reference Rate Loans on either (1) the last day of the Interest Period(s) applicable to such IBOR Loans if Bank may lawfully continue to maintain and fund such IBOR Loans to such day(s) or (2) immediately if Bank may not lawfully continue to fund and maintain such IBOR Loans to such day(s), provided that in such event the conversion shall not be subject to payment of a prepayment fee under Section 3.8(d). In the event that Bank is unable, for the reasons set forth above, to make, maintain or fund its portion of any IBOR Loan, Bank shall fund such amount as a Reference Rate Loan for the same period of time, and such amount shall be treated in all respects as a Reference Rate Loan.

- (c) If, with respect to any proposed IBOR Loan, the Bank reasonably determines that, by reason of circumstances affecting the IBOR Market generally that are beyond the reasonable control of Bank, deposits in dollars (in the applicable amounts) are not being offered to Bank in the IBOR Market for the applicable Interest Period, then Bank shall give notice thereof to Borrower, whereupon until Bank notifies Borrower that the circumstances giving rise to such suspension no longer exist, the obligation of Bank to make any future IBOR Loan shall be suspended. If at the time of such notice there is then pending a Request for Loan that specifies an IBOR Loan, such Request for Loan shall be deemed to specify a Reference Rate Loan.
- (d) Upon payment or prepayment of any IBOR Loan (other than as the result of a conversion required under Section 3.8(b)) on a day other than the last day in the applicable Interest Period (whether voluntarily, involuntarily, by reason of acceleration, or otherwise), or upon the failure of Borrower to borrow on the date or in the amount specified for a IBOR Loan in any Request for Loan, Borrower shall pay to Bank any cost or expense incurred by Bank as a result of such payment or prepayment, together with a prepayment fee in an amount equal to the amount, if any, by which (x) the additional interest that would have accrued (without any applicable IBOR Spread) on the principal amount prepaid on account of the IBOR Loan had it remained outstanding until the last day of the applicable Interest Period, exceeds (y) the interest that Bank could recover by placing funds in the amount of the prepayment on deposit in the IBOR Market selected by Bank for a period beginning on the date of the prepayment and ending on the last day of the applicable Interest Period, or for a comparable period for which an appropriate rate

quote may be obtained. Bank's determination of the amount of any prepayment fee or failure to borrow fee payable under this Section 3.5(d) shall be conclusive in the absence of manifest error.

- (e) Any statement or certificate given by Bank under this Section 3.5(e) shall satisfy the requirements set forth in Section 3.3(c) with respect to requests for reimbursement under Section 3.3(a)
- 3.6 Default Rate/Late Payments. Upon the occurrence and during the continuation of any default under this Agreement, Loans under this Agreement shall, at the option of Bank, bear interest at a rate per annum which is 2.5 percentage points higher than the highest rate of interest otherwise provided under this Agreement. This will not constitute a waiver of any Default. In addition, should any installment of principal or interest or any fee or cost or other amount payable under any Loan Document to Bank not be paid when due, it shall thereafter bear interest at a fluctuating interest rate per annum at all times equal to the highest rate of interest otherwise provided under this Agreement plus the Reference Rate Spread plus 2.00% per annum, to the fullest extent permitted by applicable Law. This will not constitute a waiver of any default. Accrued and unpaid interest on past due amounts (including, without limitation, interest on past due interest) shall be compounded monthly, on the last day of each calendar month, to the fullest extent permitted by applicable Law.
- 3.7 Computation of Interest and Fees. All computations of interest and fees under any Loan Document shall be calculated on the basis of a year of 360 days and the actual number of days elapsed, which results in more interest and fees than if a 365-day year were used.
- 3.8 Non-Banking Days. If any payment to be made by Borrower or any other Party under any Loan Document shall come due on a day other than a Banking Day, payment shall be made on the next succeeding Banking Day and the extension of time shall be reflected in computing interest.
 - 3.9 Manner and Treatment of Payments.
 - (a) Each payment hereunder or on the Notes or under any other Loan Document shall be made to Bank, at Bank's Office, in immediately available funds not later than 12:00 noon, Los Angeles time, on the day of payment (which must be a Banking Day). All payments received after 12:00 noon, Los Angeles time, on any Banking Day, shall be deemed received on the next succeeding Banking Day. All payments shall be made in lawful money of the United States of America.

- (b) Bank shall use its best efforts to keep a record of Loans made by it and payments received by it with respect to the Notes and such record shall be presumptive evidence of the amounts owing. Notwithstanding the foregoing sentence, Bank shall not be liable to any Party for any failure to keep such a record.
- (c) Each payment of any amount payable by Borrower and/or any other Party under this Agreement and/or any other Loan Document shall be made free and clear of, and without reduction by reason of, any taxes, assessments or other charges imposed by any Governmental Agency, central bank or comparable authority.
- 3.10 Funding Sources. Nothing in this Agreement shall be deemed to obligate Bank to obtain the funds for any Loan in any particular place or manner or to constitute a representation by Bank that it has obtained or will obtain the funds for any Loan in any particular place or manner.
- 3.11 Failure to Charge Not Subsequent Waiver. Any decision by Bank not to require payment of any interest (including default interest), fee, cost or other amount payable under any Loan Document on any occasion shall in no way limit or be deemed a waiver of Bank's right to require full payment of any interest (including default interest), fee, cost or other amount payable under any Loan Document on any other or subsequent occasion.
- 3.12 Survivability. All of Borrower's obligations under this Article 3 shall survive for one year following the date on which all Loans hereunder were fully paid.
 - 3.13 Direct Debit (Pre-Billing).
 - (a) Borrower agrees that Bank will debit Borrower's Designated Deposit Account on the date each payment of principal and interest and any fees from Borrower becomes due (the "Due Date"). If the Due Date is not a Banking Day, the Designated Deposit Account will be debited on the next Banking Day.
 - (b) Approximately 10 days prior to each Due Date, Bank will mail to Borrower a statement of the amounts that will be due on that Due Date (the "Billed Amount"). The calculation will be made on the assumption that no new extensions of credit or payments will be made between the date of the billing statement and the Due Date, and that there will be no changes in the applicable interest rate.

- (c) Bank will debit the Designated Deposit Account for the Billed Amount, regardless of the actual amount due on that date (the "Accrued Amount"). If the Billed Amount debited to the Designated Deposit Account differs from the Accrued Amount, the discrepancy will be treated as follows:
 - (i) If the Billed Amount is less than the Accrued Amount, the Billed Amount for the following Due Date will be increased by the amount of the discrepancy. Borrower will not be in default by reason of any such discrepancy.
 - (ii) If the Billed Amount is more than the Accrued Amount, the Billed Amount for the following Due Date will be decreased by the amount of the discrepancy.

Regardless of any such discrepancy, interest will continue to accrue based on the actual amount of principal outstanding without compounding. Bank will not pay Borrower interest on any overpayment.

(d) Borrower will maintain sufficient funds in the Designated Deposit Account to cover each debit. If there are insufficient funds in the Designated Deposit Account on the date Bank enters any debit authorized by this Agreement, the debit will be reversed.

ARTICLE 4. REPRESENTATIONS AND WARRANTIES

Borrower represents and warrants to Bank, as of the Restated Closing Date, that:

- 4.1 Existence and Qualification; Power; Compliance With Laws. Borrower is a corporation duly formed, validly existing and in good standing under the Laws of Delaware. The chief executive offices of Borrower are in Carson, California. Borrower is duly qualified or registered to transact business and is in good standing in California and each other jurisdiction in which the conduct of its business or the ownership or leasing of its Properties makes such qualification or registration necessary, except where the failure so to qualify or register and to be in good standing would not have a material adverse effect on the business, operations or condition (financial or otherwise) of Borrower and its Subsidiaries, taken as a whole. Borrower has all requisite power and authority to conduct its business, to own and lease its Properties and to execute, deliver and perform all of its Obligations under the Loan Documents. All outstanding shares of capital stock of Borrower are duly authorized, validly issued, fully paid, non-assessable and issued in compliance with all applicable state and federal securities and other Laws. Borrower is in compliance with all Laws and other legal requirements applicable to its business, has obtained all authorizations, consents, approvals, orders, licenses and permits from, and has accomplished all filings, registrations and qualifications with, or obtained exemptions from any of the foregoing from, any Governmental Agency that are necessary for the transaction of its business, except where the failure so to comply, file, register, qualify or obtain exemptions would not have a material adverse effect on the business, operations or condition (financial or otherwise) of Borrower and its Subsidiaries, taken as a whole.
- 4.2 Authority; Compliance With Other Agreements and Instruments and Government Regulations. The execution, delivery and performance by each of Borrower and its Subsidiaries of the Loan Documents to which it is a Party have been duly authorized by all necessary action, and do not and will not:
 - (a) Require any consent or approval not heretofore obtained of any partner, director, stockholder, security holder or creditor;
 - (b) Violate or conflict with any provision of such Party's partnership agreement, certificate of limited

partnership, charter, articles of incorporation or bylaws, or amendments thereto, as applicable;

- (c) Result in or require the creation or imposition of any Lien or Right of Others (other than as provided under the Loan Documents) upon or with respect to any Property now owned or leased or hereafter acquired by such Party;
- (d) Violate any provision of any Law (including, without limitation, Regulations G, T, U and/or X of the Board of Governors of the Federal Reserve System), order, writ, judgment, injunction, decree, determination or award presently in effect and having applicability to such Party; or
- (e) Result in a breach of or constitute a default under, or cause or permit the acceleration of any obligation owed under, any indenture or loan or credit agreement or any other material agreement, lease or instrument to which such Party is a party or by which such Party or any of its Property is bound or affected;

and neither Borrower nor any Subsidiary thereof is in default under any Law, order, writ, judgment, injunction, decree, determination or award, or any indenture, agreement, lease or instrument described in Section 4.2(e), in any respect that is materially adverse to the interests of Bank or that would have any material adverse effect on the business, operations or condition (financial or otherwise) of Borrower and its Subsidiaries, taken as a whole.

4.3 No Governmental Approvals Required. No authorization, consent, approval, order, license or permit from, or filing, registration or qualification with, or exemption from any of the foregoing from, any Governmental Agency is or will be required to authorize or permit under applicable Law the execution, delivery and performance by Borrower or any Subsidiary thereof of the Loan Documents to which it is a Party.

4.4 Subsidiaries.

(a) Schedule 4.4 hereto correctly sets forth the names, forms of legal entity and jurisdictions of formation of all Subsidiaries of Borrower. Except as described in Schedule 4.4, Borrower does not own any capital stock, partnership interest, joint venture interest or other equity interest in any Person. Unless otherwise indicated in Schedule 4.4, all of the outstanding shares of capital stock or partnership or joint venture interests of each Subsidiary of Borrower are owned of record and benefici-

ally by Borrower, and all securities and interests so owned are duly authorized, validly issued, fully paid, non-assessable and issued in compliance with all applicable state and federal securities and other Laws, and are free and clear of all Liens and Rights of Others.

- (b) Each Subsidiary of Borrower is a legal entity of the form described for that Subsidiary in Schedule 4.4, duly formed, validly existing and in good standing under the Laws of its jurisdiction of formation, is duly qualified or registered to transact business and is in good standing in each other jurisdiction in which the conduct of its business or the ownership or leasing of its Properties makes such qualification or registration necessary, except where the failure so to qualify or register and to be in good standing would not have a material adverse effect on the business, operations or condition (financial or otherwise) of Borrower and its Subsidiaries, taken as a whole, and has all requisite legal power and authority to conduct its business and to own and lease its Properties and to execute, deliver and perform all of its Obligations under the Loan Documents.
- (c) Each Subsidiary of Borrower is in compliance with all Laws (subject to the matters referred to in Schedule 4.20) and other legal requirements applicable to its business, has obtained all authorizations, consents, approvals, orders, licenses and permits from, and has accomplished all filings, registrations and qualifications with, or obtained exemptions from any of the foregoing from, any Governmental Agency that are necessary for the transaction of its business, except where the failure so to comply, file, register, qualify or obtain exemptions would not have a material adverse effect on the business, operations or condition (financial or otherwise) of Borrower and its Subsidiaries, taken as a whole.
- 4.5 Financial Statements. Borrower has furnished to Bank (a) the audited consolidated balance sheet of Borrower and its Subsidiaries as at December 31, 1995, and audited consolidated statements of profit and loss and of changes in financial position of Borrower and its Subsidiaries for their fiscal year then ended, and (b) the unaudited consolidated and consolidating balance sheets of Borrower and its Subsidiaries as at March 31, 1996, and unaudited consolidated and consolidating statements of profit and loss and of changes in financial position of Borrower and its Subsidiaries for such quarter and for the portion of their fiscal year ended with such quarter. Such financial statements fairly present in all material respects the financial condition, results of operations and changes in financial position of Borrower and its Subsidiaries as at such dates and for such periods, in conformity with

generally accepted accounting principles, consistently applied, provided that the balance sheets and statements referred to in (b) above are subject to normal year-end audit adjustments.

- 4.6 No Other Liabilities; No Material Adverse Changes. Borrower and its Subsidiaries do not have any material liability or material contingent liability not reflected or disclosed in the financial statements or notes thereto described in Section 4.5. There has been no material adverse change in the business, operations or condition (financial or otherwise) of Borrower and its Subsidiaries, taken as a whole, since the date of the financial statements described in Section 4.5(b).
- 4.7 Title to and Location of Property. Borrower and its Subsidiaries have good and valid title to all the Property reflected in the financial statements described in Section 4.5, other than Property subsequently sold in the ordinary course of business, free and clear of all Liens and Rights of Others other than Liens or Rights of Others permitted pursuant to Section 6.6. Schedule 4.7 hereto sets forth the street address for all material interest in real Property owned or held by Borrower or any of its Subsidiaries. All tangible personal Property owned or held by Borrower or any of its Subsidiaries is located within the States of California and Texas and the Country of Australia.
- 4.8 Intangible Assets. Borrower and its Subsidiaries own, or possess the unrestricted right to use, all trademarks, trade names, copyrights, patents, patent rights, licenses and other Intangible Assets that are used in the conduct of their businesses as now operated, and no such Intangible Asset, to the best knowledge of Borrower, conflicts with the valid trademark, trade name, copyright, patent, patent right or Intangible Asset of any other Person to the extent that such conflict would have a material adverse effect on the business, operations or condition (financial or otherwise) of Borrower and its Subsidiaries, taken as a whole. Schedule 4.8 contains a complete and accurate description of all pending and existing trademarks, tradenames, patents and patent rights.
- 4.9 Filing of Financing Statements. Except for the requirement that continuation statements periodically be filed and/or recorded with respect to financing statements describing the Collateral previously filed with the California Secretary of State, all necessary steps will have been taken to fully perfect and to maintain fully perfected the Liens of Bank on the Collateral, to the fullest extent that such Liens may be perfected by the filing and/or recording of financing statements pursuant to Division 9 of the California Uniform Commercial Code.

- 4.10 Public Utility Holding Company Act. Neither Borrower nor any Subsidiary thereof is a "holding company", or a "subsidiary company" of a "holding company", or an "affiliate" of a "holding company" or of a "subsidiary company" of a "holding company", within the meaning of the Public Utility Holding Company Act of 1935, as amended.
- 4.11 Litigation. Except for (i) any matter fully covered (subject to usual deductibles and retentions) by insurance for which the insurance carrier has assumed full responsibility, (ii) any matter, or series of related or similar matters, involving a claim against Borrower or any of its Subsidiaries of less than \$500,000, and (iii) matters set forth in Schedule 4.11 and in the financial statements (or the notes thereto) described in Section 4.5, there are no actions, suits or proceedings pending or, to the best knowledge of Borrower, threatened against or affecting Borrower or any of its Subsidiaries or any Property of any of them in any court of Law or before any Governmental Agency that would impair Borrower's or any Subsidiary's financial condition or ability to repay the Loans.
- 4.12 Binding Obligations. Each of the Loan Documents to which Borrower or any Subsidiary thereof is a Party will, when executed and delivered by such Party, constitute the legal, valid and binding obligation of such Party, enforceable against such Party in accordance with its terms, except as enforcement may be limited by bankruptcy, insolvency, reorganization, arrangement, moratorium or other similar Laws relating to or affecting creditors' rights generally or equitable principles relating to the granting of specific performance and other equitable remedies as a matter of judicial discretion.
- $4.13\,$ No Default. No event has occurred and is continuing that is a Default.

4.14 ERISA Plans.

- (a) Borrower and each of its Subsidiaries has fulfilled its obligations, if any, under the minimum funding standards of ERISA and the Code with respect to each Plan and is in compliance in all material respects with the presently applicable provisions of ERISA and the Code, and has not incurred any liability with respect to any Plan under Title IV of ERISA.
- (b) No reportable event has occurred under Section 4043(b) of ERISA for which the PBGC requires 30 days notice.
- (c) No action by Borrower or any Subsidiary to terminate or withdraw from any Plan has been taken and no $\,$

notice of intent to terminate a Plan has been filed under Section 4041 of ERISA.

- (d) No proceeding has been commenced with respect to a Plan under Section 4042 of ERISA, and no event has occurred or condition exists which might constitute grounds for the commencement of such a proceeding.
- 4.15 Regulations G, T, U and X; Investment Company Act. Neither Borrower nor any of its Subsidiaries is engaged principally, or as one of its important activities, in the business of extending credit for the purpose of "purchasing" or "carrying" any "margin stock" or "margin security" within the meanings of Regulations G, T, U or X, respectively, of the Board of Governors of the Federal Reserve System. If requested by Bank, Borrower will furnish or will cause its Subsidiaries, as requested, to furnish Bank with a statement or statements in conformity with the requirements of Federal Reserve Forms G-3 and/or U-1 referred to in Regulations G or U of said Board of Governors. No part of the proceeds of any Loan hereunder will be used to purchase or carry any such "margin security" or "margin stock" or to extend credit to others for the purpose of purchasing or carrying any such "margin security" or "margin stock" in violation of Regulations G, T, U or X of said Board of Governors. Neither Borrower nor any of its Subsidiaries is or is required to be registered under the Investment Company Act of 1940.
- 4.16 Disclosure. No written statement made by Borrower or any Subsidiary thereof to Bank in connection with this Agreement, or in connection with any Loan, or in connection with the issuance of any Standby Letter of Credit, contains any untrue statement of a material fact or omits a material fact necessary to make the statement made not misleading. To the best knowledge of Borrower, there is no fact which Borrower has not disclosed to Bank in writing which materially and adversely affects nor, so far as Borrower can now foresee, is reasonably likely to prove to affect materially and adversely the business, operations, Properties, prospects, profits or condition (financial or otherwise) of Borrower and its Subsidiaries, taken as a whole, or the ability of Borrower and its Subsidiaries to perform their Obligations under the Loan Documents.
- 4.17 Tax Liability. Borrower and its Subsidiaries have filed all income tax returns which are required (including any extensions obtained pursuant to applicable Law) to be filed, and have paid, or made provision for the payment of, all taxes with respect to the periods, Property or transactions covered by said returns, or pursuant to any assessment received by Borrower or any Subsidiary thereof, except (a) such taxes, if any, as are being contested in good faith by appropriate proceedings and as to which adequate reserves have been

established and maintained and (b) miscellaneous taxes not willfully unpaid in an aggregate amount not greater than \$200,000.

- 4.18 Fiscal Year. Borrower and its Subsidiaries each operate on a fiscal year ("Fiscal Year") ending on December 31, with four fiscal quarters of thirteen weeks ending on or about March 31, June 30, September 30 and December 31.
- 4.19 Employee Matters. There is no strike, work stoppage or labor dispute with any union or group of employees pending or overtly threatened involving Borrower or any of its Subsidiaries.
- 4.20 Environmental Matters. Except as disclosed on Schedule 4.20 and in the financial statements (or the notes thereto) described in Section 4.5, (i) the real Property of Borrower and the operations conducted thereon do not violate any applicable Law, statute, ordinance, rule, regulation, order or determination of any governmental authority or any restrictive covenant or deed restriction (recorded or otherwise), including without limitation all applicable zoning ordinances and building codes, flood disaster Laws and Environmental Laws and regulations, in any respect which could have a material adverse effect on the business, operations or condition (financial or otherwise) of Borrower; (ii) without limitation of clause (i) above, Borrower's real Property and the operations conducted by Borrower or any current or prior owner or operator of such real Property or operation, are not in violation of or subject to any existing, pending or threatened action, suit, investigation, inquiry or proceeding by any governmental authority or to any remedial obligations under any Environmental Laws which, if determined adversely to Borrower, could have a material adverse effect on the business, operations or condition (financial or otherwise) of Borrower; (iii) all notices, permits, licenses or similar authorizations, if any, required to be obtained or filed in connection with the operation or use of the real Property of Borrower, including without limitation past or present treatment, storage, disposal or release of a hazardous substance or solid waste into the environment, have been duly obtained or filed, except where the failure to obtain or file same would not have a material adverse effect on the business, operations or condition (financial or otherwise) of Borrower; (iv) all significant amounts of hazardous waste generated at the real Property of Borrower have in the past been and shall continue to be transported, treated and disposed of only by carriers maintaining valid permits under RCRA and any other Environmental Laws and only at treatment, storage and disposal facilities maintaining valid permits under RCRA and any other Environmental Law, which carriers and facilities have been and are, to the best of Borrower's knowledge, operating in

compliance with such permits or are exempt from obtaining such permits; (v) Borrower has no actual or constructive knowledge of the disposal or other release of any hazardous substance or solid waste, or the threatened release of hazardous substances, on or to the real Property of Borrower except in compliance with Environmental Laws, and that it has not been notified of same by any Governmental Agency; (vi) Borrower has no material contingent liability in connection with any release or threatened release of any hazardous substance or solid waste into the environment; and (vii) the use which Borrower makes or intends to make of Borrower's real Property will not result in the unlawful or unauthorized disposal or other release of any hazardous substance or solid waste on or to the real Property of Borrower. The terms "hazardous substance," "release" and "threatened release" have the meanings specified in CERCLA, and the terms "solid waste" and "disposal" (or "disposed") have the meanings specified in RCRA; provided, however, in the event either CERCLA or RCRA is amended so as to broaden the meaning of any term defined thereby, such broader meaning shall apply subsequent to the effective date of such amendment, and provided further that, to the extent the laws of any state in which any of the real Property of Borrower is located establish a meaning for "hazardous substance," "release," "solid waste" or "disposal" which is broader than that specified in either CERCLA or RCRA, such broader meaning shall apply with regard to the real Property of Borrower located in such state.

- 4.21 Intercompany Indebtedness. All intercompany indebtedness between Borrower and any of its divisions or Subsidiaries, or between any such division or Subsidiary and any other division or Subsidiary, is evidenced only as a book entry in the financial records of the parties thereto. There exists no instrument or other transferable document evidencing such intercompany debt.
- 4.22 Promissory Notes. Except as set forth on Schedule 4.22, neither Borrower nor its Subsidiaries own or hold any promissory note or other debt instrument with a book value in excess of \$50,000.
- 4.23 Projections. The pro forma financial statements delivered to Bank in accordance with Section 7.1(g) represent Borrower's best estimate, as of the date of each such statement, of the future financial performance of Borrower and its Subsidiaries for the periods set forth therein. Such pro forma financial statements are based on facts, as of the date of each such statement, known to Borrower and on assumptions that, as of the date of each such statement, are reasonable and consistent with such facts. No material (in amount and likelihood) fact or assumption is omitted as a basis for such pro forma financial statements which, in Borrower's reasonable business

judgment, should be included, and such pro forma financial statements are reasonably based on such facts and assumptions.

4.24 Teledyne Acquisition. In connection with, or as a result of, the transaction contemplated by the Teledyne Acquisition Agreement, neither Borrower nor any of its Subsidiaries will (i) assume liabilities, including Contingent Obligations, except for those liabilities with (A) a contractual amount of less than \$100,000 in the aggregate if the purchase price to be paid in connection therewith will not be decreased by an amount equal to the assumed liabilities, or (B) a contractual amount of less than \$1,000,000 in the aggregate if the purchase price to be paid in connection therewith will be decreased by an amount equal to the assumed liabilities; (ii) incur losses over the twelve month period following the consummation of such transaction; or (iii) cause a Default or Event of Default to occur. In addition, (1) the Teledyne Acquisition Price will not exceed \$4,000,000, and (2) the sum of the Teledyne Acquisition Price plus the payments made by Borrower after January 1, 1996 to purchase or redeem its Subordinated Obligations will not exceed \$12,000,000.

ARTICLE 5. AFFIRMATIVE COVENANTS (OTHER THAN INFORMATION AND REPORTING REQUIREMENTS)

So long as any Loan remains unpaid, or any other Obligation remains unpaid or unperformed, or any portion of the Commitments remains outstanding, Borrower shall, and shall cause each of its Subsidiaries to, unless Bank otherwise consents in writing:

- 5.1 Payment of Taxes and Other Potential Liens. Pay and discharge promptly all taxes, assessments and governmental charges or levies imposed upon any of them, upon their respective Property or any part thereof, upon their respective income or profits or any part thereof or upon any right or interest of Bank under any Loan Document, except that Borrower and its Subsidiaries shall not be required to pay or cause to be paid (a) any income or gross receipts tax generally applicable to banks or (b) any tax, assessment, charge or levy that is not yet past due, or is being contested in good faith by appropriate proceedings, so long as the relevant entity has established and maintains adequate reserves for the payment of the same and by reason of such nonpayment and contest no material item or portion of Property of Borrower and its Subsidiaries, taken as a whole, is in jeopardy of being seized, levied upon or forfeited, or (c) miscellaneous taxes not willfully unpaid in an aggregate amount not greater than \$200,000 so long as by reason of such nonpayment no material item or portion of Property of Borrower and its Subsidiaries, taken as a whole, is in jeopardy of being seized, levied upon or forfeited.
- 5.2 Preservation of Existence. Preserve and maintain their respective existences, licenses, rights, franchises and privileges in the jurisdiction of their formation and all authorizations, consents, approvals, orders, licenses, permits, or exemptions from, or registrations with, any Governmental Agency that are necessary for the transaction of their respective business, and qualify and remain qualified to transact business in each jurisdiction in which such qualification is necessary in view of their respective business or the ownership or leasing of their respective Properties, except that Borrower's or any Subsidiaries' failure to comply with the foregoing shall not constitute a violation of this covenant to the extent that such failure does not cause Borrower and its Subsidiaries, taken as a whole, to expend more than \$100,000 in the aggregate.
- 5.3 Maintenance of Properties. Maintain, preserve and protect all of their respective Properties and equipment in

good order and condition, subject to wear and tear in the ordinary course of business, and not permit any waste of their respective Properties, except that the failure to maintain, preserve and protect a particular item of Property or equipment that is not of significant value, either intrinsically or to the operations of Borrower and its Subsidiaries, taken as a whole, shall not constitute a violation of this covenant.

- 5.4 Maintenance of Insurance. Maintain liability and casualty insurance with responsible insurance companies in such amounts and against such risks as is usually carried by responsible companies engaged in similar businesses and owning similar Properties in the general areas in which Borrower and its Subsidiaries operate; and, as requested by Bank, cause Bank to be designated as additional insured and loss payee with respect to such insurance, and obtain the written agreement of such insurers that such insurance shall not be cancelled or terminated, nor shall the coverage or terms or exclusions thereof be materially modified, without at least thirty (30) days prior written notice to Bank.
- 5.5 Compliance With Laws. Comply with the requirements of all applicable Laws and orders of any Governmental Agency, noncompliance with which could materially adversely affect the business, operations or condition (financial or otherwise) of Borrower and its Subsidiaries, taken as a whole, except that Borrower and its Subsidiaries need not comply with a requirement then being contested by any of them in good faith by appropriate proceedings so long as no interest of Bank would be materially impaired thereby.
- 5.6 Inspection Rights. At any time during regular business hours and as often as requested, permit the Bank, or any employee, agent or representative thereof, to examine, audit and make copies and abstracts from the records and books of account of, and to visit and inspect the Properties of, Borrower and its Subsidiaries and to discuss the affairs, finances and accounts of Borrower and its Subsidiaries with any of their officers and key employees, customers or vendors, and, upon request, furnish promptly to Bank true copies of all financial information and internal management reports made available to the senior management of Borrower or any of its Subsidiaries.
- 5.7 Keeping of Records and Books of Account. Keep adequate records and books of account reflecting all financial transactions in conformity with generally accepted accounting principles, consistently applied, and in material conformity with all applicable requirements of any Governmental Agency having regulatory jurisdiction over Borrower or any of its Subsidiaries.

5.8 Compliance With Agreements, Duties and Obligations. Promptly and fully comply with all their respective agreements, duties and obligations under the Loan Documents, and under any other agreements, indentures, leases and/or instruments to which any one or more of them is a party, whether such other agreements, indentures, leases and/or instruments are with Bank or another Person, except where failure to comply with any such agreement, indenture, lease or instrument (other than any Loan Document) would not have a material adverse effect on the condition of the Borrower or any Subsidiaries.

5.9 Use of Proceeds.

- (a) Use the proceeds of the Revolving Loans for the following purposes only: (i) working capital purposes of Borrower and its Subsidiaries, (ii) to provide letters of credit required in the ordinary course of Borrower's business (other than to satisfy workers' compensation requirements), and (iii) other lawful corporate purposes in the ordinary course of business; and
- (b) Use the proceeds of Line B Loans for the following purposes only: (i) to redeem or purchase Borrower's Subordinated Obligations, and (ii) to consummate the Teledyne Acquisition.
- 5.10 Collateral and Collateral Documents. At any time, and from time to time, upon the request of Bank, promptly execute and deliver or cause to be executed and delivered to Bank Collateral Documents covering any or all of the Property of Borrower and/or its Subsidiaries, as requested, to secure payment and performance of the Obligations, or such portion thereof as may be specified by Bank, whether or not a Default shall have occurred and be continuing. Borrower hereby grants to Bank, and acknowledges the existence of, a validly perfected first priority security interest in favor of Bank in any and all personal Property of Borrower and/or its Subsidiaries other than the Excluded Equipment to secure the Obligations under the Loan Documents. From time to time, and promptly upon the request of Bank, Borrower shall (i) use its best efforts to obtain the written consents, releases, certificates, waivers, assignments, subordination agreements, landlord or mortgagee consents to removal of personal property or other agreements of such third parties as may be specified by Bank with respect to Collateral Documents, in the form furnished to Borrower by Bank in connection with such request, and (ii) provide Bank with a true and complete schedule of the value and location of such of the Property as Bank may designate. Upon such time as any of the Excluded Equipment is no longer subject to a first priority lien in favor of CIT Group/Equipment Financing, Inc. or General Electric Credit Corporation, Borrower shall take such action, and cause its Subsidiaries to take such action, as is necessary

to grant to Bank a first priority security interest in such Excluded Equipment, if any such action is necessary.

- 5.11 Subsidiary Guaranty. Cause each of its Subsidiaries, hereafter formed or acquired, to execute and deliver a joinder of the Guaranty and of each of the Collateral Documents concurrently with such formation or acquisition.
- 5.12 Bank Audit. Permit Bank, within 60 days of the Restated Closing Date and from time to time thereafter upon Bank's reasonable request, either directly or through Persons retained by Bank, to conduct a thorough annual bank audit of the accounts receivable and inventories of Borrower and its Subsidiaries and to have access to all of the books, records and employees of Borrower and its Subsidiaries reasonably incidental to such bank audit and pay the reasonable expenses of Bank or such Persons incurred in connection with such bank audit.
- 5.13 Notice of Location Change. Promptly notify Bank in writing of the occurrence of any change in the location of, or the addition of, any branch office, any field office, any warehouse or any other place of business of Borrower or any Subsidiary if such change requires any action by Bank in order to maintain the validity or to perfect any Lien in favor of Bank or results in or is accompanied by the acquisition or ownership of any property other than Property not subject to a Lien in favor of the Bank pursuant to the Collateral Documents; provided, however, that no such notification shall be required if the change, together with any other change since the Restated Closing Date, would not in the aggregate involve Property with a book value or fair market value, whichever is higher, in excess of \$200,000.
- 5.14 Interest Rate Protection Agreement. If requested by Bank, promptly enter into an interest rate protection agreement, in form and content and on terms and conditions satisfactory to the Bank, which shall provide interest rate protection for at least two years in an amount equal to or greater than at least fifty percent (50%) of the principal amount of the Loans made hereunder.
- 5.15 Teledyne Acquisition. Prior to the consummation of the Teledyne Acquisition, (i) Borrower shall deliver to Bank a list of all patents and trademarks to be acquired in connection therewith, together with such amendments to the Collateral Documents as Bank may reasonably require, (ii) Bank shall have reviewed and found satisfactory the historical compiled balance sheet, income statement and cash flow for Teledyne, to the extent that any of such items have been prepared, and (iii) Bank and/or its counsel shall have reviewed and found

 $55\,$ satisfactory the Teledyne Acquisition Agreement and the documents related thereto.

ARTICLE 6. NEGATIVE COVENANTS

So long as any Loan remains unpaid, or any other Obligation remains unpaid or unperformed, or any portion of the Commitments remains outstanding, Borrower shall not, and shall cause each of its Subsidiaries to not, unless Bank otherwise consents in writing:

- 6.1 Disposition of Property. Sell, assign, exchange, transfer, lease or otherwise dispose of, or contract to sell, assign, exchange, transfer, lease or otherwise dispose of, any of their respective Properties, whether now owned or hereafter acquired, and whether to an Affiliate or otherwise, except Properties sold, assigned, exchanged, transferred, leased or otherwise disposed of (a) in the ordinary course of business for full fair market value, (b) as permitted by Section 6.2, or (c) for consideration not to exceed \$1,000,000 in the aggregate in any twelve month period provided that such sale, assignment, exchange, transfer, lease or other disposition does not have a material adverse effect on the condition of Borrower and its Subsidiaries, taken as a whole.
- 6.2 Mergers. Merge, consolidate or amalgamate with or into any Person, except mergers, consolidations or amalgama- tions of a Subsidiary of Borrower into Borrower (with Borrower as the surviving entity) or into any other Subsidiary of Borrower.
- 6.3 Redemption, Dividends and Distributions. Redeem or repurchase stock or partnership interests, declare or pay any dividends or make any other distribution, whether of capital, income or otherwise, and whether in Cash or other Property, except (a) that any of the Subsidiaries of Borrower may declare and pay dividends or make distributions directly or indirectly to Borrower or to another Subsidiary of Borrower, (b) Borrower may repurchase shares of its common stock held by employees or former employees pursuant to employee stock ownership plans in existence on the Restated Closing Date in any aggregate amount not to exceed \$50,000 in any Fiscal Year.

6.4 ERISA.

- (a) At any time, maintain, or be or become obligated to contribute on behalf of its employees to, any pension Plan, other than those Plans in existence as of the Restated Closing Date.
 - (b) At any time, permit any Plan to:

- (1) engage in any "prohibited transaction", as such term is defined in Section 4975 of the Code;
- (2) incur any material "accumulated funding deficiency", as that term is defined in Section 302 of ERISA; or
- (3) terminate in a manner which could result in liability of Borrower or any Subsidiary thereof to the Plan or to the PBGC or the imposition of a Lien on the Property of Borrower or any Subsidiary thereof pursuant to Section 4068 of ERISA.
- (c) At any time, assume any obligation to contribute to any Multiemployer Plan, nor shall Borrower or any Subsidiary thereof acquire any Person or assets of any Person which has, or has had at any time from and after January 2, 1974, an obligation to contribute to any Multiemployer Plan.
- (d) Fail immediately to notify Bank of the occurrence of any "reportable event" (as defined in Section 4043 of ERISA) or of any "prohibited transaction" (as defined in Section 4975 of the Code) with respect to any Plan or any trust created thereunder that may have a material adverse impact on the condition of Borrower and/or its Subsidiaries, financial or otherwise. Upon request by Bank, Borrower promptly shall furnish to Bank copies of any reports or other documents filed by Borrower or any Subsidiary thereof with the United States Secretary of Labor, the PBGC and/or the Internal Revenue Service, with respect to any Plan.
- (e) At any time, permit any Plan to fail to comply with ERISA or other applicable Law in any material respect.
- 6.5 Change in Nature of Business. Make any material change in the nature of the business of Borrower and its Subsidiaries, as conducted and presently proposed to be conducted.
- 6.6 Indebtedness, Guaranties and Liens. Create, incur, assume or suffer to exist any Lien of any nature upon or with respect to any of their respective Properties, whether now owned or hereafter acquired; create, incur or assume any indebtedness for borrowed money or in connection with the purchase of Property or any liability to the issuer of any letter of credit; guaranty or otherwise become responsible (including, but not limited to, any agreement to purchase any obligations, stock, Property, goods or services or to supply or advance any funds, Property, goods or services) for the

indebtedness or obligations of any other Person; or incur any lease obligation that is required to be capitalized under generally accepted accounting principles, except:

- (a) Liens securing taxes, assessments or governmental charges or levies, or in connection with workers' compensation, unemployment insurance or social security obligations, or the claims or demands of materialmen, mechanics, carriers, warehousemen, landlords and other like Persons not yet delinquent or which are being contested in good faith by appropriate proceedings with adequate reserves set aside;
- (b) Attachment, judgment or other similar Liens arising in connection with court proceedings that do not, in the aggregate, materially detract from the value of their Property, materially impair the use thereof in the operation of their businesses, or materially impair their ability to perform the Obligations and (1) that are discharged or stayed within thirty (30) days of attachment or levy, or (2) payment of which is covered in full (subject to customary and reasonable deductibles) by surety bond;
- (c) Easements, rights of ways, restrictions and other similar charges or encumbrances on real Property that do not interfere with the orderly conduct of their businesses or materially detract from the value of the affected real Property;
- (d) Minor defects and irregularities in the title of real Properties existing on the Restated Closing Date that do not materially detract from the value or impair the use of such Properties for the purposes for which they are held;
- (e) Liens existing or arising by virtue of the leasing or rental of their Property to the extent leases and rentals are permitted by this Agreement, whether the same are capital leases or operating leases or rentals;
- (f) Indebtedness, liabilities, guaranties or Liens in favor of Bank under this Agreement, the Notes and the other Loan Documents;
 - (g) Indebtedness consisting of Subordinated Obligations;
- (h) Existing indebtedness, Liens and Rights of Others listed on Schedule 6.6;

- (i) Guaranties arising from endorsement, in the ordinary course of collection, of negotiable instruments;
- (j) Indebtedness incurred to vendors in the ordinary course of business;
- (k) Agreements to provide credit assurance for the obligations of customers of Borrower or any of its Subsidiaries that consist solely of an agreement to repurchase salable inventory of such customers purchased from Borrower or any of its Subsidiaries at a purchase price not in excess of the sales price paid therefor by such customers;
- (1) Purchase money Liens securing obligations incurred in connection with purchases or capitalized leases permitted hereby, provided that such Liens shall be limited to the item or items being so purchased or leased;
- (m) Guaranties of any of the Subsidiaries' obligations under operating leases, or, with the written consent of Bank, which consent will not be unreasonably withheld, guaranties of any other obligations of the Subsidiaries, or the guaranty by Borrower or any Subsidiary of any obligation owed to Bank;
- (n) Indebtedness incurred with respect to purchase money obligations for capital expenditures not to exceed an aggregate principal amount of \$3,000,000 at any time; and
- (o) The unsecured promissory notes, each dated as of December 6, 1994, executed by Borrower in favor of (i) J. Nelson Hoffman and Joan Hoffman as Co-Trustees of the Hoffman Family Revocable Trust UTA September 30, 1991 (the "Hoffman Family Trust") in an original principal amount of \$644,112, (ii) Bruce J. Greenbaum and Teri Greenbaum as Co-Trustees of the Greenbaum Family Trust in an original principal amount of \$555,888, (iii) J. Nelson Hoffman and Joan Hoffman as Co-Trustees of the Hoffman Family Trust in an original principal amount of \$97,593, and (iv) J. Greenbaum and Teri Greenbaum as Co-Trustees of the Greenbaum Family Trust in an original principal amount of \$84,225.
- 6.7 Transactions with Affiliates. Enter into any transaction of any kind with any Affiliate of Borrower other than (a) transactions between or among Borrower and its Subsidiaries, and (b) arms-length transactions with Affiliates which are permitted with non-Affiliates pursuant this Loan Agreement.

- 6.8 Change in Fiscal Year. Change its Fiscal Year.
- 6.9 Payment or Prepayment of Subordinated Obligations. Pay or prepay any principal (including sinking fund payments), interest or any other amount with respect to any Subordinated Obligation, or purchase or redeem any Subordinated Obligation, except Borrower may (a) pay interest in accordance with the terms of any Subordinated Obligation so long as there has not occurred and is then continuing any Default or Event of Default, (b) purchase or redeem its Subordination Obligations so long as there has not occurred and is then continuing any Default or Event of Default, provided, that, taking into account the purchase or redemption to be made, (i) the sum of (1) such purchase or redemption payments after January 1, 1996, plus (2) the total Teledyne Acquisition Price does not exceed \$12,000,000; and (ii) the Premium Payments made in connection with purchases, redemptions or conversions of Borrower's Subordinated Obligations do not exceed \$2,000,000 in the aggregate for any rolling twelve month period or \$.15 for each dollar of the Subordinated Obligations purchased, redeemed or converted, and provided further that, prior to the payment of any Premium Payments, Borrower has delivered to Bank a certificate of a Responsible Official of Borrower certifying (A) as to the amount of the Premium Payments being paid with respect to each dollar of the Subordinated Obligations being purchased or redeemed, (B) as to the aggregate amount of the Premium Payments being paid, and (C) that, taking into account the Premium Payments being made, no Default or Event of Default exists or will have occurred.
- 6.10 Intercompany Indebtedness. Incur or cause to exist any intercompany indebtedness between Borrower and any of its Subsidiaries or divisions, or between any such division or Subsidiary and any other division or Subsidiary, to be evidenced in any manner other than by a book entry in the financial records of the parties thereto.
- 6.11 Leverage Ratio. Permit the Leverage Ratio as of the last day of any fiscal quarter of Borrower and its Subsidiaries

ending during any period specified below to exceed the designated ratios for the periods specified below:

Period	Maximum Ratio	Numerator (1)(2) Denominator
l Closing Date September 29, 1996	1.20 : 1.0	40 / 34
er 30, 1996 December 30, 1996	1.15 : 1.0	40 / 35
31, 1996 December 30, 1997	1.0 : 1.0	36 / 36
31, 1997 and at all llowing thereafter	1.0 : 1.0	32 / 42

provided, however, to the extent that any Line B Loan is made hereunder at any time on or after the Restated Closing Date, the "Maximum Ratio" set forth above shall be (i) 2.25 to 1.00 for the fiscal quarter ending June 30, 1996 and (ii) subject to a one-time adjustment, effective as of the fiscal quarter ending September 30, 1996, as follows: the numerator of the Maximum Ratio shall be increased by the sum of the Teledyne Acquisition Price plus the total consideration paid by Borrower to purchase or redeem its Subordinated Obligations (up to a maximum aggregate increase of \$12,000,000 for the fiscal quarter ending September 30, 1996, and \$10,000,000 for all fiscal quarters ending thereafter), and the denominator shall be decreased by the sum of that portion of the Teledyne Acquisition Price allocated to goodwill plus the total consideration paid by Borrower to purchase or redeem its Subordinated Obligations (up to a maximum decrease of \$12,000,000 for all such fiscal quarters); provided, further, that in no event shall the Maximum Ratio be less than 1.00 to 1.00, nor shall the Maximum Ratio ever exceed the following amounts for the following periods:

Period	Maximum Ratio
Fiscal quarter ending	2.25:1.00
June 30, 1996	

⁽¹⁾ In each instance, in millions and rounded up to the nearest one tenth of one percent.

⁽²⁾ Numbers used for calculating applicable ratio.

Fiscal quarter ending September 30, 1996	2.10:1.00
Fiscal quarter ending December 31, 1996	1.75:1.00
Fiscal quarter ending December 31, 1997 and thereafter	1.30:1.00

For example, assuming that Bank makes a Line B Loan hereunder, and the Teledyne Purchase Price is \$4,000,000, with goodwill allocation of \$2,000,000, and Borrower pays \$6,000,000 to redeem its Subordinated Obligations, the Maximum Ratio for purposes of this Section 6.11 at September 30, 1996 will be set as follows:

6.12 Fixed Charge Coverage Ratio. Permit the Fixed Charge Coverage Ratio as at the end of any fiscal quarter for the twelve month period ending on the last day of such fiscal quarter to be less than the designated ratios for the periods specified below:

Period	Minimum Ratio
Restated Closing Date through March 30, 1997	1.15 : 1.00
March 31, 1997 through March 30, 1998	1.25 : 1.00
March 31, 1998 and at all times thereafter	1.30 : 1.00

- 6.13 Losses/Earnings. Permit consolidated Net Income to be less than zero for Borrower and its Subsidiaries for any fiscal quarter.
- 6.14 Capital Expenditures. Spend or incur obligations (including the total amount of any capital lease) to acquire fixed or capital assets in the aggregate for more than \$6,000,000 in the Fiscal Year ending December 31, 1996, and for more than \$4,000,000 in any Fiscal Year ending thereafter.
- $\,$ 6.15 Acquisitions. Borrower shall not, and shall not suffer or permit any of its Subsidiaries to, make any Acquisition.

ARTICLE 7. INFORMATION AND REPORTING REQUIREMENTS

- 7.1 Financial and Business Information. So long as any Loan remains unpaid, or any other Obligation remains unpaid or unperformed, or any portion of the Commitment remains outstanding, Borrower shall, unless Bank otherwise consents in writing, deliver to Bank, at Borrower's sole expense:
 - (a) As soon as practicable, and in any event within 30 days after the end of each fiscal month of Borrower, (i) consolidated and consolidating balance sheets of Borrower and its Subsidiaries as at the end of such month, setting forth in comparative form the corresponding figures as at the end of the corresponding month of their preceding fiscal year and (ii) consolidated and consolidating statements of profit and loss and of changes in financial position of Borrower and its Subsidiaries for such month and for the portion of their fiscal year ended with such month, setting forth in comparative form the corresponding figures for the corresponding periods of their preceding fiscal year, all in reasonable detail. The preceding financial statements shall be certified by a Responsible Official of Borrower as fairly presenting the financial condition, results of operations and changes in financial position of Borrower and its Subsidiaries in accordance with generally accepted accounting principles, consistently applied, as at such date and for such periods, subject only to normal year-end audit adjustments.
 - (b) As soon as practicable, and in any event within 45 days after the end of each fiscal quarter of Borrower (including the last fiscal quarter of each fiscal year, provided that with respect to such last quarter the financial statements required hereby may be in preliminary form, prior to year-end audit adjustments), (i) consolidated and consolidating balance sheets of Borrower and its Subsidiaries as at the end of such quarter, setting forth in comparative form the corresponding figures as at the end of the corresponding quarter of their preceding fiscal year and (ii) consolidated and consolidating statements of profit and loss and of changes in financial position of Borrower and its Subsidiaries for such quarter and for the portion of their fiscal year ended with such quarter, setting forth in comparative form the corresponding figures for the corresponding periods of their preceding fiscal year, all in reasonable detail. The preceding financial statements shall be certified by a Responsible Official of Borrower as fairly presenting the financial condition, results of operations and changes in financial

position of Borrower and its Subsidiaries in accordance with generally accepted accounting principles, consistently applied, as at such date and for such periods, subject only to normal year-end audit adjustments.

- (c) As soon as practicable, and in any event within 90 days after the close of each Fiscal Year of Borrower, (i) consolidated and consolidating balance sheets of Borrower and its Subsidiaries as at the end of such fiscal year, setting forth in comparative form the corresponding figures as at the end of their preceding fiscal year, and (ii) consolidated and consolidating statements of profit and loss and of changes in financial position of Borrower and its Subsidiaries for such fiscal year, setting forth in comparative form the corresponding figures for their previous fiscal year, all in reasonable detail. Such balance sheets and statements shall be prepared in accordance with generally accepted accounting principles, consistently applied, and such consolidated balance sheet and consolidated statements shall be accompanied by a report and opinion of independent public accountants of recognized standing selected by Borrower and reasonably satisfactory to Bank, which report and opinion shall be prepared in accordance with generally accepted auditing principles as at such date, and shall be subject only to such qualifications and exceptions as are acceptable to Bank.
- (d) Within 45 days after the end of each fiscal quarter of Borrower (or more frequently, if requested by the Bank), a consolidated and consolidating project backlog (with work in progress information on major projects, if requested by Bank) of Borrower and its Subsidiaries.
- (e) Promptly after request by Bank, copies of any detailed audit reports or recommendations submitted to Borrower or any of its Subsidiaries by independent accountants in connection with the accounts or books of Borrower or any of its Subsidiaries, or any audit of any of them.
- (f) Promptly after request by Bank, copies of any report or other document filed by Borrower or any of its Subsidiaries with any Governmental Agency.
- (g) Within ten days of the filing or sending thereof, copies of each annual report, proxy or financial statement or other report or communication sent to the shareholders of Borrower, and copies of all annual, regular, periodic and special reports and registration statements which Borrower may file or be required to file

with the Securities and Exchange Commission or any similar or corresponding Governmental Agency or with any securities exchange.

- (h) As soon as practicable, and in any event within 2 Banking Days after a corporate officer of Borrower becomes aware of the existence of any condition or event which constitutes a Default, a written notice specifying the nature and period of existence thereof and what action Borrower or its Subsidiaries are taking or propose to take with respect thereto.
- (i) Promptly upon a corporate officer of Borrower becoming aware that (i) a dispute exists between Borrower or any of its Subsidiaries and any Governmental Authority or law enforcement authority which, if determined adversely to Borrower or its Subsidiary, would have a material adverse effect on Borrower or its Subsidiary, together with written notice describing the pertinent facts relating thereto and what action Borrower or its Subsidiary is taking or proposes to take with respect thereto, (ii) any Person commenced a legal proceeding with respect to a claim against Borrower or any of its Subsidiaries in excess of \$500,000 that is not fully covered by insurance or (iii) any creditor or lessor under a written credit agreement or material lease has asserted a default thereunder on the part of Borrower or any of its Subsidiaries or (iv) any Person commenced a legal proceeding with respect to a claim against Borrower or any of its Subsidiaries under a contract that is not a credit agreement or material lease in excess of \$500,000 or which otherwise may reasonably be expected to result in a material adverse effect on Borrower and its Subsidiaries, taken as a whole, together with a written notice describing the pertinent facts relating thereto and what action Borrower or its Subsidiaries are taking or propose to take with respect thereto.
- (j) Updated projections for Borrower and its Subsidiaries (prepared on a consolidated and consolidating basis), as follows: (1) within 60 days prior to the end of each Fiscal Year, the strategic/long range plan on an annual basis for the Fiscal Years from present through December 31, 1999, and (2) within 30 days after the end of each Fiscal Year, the annual operating plan for the current Fiscal Year, on a monthly basis; such projections required by (1) and (2) above each to be in form and detail satisfactory to Bank, and each to be submitted together with the certification of Borrower's chief financial officer stating that the projections are based on facts known to Borrower and on assumptions that are reasonable and consistent with such facts, that no

material (in amount and likelihood) fact or assumption has been omitted as a basis for such projections which, in Borrower's reasonable business judgment, should be included, and that such projections are reasonably based on such facts and assumptions.

- (k) As soon as practicable, and in any event within 30 days after the end of each fiscal month of Borrower, a variance report comparing Borrower's actual performance for such month to Borrower's projected performance for such month as indicated in the projections delivered to Bank pursuant to Section 7.1(j).
- (1) Within 45 days of the Restated Closing Date, a consolidated and consolidating post-closing balance sheet of Borrower and its Subsidiaries, taking into effect the acquisition of the assets of Teledyne, together with a letter from the Borrower's Chief Financial Officer stating that he has reviewed such post-closing balance sheets and, to the best knowledge of the Chief Financial Officer, taking into account such accountant's review thereof, such post-closing balance sheets are in conformity with generally accepted accounting principles.
- (m) Within 30 days of the Restated Closing Date, annual pro-forma consolidating and consolidated financial statements for Borrower and its Subsidiaries for the Fiscal Years ending December 31, 1996 through December 31, 1999.
- (n) Promptly upon request, such other information, financial or otherwise, as Bank may reasonably request, including but not limited to, accounts receivable agings, a breakdown of the fixed assets of Borrower and its Subsidiaries and work in progress reports.
- 7.2 Compliance Certificates. So long as any Loan remains unpaid, or any other Obligation remains unpaid or unperformed, or any portion of the Commitments remains outstanding, Borrower shall, unless Bank otherwise consents in writing, deliver to Bank, at Borrower's sole expense, not later than 60 days after the end of each fiscal quarter of Borrower, a Certificate of a Responsible Official of Borrower in the form of Exhibit F (a) setting forth computations showing, in detail satisfactory to Bank, whether Borrower and its Subsidiaries were in compliance with their obligations pursuant to Sections 6.11 through 6.14; and (b) stating that a review of the activities of Borrower and its Subsidiaries during such fiscal period has been made under supervision of the certifying Responsible Official with a view to determining whether during such fiscal period Borrower and its Subsidiaries performed and observed all their respective Obligations under the Loan Documents, and

either (i) stating that, to the best knowledge of the certifying Responsible Official, during such fiscal period, Borrower and its Subsidiaries performed and observed each covenant and condition of the Loan Documents applicable to them, or (ii) if Borrower and its Subsidiaries have not performed and observed such covenants and conditions, specifying all such Defaults and their nature and status.

7.3 Revisions or Updates to Schedules. Should any of the information or disclosures provided on any of the Schedules originally attached hereto become outdated or incorrect in any material respect, Borrower promptly shall provide to Bank such revisions or updates to such Schedule(s) as may be necessary or appropriate to update or correct such Schedule(s); provided that no such revisions or updates to any Schedule(s) shall be deemed to have amended, modified or superseded such Schedule(s) as originally attached hereto, or to have cured any breach of warranty or representation resulting from the inaccuracy or incompleteness of any such Schedule(s), unless and until Bank, in its sole and absolute discretion, shall have accepted in writing such revisions or updates to such Schedule(s).

ARTICLE 8. CONDITIONS

- 8.1 Initial Loan, Etc. The obligation of Bank to make the initial Loan to be made by it and to issue the initial Standby Letter of Credit following the Restated Closing Date, each are subject to the following conditions precedent (in addition to any applicable conditions precedent set forth elsewhere in this Article 8), each of which shall be satisfied prior to or concurrently with the making of the initial Loan and the issuance of the initial Standby Letter of Credit following the Restated Closing Date (unless Bank, in its sole and absolute discretion, shall agree otherwise):
 - (a) Bank shall have received all of the following, each of which shall be originals unless otherwise specified, each properly executed by a Responsible Official of each party thereto, each dated as of the Restated Closing Date and each in form and substance satisfactory to the Bank and its legal counsel (unless otherwise specified or, in the case of the date of any of the following, unless Bank otherwise agrees or directs):
 - (1) executed counterparts of this Agreement, sufficient in number for distribution to Bank, Borrower and their respective counsel;
 - (2) the Notes executed by Borrower payable to the order of Bank, in a principal amount equal to the Line A Commitment and the Line B Commitment;
 - (3) with respect to Borrower and any and each Subsidiary of Borrower, such documentation as Bank may require to establish the due organization, valid existence and good standing of Borrower and each such Subsidiary, its qualification to engage in business in each jurisdiction in which it is engaged in business or required to be so qualified, its authority to execute, deliver and perform any Loan Documents to which it is a Party, and the identity, authority and capacity of each Responsible Official thereof authorized to act on its behalf, including, without limitation, certified copies of articles of incorporation and amendments thereto, bylaws and amendments thereto, certificates of good standing and/or qualification to engage in business, tax clearance certificates, certificates of corporate resolutions, incumbency certificates, Certificates of Responsible Officials, and the like;

- (4) the Reaffirmation of Borrower Security Agreement;
- (5) the Continuing Guaranty executed by all Subsidiaries of Borrower;
- (6) the Reaffirmation of Subsidiary Security Agreement executed by all Subsidiaries of Borrower;
- (7) such Collateral Documents (in addition to the Security Agreement) as Bank may require granting Liens on Property of Borrower and/or any of its Subsidiaries, together with such related financing statements or other documents as Bank may request to perfect, effect, facilitate, consent to, give notice of or otherwise evidence such Liens;
- (8) a Certificate of a Responsible Official of Borrower certifying that the conditions specified in Sections 8.1(c) and 8.1(d) have been satisfied;
- (9) evidence that all Liens or Rights of Others on or in the Property of Borrower and/or its Subsidiaries (other than such Liens and Rights of Others as are permitted by Section 6.6) have been terminated or discharged;
- (10) such other certificates, documents, consents or opinions as the Agent reasonably may require.
- (b) Duly executed financing statements with respect to the Collateral shall have been delivered to Bank for filing and/or recording with such Governmental Agencies, and in such jurisdictions and locales, as Bank may specify.
- (c) The representations and warranties of Borrower contained in Article 4 shall be true and correct.
- (d) Borrower and its Subsidiaries and any other Parties shall be in compliance with all the terms and provisions of the Loan Documents, and no Default shall have occurred and be continuing.
- (e) There shall have occurred no material adverse change in the condition of the Borrower or its Subsidiaries (financial or otherwise).
- (f) Bank's counsel shall have reviewed the Indenture Agreement executed to evidence the Subordinated Obligations, specifically the provisions relating to the $\,$

conversion/redemption of the Subordinated Obligations, and found same to be satisfactory.

- 8.2 Any Increasing Loan, Etc. In addition to any applicable conditions precedent set forth elsewhere in this Article 8, the obligation of Bank to make any Loan which would increase the principal amount outstanding under the Notes, or to issue any Standby Letter of Credit, are subject to the following conditions precedent:
 - (a) the representations and warranties contained in Article 4, other than Sections 4.4(a), 4.6, 4.11, 4.14 and 4.19, shall be true and correct on and as of the date of the Loan or issuance, as the case may be, as though made on and as of that date; there shall be no actions, suits or proceedings pending against or affecting Borrower or any of its Subsidiaries or any Property of any of them in any court of Law or before any Governmental Agency which might reasonably be expected adversely to affect materially the business, operations or condition (financial or otherwise) of Borrower and its Subsidiaries, taken as a whole; no material adverse change shall have occurred in the business, operations or condition (financial or otherwise) of Borrower and its Subsidiaries, taken as a whole, since the Restated Closing Date; no Default shall have occurred and be continuing; the Bank shall have timely received a properly completed Request for Loan or Request for Standby Letter of Credit, as the case may be, in compliance with all applicable provisions of Article 2; and Bank shall have received, dated as of the date of the Loan or issuance, as the case may be, a Certificate of a Responsible Official of Borrower to the effect that all of the above conditions have been satisfied, with any changes or exceptions thereto being described in a schedule attached to such certificate and with such changes or exceptions being subject to the approval of Bank; and
 - (b) Bank shall have received, in form and substance satisfactory to Bank such other certificates, documents, consents or opinions as Bank reasonably may require.
- 8.3 Any Loan. In addition to any applicable conditions precedent set forth elsewhere in this Article 8, the obligation of Bank to make any Loan is subject to the conditions precedent that the representations and warranties contained in Section 4.12 shall be true and correct in all material respects on and as of the date of such Loan as though made on and as of that date, and that there shall not have occurred any Default that is then continuing.

ARTICLE 9. EVENTS OF DEFAULT AND REMEDIES UPON EVENT OF DEFAULT

- 9.1 Events of Default. The existence or occurrence of any one or more of the following events, whatever the reason therefor, shall constitute an Event of Default:
 - (a) Borrower fails to pay any installment of principal when due, or Borrower fails to pay any installment of interest of any indebtedness on any of the Notes or any portion thereof, or to reimburse, or to pay any fee or any other amount due Bank under any Loan Document within five (5) days after the date when due; or
 - (b) Any failure to comply with Section 7.1(h) that is materially adverse to the interest of Bank; or
 - (c) Borrower, any of its Subsidiaries or any other Party fails to perform or observe any other term, covenant or agreement contained in any Loan Document on its part to be performed or observed; or
 - (d) Any representation or warranty made in any Loan Document or in any certificate, agreement, instrument or other document made or delivered by any Party pursuant to or in connection with any Loan Document proves to have been incorrect when made in any respect that is materially adverse to the interests of Bank; or
 - (e) Borrower or any of its Subsidiaries (i) fails to pay the principal, or any principal installment, of any present or future indebtedness for borrowed money or in connection with the purchase or lease of Property, or any guaranty of present or future indebtedness for borrowed money or issued in connection with the purchase or lease of Property, on its part to be paid, when due (or within any stated grace period), whether at the stated maturity, upon acceleration, by reason of required prepayment or otherwise, which equals or exceeds in the aggregate, \$500,000 or (ii) fails to perform or observe any other term, covenant or agreement on its part to be performed or observed in connection with any present or future indebtedness for borrowed money or in connection with the purchase or lease of Property, or of any guaranty of present or future indebtedness for borrowed money or issued in connection with the purchase or lease of Property, which equals or exceeds in the aggregate, \$500,000 if as a result of such failure any holder or holders thereof (or an agent or trustee on its or their behalf) has the right to declare such indebtedness due before the date on which it otherwise would become due, or

has commenced judicial or nonjudicial action to collect such indebtedness or to foreclose or otherwise realize upon security held therefor, or has taken or is taking such other actions as might materially adversely affect the Collateral, the interests of Bank under the Loan Documents or the ability of Borrower or its Subsidiaries to pay and perform their Obligations under the Loan Documents; or

- (f) Any Loan Document, at any time after its execution and delivery and for any reason other than the agreement of Bank or satisfaction in full of all the Obligations, ceases to be in full force and effect or is declared by a court of competent jurisdiction to be null and void, invalid or unenforceable in any respect which, in the reasonable opinion of Bank, is materially adverse to the interests of Bank; or any Party thereto denies that it has any or further liability or obligation under any Loan Document, or purports to revoke, terminate or rescind same; or
- (g) A final judgment against Borrower or any of its Subsidiaries is entered for the payment of money in excess of \$500,000 and such judgment remains unsatisfied without procurement of a stay of execution within thirty (30) calendar days after the date of entry of judgment, or in any event later than five (5) days prior to the date of any proposed sale thereunder; or
- (h) Borrower or any of its Subsidiaries is the subject of an order for relief in a bankruptcy case, or is unable or admits in writing its inability to pay its debts as they mature, or makes an assignment for the benefit of creditors; or applies for or consents to the appointment of any receiver, trustee, custodian, conservator, liquidator, rehabilitator or similar officer for it or for all or any part of its Property; or any receiver, trustee, custodian, conservator, liquidator, rehabilitator or similar officer is appointed without the application or consent of that Person and the appointment continues undischarged or unstayed for forty-five (45) calendar days; or institutes or consents to any bankruptcy, insolvency, reorganization, arrangement, readjustment of debt, dissolution, custodianship, conservatorship, liquidation, rehabilitation or similar case or proceedings relating to it or to all or any part of its Property under the Laws of any jurisdiction; or any similar case or proceeding is instituted without the consent of that Person and continues undismissed or unstayed for forty-five (45) calendar days; or any judgment, writ, warrant of attachment or execution or similar process is issued or levied against all or any material part of the Property of any such

Person and is not released, vacated or fully bonded within forty-five (45) calendar days after its issue or levy; or

- (i) Except as otherwise expressly permitted by any Loan Document or agreed to by Bank, any Lien on any Collateral created by any Loan Document, at any time after the execution and delivery of that Loan Document and for any reason other than satisfaction in full of all Obligations that is not attributable to any action taken or not taken by Bank, ceases or fails to constitute a valid, perfected and subsisting first priority Lien on the Collateral purported to be covered thereby; or
- (j) Borrower or any Subsidiary thereof is dissolved or liquidated or all or substantially all of the assets of Borrower or any Subsidiary of Borrower are sold or otherwise transferred in violation of the provisions of this Agreement without the written consent of Bank; or
- (k) An event or set of events occurs which has or could reasonably be expected to have a material adverse effect on (i) the validity or enforceability of this Agreement, any other agreement or document executed in connection with this Agreement, or the ability of the Bank to enforce its legal remedies thereunder; or (ii) the condition (financial or otherwise), business operations or prospects of Borrower and its Subsidiaries, taken as a whole, which could in turn reasonably be expected to impair the ability of Borrower and its Subsidiaries, taken as a whole, to perform their obligations under this Agreement and all such other documents and agreements; or
- (1) A default or defined Event of Default occurs under any other agreement Borrower or any Subsidiary has with Bank or any affiliate of Bank, and is not cured within any applicable cure period; or
- (m) The occurrence of any one or more of the following events with respect to Borrower or any Subsidiary, provided such event or events could reasonably be expected, in the judgment of Bank, to subject the Borrower to any tax, penalty or liability (or any combination of the foregoing) which, in the aggregate, could have a material adverse effect on the financial condition of Borrower with respect to a Plan:
 - (1) A reportable event shall occur with respect to a Plan which is, in the reasonable judgment of Bank likely to result in the termination of such Plan for purposes of Title IV of ERISA.

- (2) Any Plan termination (or commencement of proceedings to terminate a Plan) or Borrower's full or partial withdrawal from a Plan.
- 9.2 Remedies Upon Event of Default. Without limiting any other rights or remedies of Bank provided for elsewhere in this Agreement, or the Loan Documents, or by applicable Law, or in equity, or otherwise:
 - (a) Upon the occurrence of any Event of Default other than an Event of Default described in Section 9.1(h):
 - (1) the Commitment to make Loans and all other obligations of Bank and all rights of Borrower and any other Parties under the Loan Documents shall terminate without notice to or demand upon Borrower, which are expressly waived by Borrower, except that Bank may waive the Event of Default or, without waiving, determine, upon terms and conditions satisfactory to Bank, to make further Loans; and
 - (2) Bank may declare all or any part of the unpaid principal of the Notes, all interest accrued and unpaid thereon and all other amounts payable under the Loan Documents to be forthwith due and payable, whereupon the same shall become and be forthwith due and payable, without protest, presentment, notice of dishonor, demand or further notice of any kind, all of which are expressly waived by Borrower.

Despite the foregoing, if, in Bank's opinion, a breach of any provision of this Agreement (other than any provision specifically referred to in Section 9.1 or Article 6), is capable of being remedied, and Borrower submits a written request for a cure period along with its plan to remedy the default, then the breach will not be considered an Event of Default under this Agreement for a period of thirty (30) days after the date such breach first occurs; provided, however, that Bank will not be obligated to make further Loans to Borrower during that period.

- (b) Upon the occurrence of any Event of Default described in Section 9.1(h):
 - (1) the Commitment to make Loans and all other obligations of Bank and all rights of Borrower and any other Parties under the Loan Documents shall terminate without notice to or demand upon Borrower, which are expressly waived by Borrower, except that Bank may waive the Event of Default or, without waiving, determine, upon terms and conditions

satisfactory to Bank, to make further Loans, which determination shall apply equally to, and shall be binding upon, Bank; and

- (2) the unpaid principal of the Notes, all interest accrued and unpaid thereon and all other amounts payable under the Loan Documents shall be forthwith due and payable, without protest, presentment, notice of dishonor, demand or further notice of any kind, all of which are expressly waived by Borrower.
- (c) Upon the occurrence of any Event of Default, Bank without notice to or demand upon Borrower, which are expressly waived by Borrower, may proceed to protect, exercise and enforce their rights and remedies under the Loan Documents against Borrower and such other rights and remedies as are provided by Law or equity.
- (d) The order and manner in which the Bank's rights and remedies are to be exercised shall be determined by Bank in its sole discretion, and all payments received by Bank shall be applied first to the costs and expenses (including attorneys' fees and disbursements) of Bank, and thereafter to the Obligations in such order as Bank may determine. For the purpose of computing Borrower's Obligations hereunder, under the Notes, and the Standby Letters of Credit, payments shall be applied, first, to the costs and expenses of Bank, as set forth above, second, to the payment of accrued and unpaid interest due under any Loan Documents to and including the date of such application (ratably, and without duplication, according to the accrued and unpaid interest due under each of the Loan Documents), third, to the ratable payment of all unpaid principal amounts due under any Loan Documents (including, for the purposes hereof, principal due under the Notes, and reimbursement due for payments made under Standby Letters of Credit) and fourth, to the payment of all other amounts (including fees) then owing to Bank under the Loan Documents. No application of payments will cure any Event of Default, or prevent acceleration, or continued acceleration, of amounts payable under the Loan Documents, or prevent the exercise, or continued exercise, of rights or remedies of Bank hereunder or thereunder or at Law or in equity. The term "attorneys' fees" as used in this Agreement shall include the allocated costs of in-house counsel.
- (e) Upon the occurrence of any event that would be an Event of Default under Section 9.1(h) with the passage of time, Bank may take such action as Bank deems necessary to protect the interests of Bank under the Loan Documents.

ARTICLE 10. MISCELLANEOUS

- 10.1 Cumulative Remedies; No Waiver. The rights, powers, privileges and remedies of Bank provided herein or in the Note or any other Loan Document are cumulative and not exclusive of any right, power, privilege or remedy provided by Law or equity. No failure or delay on the part of Bank in exercising any right, power, privilege or remedy may be, or may be deemed to be, a waiver thereof; nor may any single or partial exercise of any right, power, privilege or remedy preclude any other or further exercise of the same or any other right, power, privilege or remedy. The terms and conditions of Article 8 hereof are inserted for the sole benefit of Bank and Bank may waive them in whole or in part, with or without terms or conditions, in respect of any Loan or Standby Letter of Credit, without prejudicing the Bank's rights to assert them in whole or in part in respect of any other Loan or Standby Letter of Credit.
- 10.2 Amendments; Consents. No amendment, modification, supplement, extension, termination or waiver of any provision of this Agreement or any other Loan Document, no approval or consent thereunder, and no consent to any departure by Borrower or any other Party therefrom, may in any event be effective unless in writing signed by Bank (and, in the case of amendments, modifications or supplements of or to any Loan Document to which Borrower is a Party, the approval in writing of Borrower), and then only in the specific instance and for the specific purpose given.
- 10.3 Costs, Expenses and Taxes. Borrower shall pay on demand the reasonable costs and expenses of Bank in connection with the negotiation, preparation, execution and delivery of the Loan Documents, and of Bank in connection with the amendment, waiver, refinancing, restructuring, reorganization (including a bankruptcy reorganization) and enforcement or attempted enforcement of the Loan Documents, and any matter related thereto, including, without limitation, filing fees, recording fees, title insurance fees, appraisal fees, search fees and other out-of-pocket expenses and the reasonable fees and out-of-pocket expenses of any legal counsel (including, without limitation, costs and fees of Bank's in-house counsel), independent public accountants and other outside experts retained by Bank, and including, without limitation, any costs, expenses or fees incurred or suffered by Bank in connection with or during the course of any bankruptcy or insolvency proceedings of Borrower or any Subsidiary thereof. Borrower shall pay any and all documentary and other taxes (other than income or gross receipts taxes generally applicable to banks) and all costs, expenses, fees and charges payable or determined

to be payable in connection with the filing or recording of this Agreement, any other Loan Document or any other instrument or writing to be delivered hereunder or thereunder, or in connection with any transaction pursuant hereto or thereto, and shall reimburse, hold harmless and indemnify Bank from and against any and all loss, liability or legal or other expense with respect to or resulting from any delay in paying or failure to pay any tax, cost, expense, fee or charge or that any of them may suffer or incur by reason of the failure of any Party to perform any of its Obligations. Any amount payable to Bank under this Section 10.3 shall bear interest from the second Banking Day following the date of demand for payment at the rate provided for in Section 3.6.

- 10.4 Survival of Representations and Warranties. All representations and warranties contained herein or in any other Loan Document, or in any certificate or other writing delivered by or on behalf of any one or more of the Parties to any Loan Document, will survive the making and repayment of the Loans hereunder and the execution and delivery of the Note, and have been or will be relied upon by Bank, notwithstanding any investigation made by Bank or on its behalf.
- 10.5 Notices. Except as otherwise expressly provided in the Loan Documents: (a) All notices, requests, demands, directions and other communications provided for hereunder or under any other Loan Document must be in writing and must be mailed, telegraphed, telecopied, delivered or sent by telex or cable to the appropriate party at the address set forth on the signature pages of this Agreement or other applicable Loan Document or, as to any party to any Loan Document, at any other address as may be designated by it in a written notice sent to all other parties to such Loan Document in accordance with this Section 10.5; and (b) Any notice, request, demand, direction or other communication given by telegram, telecopier, telex or cable must be confirmed within 48 hours by letter mailed or delivered to the appropriate party at its respective address. Except as otherwise expressly provided in any Loan Document, if any notice, request, demand, direction or other communication required or permitted by any Loan Document is given by mail it will be effective on the earlier of receipt or the third calendar day after deposit in the United States mail with first class or airmail postage prepaid; if given by telegraph or cable, when delivered to the telegraph company with charges prepaid; if given by telex or telecopier, when sent; or if given by personal delivery, when delivered.
- 10.6 Execution of Loan Documents. Unless Bank otherwise specifies with respect to any Loan Document, this Agreement and any other Loan Document may be executed in any number of counterparts and any party hereto or thereto may execute any counterpart, each of which when executed and delivered will be

deemed to be an original and all of which counterparts of this Agreement or any other Loan Document, as the case may be, when taken together will be deemed to be but one and the same instrument. The execution of this Agreement or any other Loan Document by any party hereto or thereto will not become effective until counterparts hereof or thereof, as the case may be, have been executed by all the parties hereto or thereto.

10.7 Binding Effect; Assignment. This Agreement and the other Loan Documents shall be binding upon and shall inure to the benefit of the parties hereto and thereto and their respective successors and assigns, except that Borrower and/or its Affiliates may not assign their rights hereunder or thereunder or any interest herein or therein without the prior written consent of Bank. Bank may, at any time and from time to time, sell, assign, pledge or transfer participations in its rights hereunder or any interest herein provided that any actual or potential participants, assignees, pledge holders or transferees shall agree to treat all financial information exchanged as confidential, and provided further that Borrower has given its prior consent with respect to any proposed participant, or assignee. If a participation is sold or the loan is assigned, the purchaser will have the right of set-off against Borrower.

10.8 Lien on Deposits and Property in Possession of any Bank. As security for the prompt payment and performance of all Obligations, Borrower hereby grants to Bank a Lien on and a security interest in all its right, title, and interest in and to any and all deposit accounts now or hereafter maintained with Bank and in and to any and all of its Property and the proceeds thereof now or hereafter in the possession of Bank. If an Event of Default has occurred and is continuing, Bank may exercise its rights under Article 9 of the Uniform Commercial Code and other applicable Laws and apply any funds in any deposit account maintained with it by Borrower and/or any Property of Borrower in its possession against any Obligation owed to it by Borrower hereunder and/or under any other Loan Document.

10.9 Indemnity by Borrower. Borrower agrees to indemnify, save and hold harmless Bank and its directors, officers, agents, attorneys and employees (collectively the "Indemnitees") from and against: (a) Any and all claims, demands, actions or causes of action that are asserted against any Indemnitee by any person (other than Bank) if the claim, demand, action or cause of action directly or indirectly relates to a claim, demand, action or cause of action that such person asserts or may assert against Borrower or any Affiliate of Borrower or any officer, director or shareholder of Borrower relating to the Commitments, the use or contemplated use of proceeds of any Loan or Standby Letter of Credit, or the

relationship of Borrower and Bank under this Agreement; (b) Any and all claims, demands, actions or causes of action if the claim, demand, action or cause of action arises out of or relates to any action or inaction by Borrower or its Subsidiaries in connection with the Commitments or the use or contemplated use of proceeds of any Loan or Standby Letter of Credit; (c) Any administrative or investigative proceeding by any Governmental Agency arising out of or related to a claim, demand, action or cause of action described in clauses (a) or (b) above; and (d) Any and all liabilities, losses, costs or expenses (including attorneys' fees (including the allocated cost of in-house counsel) and disbursements and other professional services) that any Indemnitee suffers or incurs as a result of the assertion of any foregoing claim, demand, action or cause of action; provided that no Indemnitee shall be entitled to indemnification of or any loss caused by its own gross negligence or willful misconduct. Promptly after receipt by Bank of notice of any third-party claim to which the indemnification obligations hereunder would apply, Bank shall give notice thereof to Borrower in writing, stating the information then available to Bank regarding the amount and nature of such claim, but the failure to do so promptly notify Borrower shall not affect Borrower's obligations under this paragraph 10.9 unless such failure materially prejudices Borrower's right to participate in the contest of such claim, demand, action or cause of action. Borrower shall have the right to defend against all such claims, demands, actions or causes of action, at Borrower's cost and expense, with counsel selected by Borrower and reasonably satisfactory to Bank. The Borrower shall regularly keep Bank informed as to the status of, and any material developments relating to, any claim subject to this paragraph 10.9, and shall respond promptly in writing to any reasonable requests by Bank for information or other data concerning such claim. Any obligation or liability of Borrower to any Indemnitee under this Section 10.9 shall be and hereby is covered and secured by the Loan Documents and the Collateral, and shall survive the expiration or termination of this Agreement and the repayment of all Loans and the payment and performance of all other Obligations owed to Bank.

10.10 Environmental Indemnity And Other Matters.

(a) Borrower agrees that it shall indemnify, defend and hold Indemnitees, and each of them, harmless from and against any claim, action, suit, proceeding, loss, cost, damage, liability, deficiency, fine penalty, punitive damage or expense (including, without limitation, attorneys' fees and disbursements), resulting from, arising out of, or based upon (i) the presence, release, use, generation, discharge, storage or disposal of any hazardous or toxic wastes or materials on, under, in or about, or the transportation of any such wastes to or

from, any Property of Borrower, or (ii) the violation, or alleged violation, of any statute, ordinance, order, rule, regulation, permit, judgment or license relating to the use, generation, release, discharge, storage, disposal or transportation of hazardous or toxic wastes or materials on, under, in or about, to or from, any such Property. This indemnity shall include, without limitation, any damage, liability, fine, penalty, punitive damage, cost or expense (including without limitation all post-foreclosure clean-up and removal costs and expenses) arising from or out of any claim, action, suit or proceeding for personal injury (including sickness, disease or death), tangible or intangible property damage, compensation for lost wages, business income, profits, or other economic loss, damage to the natural resources or the environment, nuisance, pollution, contamination, leak, spill, release or other adverse effect on the environment. The obligations of Borrower and the rights of Indemnitees under this Section are in addition to and not in substitution of the obligations of Borrower and rights of Indemnitees under the laws and regulations cited above and any other similar applicable laws. The obligations and indebtedness of Borrower, and the rights of Indemnitees, under this Section and the laws and regulations cited above, notwithstanding anything contained herein or in any other document or agreement which may be construed to the contrary, (i) shall not be secured by the Collateral Documents, and shall not be subject to California Code of Civil Procedure Sections 580a, 580d, 726 or any other anti-deficiency laws, and (ii) shall survive the foreclosure of any deed of trust, the repayment of the Loans, and the termination of the Notes and other Loan Documents. As used herein, the terms "toxic" or "hazardous" wastes, substances or materials shall include, without limitation, all those so designated and all those in any way regulated by any of the above-cited laws or regulations, or any other present or future environmental or other similar laws or regulations. Whenever practicable, upon obtaining actual knowledge of any event that would entitle Bank to be indemnified under this Section, Bank shall endeavor to provide notice to Borrower of such fact and Bank shall cooperate with Borrower to endeavor to minimize the liabilities for which Bank is entitled to be indemnified.

(b) Upon demand by Bank, Borrower will defend any investigation, action or proceeding alleging the presence of any hazardous substance in any such location, which affects the real property owned by the Borrower or any of its Subsidiaries or which is brought or commenced against Bank, whether alone or together with Borrower or any other person, all at Borrower's own cost and by counsel to be

approved by Bank in the exercise of its reasonable judgment. In the alternative, Bank may elect to conduct its own defense at the expense of Borrower.

- (c) Bank and its agents and representatives will have the right at any reasonable time to enter and visit the properties of Borrower and its Subsidiaries and any other place where any property is located for the purposes of observing the real property owned by Borrower or any of its Subsidiaries, taking and removing soil or groundwater samples, and conducting tests on any part of the real property owned by Borrower or any of its Subsidiaries. Bank is under no duty, however, to visit or observe the real property owned by Borrower or any of its Subsidiaries or to conduct tests, and any such acts by Bank will be solely for the purposes of preserving Bank's rights under this Agreement and the other Loan Documents. No site visit, observation or testing by Bank will result in a waiver of any default of Borrower or impose any liability on Bank. In no event will any site visit, observation or testing by Bank be a representation that hazardous substances are or are not present in, on or under the real property owned by Borrower or any of its Subsidiaries, or that there has been or will be compliance with any law, regulation or ordinance pertaining to hazardous substances or any other applicable governmental law. Neither Borrower nor any other party is entitled to rely on any site visit, observation or testing by Bank. Bank owes no duty of care to protect Borrower or any other party against, or to inform Borrower or any other party of, any hazardous substances or any other adverse condition affecting the real property owned by Borrower or any of its Subsidiaries. Bank will not be obligated to disclose to Borrower or any other party any report or findings made as a result of, or in connection with, any site visit, observation or testing by Bank. In each instance, Bank will give Borrower reasonable notice before entering the real property owned by Borrower or any of its Subsidiaries or any other place Bank is permitted to enter under this paragraph. Bank will make reasonable efforts to avoid interfering with the use of the real property owned by Borrower or any of its Subsidiaries or any other property in exercising any rights provided in this paragraph.
- (d) Borrower's obligations to Bank under this Section, except the obligation to give notices to Bank, shall survive termination of this Agreement and repayment of Borrower's obligations to Bank under this Agreement, and shall also survive as unsecured obligations after any acquisition by Bank of the collateral securing this Agreement.

- 10.11 Nonliability of the Bank. Borrower acknowledges and agrees that:
- (a) Any inspections of Collateral made by or through Bank are for purposes of administration of the Loan only and Borrower is not entitled to rely upon the same;
- (b) By accepting or approving anything required to be observed, performed, fulfilled or given to Bank pursuant to the Loan Documents, including any certificate, financial statement, insurance policy or other document, Bank shall not be deemed to have warranted or represented the sufficiency, legality, effectiveness or legal effect of the same, or of any term, provision or condition thereof, and such acceptance or approval thereof shall not constitute a warranty or representation to anyone with respect thereto by Bank;
- (c) The relationship between Borrower and Bank is, and shall at all times remain, solely that of a borrower and lender; Bank shall not under any circumstance be construed to be a partner or joint venturer of Borrower or its Affiliates; Bank shall not under any circumstance be deemed to be in a relationship of confidence or trust or a fiduciary relationship with Borrower or its Affiliates, or to owe any fiduciary duty to Borrower or its Affiliates; Bank undertakes or assumes no responsibility or duty to Borrower or its Affiliates to select, review, inspect, supervise, pass judgment upon or inform Borrower or its Affiliates of any matter in connection with their Property, any Collateral held by Bank or the operations of Borrower or its Affiliates; Borrower and its Affiliates shall rely entirely upon their own judgment with respect to such matters; and any review, inspection, supervision, exercise of judgment or supply of information undertaken or assumed by Bank in connection with such matters is solely for the protection of Bank and neither Borrower nor any other Person is entitled to rely thereon; and
- (d) Bank shall not be responsible or liable to any Person for any loss, damage, liability or claim of any kind relating to injury or death to Persons or damage to Property caused by the actions, inaction or negligence of Borrower and/or its Affiliates and Borrower hereby indemnifies and holds Bank harmless from any such loss, damage, liability or claim.
- 10.12 No Third Parties Benefited. This Agreement is made for the purpose of defining and setting forth certain obligations, rights and duties of Borrower and Bank in connection with the Loans, and is made for the sole protection of Borrower and Bank, and Bank's successors and assigns. Except as

provided in Section 10.9, no other Person shall have any rights of any nature hereunder or by reason hereof.

- 10.13 Confidentiality. Bank agrees to hold any confidential information that it may receive from Borrower pursuant to this Agreement in confidence, except for disclosure: (a) To legal counsel, accountants and other professional advisors to Borrower or Bank; (b) To regulatory officials having jurisdiction over that Bank; (c) As required by Law or legal process or in connection with any legal proceeding to which Bank is a party; (d) To another financial institution in connection with a disposition or proposed disposition of all or part of Bank's interests hereunder; and (e) To prospective purchasers of Collateral in connection with any disposition thereof; provided that nothing in this Section shall be construed to create or give rise to any fiduciary duty on the part of Bank to Borrower or to create or give rise to any relationship of confidence or trust between Bank and Borrower.
- 10.14 Further Assurances. Borrower and its Subsidiaries shall, at their expense and without expense to the Bank, do, execute and deliver such further acts and documents as any Bank from time to time reasonably requires for the assuring and confirming unto Bank of the rights hereby created or intended now or hereafter so to be, or for carrying out the intention or facilitating the performance of the terms of any Loan Document, or for assuring the validity, perfection, priority or enforce- ability of any Lien under any Loan Document.
- 10.15 Integration. This Agreement, together with the other Loan Documents, comprises the complete and integrated agreement of the parties on the subject matter hereof and supersedes all prior agreements, written or oral, on the subject matter hereof. In the event of any conflict between the provisions of this Agreement and those of any other Loan Document, the provisions of this Agreement shall control and govern; provided that the inclusion of supplemental rights or remedies in favor of Bank in any other Loan Document shall not be deemed a conflict with this Agreement. Each Loan Document was drafted with the joint participation of the respective parties thereto and shall be construed neither against nor in favor of any party, but rather in accordance with the fair meaning thereof.
- 10.16 Governing Law. Except to the extent otherwise provided therein, each Loan Document shall be governed by, and construed and enforced in accordance with, the local Laws of California; provided that the local Laws of California shall not apply with respect to any foreclosure of real Property Collateral located outside California, and in no event shall California Code of Civil Procedure Sections 726 and/or 580a and/or 580b and/or 580d apply to any such foreclosure outside

of California or to the right of Bank to obtain a deficiency judgment for all Obligations remaining due following such foreclosure.

- 10.17 Severability of Provisions. Any provision in any Loan Document that is held to be inoperative, unenforceable or invalid as to any party or in any jurisdiction shall, as to that party or jurisdiction, be inoperative, unenforceable or invalid without affecting the remaining provisions or the operation, enforceability or validity of that provision as to any other party or in any other jurisdiction, and to this end the provisions of all Loan Documents are declared to be severable.
- 10.18 Headings. Article and Section headings in this Agreement and the other Loan Documents are included for convenience of reference only and are not part of this Agreement or the other Loan Documents for any other purpose.
- 10.19 Time of the Essence. Time is of the essence of the Loan Documents.
 - 10.20 Arbitration.
 - (a) This paragraph concerns the resolution of any controversies or claims between Borrower and Bank, including but not limited to those that arise from:
 - (i) This Agreement (including any renewals, extensions or modifications of this Agreement);
 - (ii) Any document, agreement or procedure related to or delivered in connection with this Agreement;
 - (iii) Any violation of this Agreement; or
 - (iv) Any claims for damages resulting from any business conducted between Borrower and Bank, including claims for injury to persons, property or business interests (torts).
 - (b) At the request of Borrower or Bank, any such controversies or claims will be settled by arbitration in accordance with the United States Arbitration Act. The United States Arbitration Act will apply even though this Agreement provides that it is governed by California law.
 - (c) Arbitration proceedings will be administered by the American Arbitration Association and will be subject to its commercial rules of arbitration.

- (d) For purposes of the application of the statute of limitations, the filing of an arbitration pursuant to this paragraph is the equivalent of the filing of a lawsuit, and any claim or controversy which may be arbitrated under this paragraph is subject to any applicable statute of limitations. The arbitrators will have the authority to decide whether any such claim or controversy is barred by the statute of limitations and, if so, to dismiss the arbitration on that basis.
- (e) If there is a dispute as to whether an issue is arbitrable, the arbitrators will have the authority to resolve any such dispute.
- (f) The decision that results from an arbitration proceeding may be submitted to any authorized court of law to be confirmed and enforced.
- (g) The procedure described above will not apply if the controversy or claim, at the time of the proposed submission to arbitration, arises from or relates to an obligation to Bank secured by real property located in California. In this case, both Borrower and Bank must consent to submission of the claim or controversy to arbitration. If both parties do not consent to arbitration, the controversy or claim will be settled as follows:
 - (i) Borrower and Bank will designate a referee (or a panel of referees) selected under the auspices of the American Arbitration Association in the same manner as arbitrators are selected in Association-sponsored proceedings;
 - (ii) The designated referee (or the panel of referees) will be appointed by a court as provided in California Code of Civil Procedure Section 638 and the following related sections;
 - (iii) The referee (or the presiding referee of the panel) will be an active attorney or a retired judge; and
 - (iv) The award that results from the decision of the referee (or the panel) will be entered as a judgment in the court that appointed the referee, in accordance with the provisions of California Code of Civil Procedure Sections 644 and 645.
- (h) This provision does not limit the right of Borrower or Bank to:
 - (i) exercise self-help remedies such as setoff;

- (ii) foreclose against or sell any real or personal property collateral; or
- (iii) act in a court of law, before, during or after the arbitration proceeding to obtain:
 - (A) an interim remedy; and/or
 - (B) additional or supplementary remedies.
- (i) The pursuit of or a successful action for interim, additional or supplementary remedies, or the filing of a court action, does not constitute a waiver of the right of Borrower or Bank, including the suing party, to submit the controversy or claim to arbitration if the other party contests the lawsuit. However, if the controversy or claim arises from or relates to an obligation to Bank which is secured by real property located in California at the time of the proposed submission to arbitration, this right is limited according to the provision above requiring the consent of both Borrower and Bank to seek resolution through arbitration.
- (j) If Bank forecloses against any real property securing this Agreement, Bank has the option to exercise the power of sale under the deed of trust or mortgage, or to proceed by judicial foreclosure.
- 10.21 Personal Liability of Responsible Officials. In the absence of actual fraud, no Certificate of a Responsible Official delivered to the Bank hereunder shall create personal liability to the Bank on the part of such Responsible Official for payment or performance of any of the Obligations or otherwise under the Loan Documents.
- 10.22 Prior Agreement Superseded. This Agreement supersedes the Third Amended and Restated Loan Agreement, and any credit outstanding thereunder shall be deemed to be outstanding under this Agreement.
- 10.23 Teledyne Indemnity. To the extent permitted by the Teledyne Acquisition Agreement, Borrower and Jay-El Products, Inc. each hereby assign to Bank the indemnity provisions (and all rights and benefits attendant thereto) granted to either of them under the Teledyne Acquisition Agreement; provided, however, that Borrower and Jay-El Products, Inc. shall be entitled to enforce all of such rights and to collect any proceeds in connection therewith so long as no Default has occurred and is continuing under this Agreement. Borrower agrees to use its best efforts, and to cause Jay-El Products,

87

Inc. to use its best efforts, to permit an assignment of such indemnity provisions to Bank.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first above written.

BANK OF AMERICA NATIONAL TRUST AND SAVINGS ASSOCIATION

DUCOMMUN INCORPORATED

By /s/ J. Thomas Fagan

J. Thomas Fagan Vice President By /s/ Joseph C. Berenato

Joseph C. Berenato President, Chief Operating Officer and Chief Financial Officer

By /s/ James S. Heiser

James S. Heiser Vice President,

Vice President, Treasurer and Secretary

Address where notices to the Bank are to be sent:

South Bay Commercial Banking Office, #1418 150 Long Beach Blvd. Third Floor Long Beach, CA 90802 Attention: J. Thomas Fagan Telecopy No.: (310) 624-4360 IBOR Lending Office: Address where notices to the Borrower are to be sent:

23301 S. Wilmington Avenue Carson, CA 90745 Attention: James S. Heiser Telecopy No.: (310) 518-0176

>INITIAL HERE>

/s/

/s/ JSH

>INITIAL HERE>

> Acknowledgment of Prepayment Fees. Borrower acknowledges that prepayment of any portion of any Loan bearing interest at an IBOR Rate may result in Bank incurring additional costs, expenses and/or liabilities. Borrower therefore agrees to pay the prepayment fee described in Section 3.5 of this Agreement if the portion is prepaid, whether voluntarily or by reason of acceleration, including acceleration upon any sale of the Collateral. Borrower agrees that the fee represents a reasonable estimate of the prepayment costs, expenses and/or

liabilities of Bank. Borrower further acknowledges that Bank's willingness to offer an IBOR Rate to Borrower is sufficient and independent consideration for this agreement to pay the fee.

CONSENT OF SIGNIFICANT SUBSIDIARIES

In order to induce Bank to enter into the foregoing Loan Agreement, each of the undersigned acknowledge that they have read the Loan Agreement and hereby consent to it, and, with respect to Jay-El Products, Inc., joins in the assignment granted pursuant to Section 10.23 of the Loan Agreement.

Dated as of May 16, 1996.

AHF-DUCOMMUN INCORPORATED	BRICE MANUFACTURING COMPANY, INC.
By: /s/ Robert P. Harmen	By: /s/ Joseph C. Berenato
Its: President	Its: Vice President
By: /s/	By: /s/
Its: Assistant Secretary	Its: Secretary
AEROCHEM, INC.	3DBM, INC.
By: /s/	By: /s/ Joseph C. Berenato
Its: Controller	Its: Vice President
By: /s/	By: /s/
Its: President	Its: Secretary
JAY-EL PRODUCTS, INC.	
By: /s/	
Its: President	
By: /s/ M. L. Strirff	
Its: Controller	

FIRST AMENDMENT TO FOURTH AMENDED AND RESTATED LOAN AGREEMENT

This FIRST AMENDMENT TO FOURTH AMENDED AND RESTATED LOAN AGREEMENT ("First Amendment"), dated as of June ____, 1996, is made by Ducommun Incorporated, a Delaware corporation (the "Borrower"), and Bank of America National Trust and Savings Association (the "Bank"), with reference to the following facts:

RECITALS

A. This First Amendment amends that certain Fourth Amended and Restated Loan Agreement dated as of May 16, 1996, by and between the Borrower and the Bank (the "Loan Agreement"). Capitalized terms used herein and not otherwise defined shall have the meanings set forth for such terms in the Loan Agreement.

B. Borrower is acquiring substantially all of the assets of MechTronics of Arizona, Inc., and transferring such assets to DUC Acquisition Corp., a newly formed subsidiary of Borrower. Borrower and Bank desire to amend the Loan Agreement as set forth herein to reflect such asset acquisition.

AGREEMENTS

NOW, THEREFORE, in consideration of the mutual covenants and benefits contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are acknowledged, Borrower and Bank hereby agree as follows:

1. AMENDMENTS TO LOAN AGREEMENT

1.1 Section 1.1. Section 1.1 of the Loan Agreement is

amended as follows:

a. The definition of the term "Fixed Charge Coverage Ratio" is amended as follows:

(i) clauses (iv) and (v) o subsection (a) thereof are revised in their entirety to read as follows:

"(iv) taxes paid in cash in that fiscal period; minus (v) non-financed capital expenditures made during that fiscal period (which, for any fiscal period ending on or before December 31, 1996, shall be all capital expenditures for that fiscal year in excess of \$3,000,000); plus"

(ii) clause (x) is revised to insert the phrase ", the MechTronics Acquisition" immediately following the phrase "the acquisition of Brice" therein;

(iii) subsection (b) thereof is amended and restated in its entirety to read as follows:

"(b) to:

(i) interest charges payable in cash for that fiscal period; plus (ii) scheduled principal payments with respect to indebtedness for that fiscal period; plus (iii) scheduled payments for that fiscal period in respect of capital leases; plus (iv) scheduled non-compete payments for that fiscal period due under the Brice Agreement; plus (v) without duplication of the amounts in clause (ii), \$750,000 for each fiscal quarter in Fiscal Year 1996, \$1,125,000 for each fiscal quarter in Fiscal Year 1998, and \$1,100,000 for each fiscal quarter in Fiscal Year 1999."

and (iv) to delete that portion thereof following subsection (b).

- b. The definition of "IBOR Rate Spread" is amended to delete the reference to the rate "2.25%" in the second line thereof immediately after the clause "with respect to any Line B Loan" and insert in lieu thereof the rate "1.75%".
- c. The definition of the term "Line A Commitment" is amended to delete the amount "\$14,000,000" in the first line thereof and to insert in lieu thereof the amount of "\$16,000,000".
- d. The definition of the term "Line B Commitment" is amended to delete the amount "\$10,000,000" in the first line thereof and to insert in lieu thereof the amount of "\$8,000,000".
- e. The definition of the term "Line B Termination Date" is amended to delete the date "August 31, 1996" and to insert in lieu thereof "July 31, 1996".
- f. The definition of the term "Reference Rate Spread" is amended to delete the rate "0.75%" in the second line thereof and insert in lieu thereof the rate "0.25%".

g. The definition of the terms "Cash Flow", "Excess Cash Flow", "Teledyne", "Teledyne Acquisition", "Teledyne Acquisition Agreement", "Teledyne Acquisition Price" and "Working Capital" are deleted in their entirety.

h. The following defined terms are added to Section 1.1 of the Loan Agreement in the appropriate alphabetical order:

> > 'MechTronics' means MechTronics of Arizona, Inc.

'MechTronics Acquisition' means the purchase by Borrower of substantially all of the assets and assumption of certain liabilities of MechTronics.

'MechTronics Purchase Agreement' means the Asset Purchase and Sale Agreement dated as of June 21, 1996 by and among Borrower, MechTronics, Michael J. Demuro and The Michael Demuro and Geraldine Demuro Family Trust u/a/d July 18, 1979, as amended.

'MechTronics Acquisition Price' means the total value of the consideration paid by Borrower and its Subsidiaries in connection with the MechTronics Acquisition."

1.2 Section 2.7. Section 2.7 of the Loan Agreement is amended to restate subsections (a) and (b) thereof in their entirety to read as follows:

"(a) The Line A Commitment shall automatically and permanently reduce on each December 31 in the amount set forth below opposite the year in which such date occurs:

	Annual Reduction To
Year	Line A Commitment
1996	\$ 3,000,000
1997	\$ 5,000,000

(b) [DELETED.]"

1.3 Section 3.2. Section 3.2 of the Loan Agreement is amended and restated in its entirety to read as follows:

- (a) Borrower shall pay to Bank a nonrefundable Line B facility fee in an aggregate amount equal to \$25,000 with respect to the Line B Commitment. Bank acknowledges that it has received such fee. On or before June 28, 1996, Borrower will pay to Bank an additional non-refundable Line B facility fee in the amount of \$25,000 with respect to the change in the Line B Commitment attributable to the MechTronics Acquisition.
- (b) On or before June 28, 1996, Borrower shall pay to Bank a nonrefundable Line A facility fee in an amount equal to 0.25% of the Line A Commitment (which amount is equal to \$40,555), less any portion of such fee previously paid to Bank (which amount is equal to \$35,485). Thereafter, on each July 1, commencing July 1, 1997, Borrower shall pay to Bank an annual Line A facility fee in an amount equal to 0.25% of the Line A Commitment on such date. Such facility fee shall be fully earned as of the date when due and shall be non-refundable under any circumstances."
- 1.4 Section 4.5. Section 4.5 of the Loan Agreement is amended by deleting the reference to the date "March 31, 1996" therein and inserting in lieu thereof the date "May 27, 1996".
- 1.5 Section 4.23. Section 4.23 of the Loan Agreement is amended by deleting the reference therein to "Section 7.1(g)" and inserting in lieu thereof "Section 7.1(j)".
- ${\hbox{1.6 Section 4.24. Section 4.24 of the Loan Agreement}} \\$
- 1.7 Section 5.9. Subsection (b) of Section 5.9 of the Loan Agreement is amended and restated in its entirety to read as follows:
 - "(b) Use the proceeds of Line B Loans for the following purpose only: to consummate the MechTronics Acquisition."
 - 1.8 Section 5.12. Section 5.12 of the Loan

Agreement is amended and restated in its entirety to read as follows:

"5.12 Bank Audit. Permit Bank, upon Bank's reasonable request, either directly or through Persons retained by Bank, to conduct a thorough annual bank audit of the accounts receivable and inventories of Borrower and

its Subsidiaries and to have access to all of the books, records and employees of Borrower and its Subsidiaries reasonably incidental to such bank audit and pay the reasonable expenses of Bank or such Persons incurred in connection with such bank audit."

1.9 Section 5.13. Section 5.13 of the Loan Agreement is amended and restated in its entirety to read as follows:

"5.13 Notice of Location Change/Defaults Under Material Contracts.

- (a) Promptly notify Bank in writing of the occurrence of any change in the location of, or the addition of, any branch office, any field office, any warehouse or any other place of business of Borrower or any Subsidiary if such change requires any action by Bank in order to maintain the validity or to perfect any Lien in favor of Bank or results in or is accompanied by the acquisition or ownership of any property other than Property not subject to a Lien in favor of the Bank pursuant to the Collateral Documents; provided, however, that no such notification shall be required if the change, together with any other change since the Restated Closing Date, would not in the aggregate involve Property with a book value or fair market value, whichever is higher, in excess of \$200,000; and
- (b) Promptly notify Bank in writing of the occurrence of any material default or event of default under any customer contract of Borrower or any of its Subsidiaries with a value of \$5,000,000 or more."
- 1.10 Section 5.15. Section 5.15 of the Loan Agreement is amended and restated in its entirety to read as follows:

"5.15 MechTronics Phase I. On or before August 1, 1996, Borrower shall deliver to Bank copies of the most current existing written Phase I environmental audit, or any such audit required by Bank, on the property commonly known as 1601 East Broadway Road, Phoenix, Arizona, the contents of which shall be satisfactory to the Bank."

1.11 Section 6.6. Section 6.6 of the Loan Agreement is amended as follows:

- a. Subsection 6.6(n) is amended by deleting the word "and" at the end thereof;
- b. Subsection 6.6(o) is amended by adding the word "and" at the end thereof;
- c. A new subsection (p) is added thereafter to read as follows:
 - "(p) Indebtedness evidenced by that certain promissory note in the principal amount of \$750,000 made by Borrower in favor of MechTronics, Michael J. Demuro and The Michael Demuro and Geraldine Demuro Family Trust u/a/d July 18, 1979, as amended."
- 1.12 Section 6.9. Section 6.9 of the Loan Agreement is amended and restated in its entirety to read as follows:

"6.9 Payment or Prepayment of Subordinated Obligations. Pay or prepay any principal (including sinking fund payments), interest or any other amount with respect to any Subordinated Obligation, or purchase or redeem any Subordinated Obligation, except Borrower may (a) pay interest in accordance with the terms of any Subordinated Obligation so long as there has not occurred and is then continuing any Default or Event of Default, (b) purchase or redeem its Subordinated Obligations so long as there has not occurred and is then continuing any Default or Event of Default, provided, that, taking into account the purchase or redemption to be made, (i) the total of such purchase or redemption payments (exclusive of any Premium Payments) after January 1, 1996 does not exceed \$3,000,000; and (ii) the Premium Payments made in connection with purchases, redemptions or conversions of Borrower's Subordinated Obligations do not exceed \$2,000,000 in the aggregate for any rolling twelve month period or \$.15for each dollar of the Subordinated Obligations purchased, redeemed or converted, and provided further that, prior to the payment of any Premium Payments, Borrower has delivered to Bank a certificate of a Responsible Official of Borrower certifying (A) as to the amount of the Premium Payments being paid with respect to each dollar of the Subordinated Obligations being purchased or redeemed, (B) as to the aggregate amount of the Premium Payments being paid, and (C) that, taking into account the Premium Payments being made, no Default or Event of Default exists or will have occurred."

1.13 Section 6.11. Section 6.11 of the Loan Agreement is amended and restated in its entirety to read as follows:

"6.11 Leverage Ratio. Permit the Leverage Ratio as of the last day of any fiscal quarter of Borrower and its Subsidiaries ending during any period specified below to exceed the designated ratios for the periods specified below:

Period 	Maximum Ratio	Numerator(1)(2) Denominator
Restated Closing Date through September 29, 1996	1.20 : 1.00	44 / 34.5
September 30, 1996 through December 30, 1996	1.15 : 1.00	43 / 35.5
December 31, 1996 through December 30, 1997	1.00 : 1.00	38 / 38.5
December 31, 1997 and at all times following thereafter	1.00 : 1.00	36 / 45.5

provided, however, to the extent that any Line B Loan is made hereunder at any time on or after the Restated Closing Date, the "Maximum Ratio" set forth above shall be (i) 1.80 to 1.00 for the fiscal quarter ending June 30, 1996 and (ii) subject to a one-time adjustment, effective as of the fiscal quarter ending September 30, 1996, as follows: the numerator of the Maximum Ratio shall be increased by the sum of the MechTronics Acquisition Price plus the consideration (not to exceed \$3,000,000) paid by Borrower to purchase or redeem its Subordinated Obligations (up to a maximum aggregate increase of \$11,000,000 for the fiscal quarters as stated above), and the denominator shall be decreased by the sum of that portion of the MechTronics Acquisition Price allocated to goodwill plus the total consideration paid by Borrower to purchase or redeem its Subordinated Obligations (up to a maximum decrease of \$5,500,000 for the fiscal quarters as stated above); provided, further, that in no event shall the Maximum Ratio be less than 1.00 to 1.00, nor shall the Maximum Ratio ever exceed the following amounts for the following periods:

Period	Maximum Ratio

- (1) In each instance, in millions and rounded up to the nearest one tenth of one percent.
- (2) Numbers used for calculating applicable ratio.

Fiscal quarter ending June 30, 1996

1.90:1.00"

Fiscal quarter ending

September 30, 1996 1.90:1.00

Fiscal quarter ending

December 31, 1996 1.60:1.00

Fiscal quarter ending December 31, 1997 and

thereafter 1.20:1.00

For example, assuming that Bank makes a Line B Loan hereunder, and the MechTronics Purchase Price is \$8,000,000, with goodwill allocation of \$2,500,000, and Borrower pays \$3,000,000 to redeem its Subordinated Obligations, the Maximum Ratio for purposes of this Section 6.11 at September 30, 1996 will be set as follows:

1.14 Section 6.12. Section 6.12 of the Loan Agreement is amended by deleting the Minimum Ratio of "1.15:1.00" for the period from June 30, 1996 through March 30, 1997 and inserting in lieu thereof the Minimum Ratio of "1.20:1.00".

1.15 Section 6.14. Section 6.14 of the Loan Agreement is amended and restated in its entirety to read as follows:

"6.14 Capital Expenditures. Spend or incur obligations (including the total amount of any capital lease) to acquire fixed or capital assets in the aggregate for more than \$7,000,000 in the Fiscal Year ending December 31, 1996, and for more than \$6,000,000 in any Fiscal Year ending thereafter."

1.16 Section 7.1. Subsections (1) and (m) of Section 7.1 of the Loan Agreement are amended to delete subsection (m) in its entirety and to amend and restate subsection (l) in its entirety to read as follows:

"(1) Within 60 days after the consummation of the MechTronics Acquisition, a consolidated and consolidating post-closing balance sheet of Borrower and its Subsidiaries, taking into effect the acquisition of the assets and the assumption of the liabilities of MechTronics being acquired or assumed

pursuant to the MechTronics Acquisition Agreement. The preceding financial statements shall be certified by the Chief Financial Officer of Borrower as fairly presenting the financial condition and results of changes in financial position of Borrower and its Subsidiaries in accordance with generally accepted accounting principles, consistently applied, as at such date, subject only to normal year-end audit adjustments."

1.17 Section 10.23. Section 10.23 of the Loan Agreement is amended and restated in its entirety to read as follows:

"10.23 MechTronics Indemnity. To the extent permitted by the MechTronics Acquisition Agreement, Borrower and DUC Acquisition Corp., each hereby assign to Bank the indemnity provisions and representations and warranties (and all rights and benefits attendant thereto) granted or made to either of them under the MechTronics Acquisition Agreement; provided, however, that Borrower and DUC Acquisition Corp. shall be entitled to enforce all of such rights and to collect any proceeds in connection therewith so long as no Default has occurred and is continuing under this Agreement. Borrower agrees to use its best efforts, and to cause DUC Acquisition Corp. to use its best efforts, to permit the foregoing assignment to Bank."

1.18 Section 10.24. A new Section 10.24 is added to the Loan Agreement which shall read as follows:

"10.24 Bulk Sales. The Borrower and DUC Acquisition Corp. each hereby assign to the Bank the representations and warranties and indemnities (and all rights and benefits attendant thereto) granted to them under Article IV of the MechTronics Purchase Agreement, provided, however, that the Borrower and DUC Acquisition Corp. shall be entitled to enforce all of their rights with respect to Article IV of the MechTronics Purchase Agreement and to collect any proceeds in connection therewith so long as no Default has occurred and is continuing under the Loan Agreement. The Borrower further agrees that it shall cause the Bank (and its successors and assigns) to be indemnified for any loss or damage caused by the failure of the parties to the MechTronics Purchase Agreement to comply with the Bulk Transfer provisions of the Arizona Commercial Code to the extent that the Bank (or its successors or assigns) suffers such loss

or damage as a result of its ownership of the assets of the Borrower or any of its Subsidiaries."

- 1.19 Schedule 4.4. Schedule 4.4 to the Loan Agreement is amended and restated in its entirety to read as set forth in the new Schedule 4.4 attached hereto and incorporated by this reference.
- 1.20 Schedule 4.7. Schedule 4.7 to the Loan Agreement is amended and restated in its entirety to read as set forth in the new Schedule 4.7 attached hereto and incorporated by this reference.

2. REPRESENTATIONS AND WARRANTIES.

Borrower makes the following representations and warranties to Bank as of the date hereof, which representations and warranties shall survive the execution, termination or expiration of this First Amendment and shall continue in full force and effect until the full and final satisfaction and discharge of all Obligations of Borrower to Bank under the Loan Agreement and the other Loan Documents:

- 2.1 Reaffirmation of Prior Representations and Warranties. Borrower hereby reaffirms and restates as of the date hereof all of the representations and warranties made by Borrower in the Loan Agreement and the other Loan Documents, except to the extent such representations and warranties specifically relate to an earlier date.
- 2.2 No Default. No Default or Event of Default, breach or failure of condition has occurred and is continuing under any of the Loan Documents.
- 2.3 Due Execution. The execution, delivery and performance of this First Amendment and any instruments, documents or agreements executed in connection herewith (collectively, the "First Amendment Documents") are within the powers of Borrower and its Subsidiaries that are a party thereto, have been duly authorized by all necessary action, and do not contravene any law or the certificate of incorporation or bylaws of Borrower or any such Subsidiary, result in a breach of, or constitute a default under, any contractual restriction, indenture, trust agreement or other instrument or agreement binding upon Borrower or any such Subsidiary.
- 2.4 No Further Consent. The execution, delivery and performance of this First Amendment and each of the other First Amendment Documents do not require any consent or approval not previously obtained of any stockholder, beneficiary or creditor of Borrower or any of its Subsidiaries.

- 2.5 Binding Agreement. This First Amendment and each of the other First Amendment Documents constitute the legal, valid and binding obligation of Borrower or its Subsidiaries as are party thereto and are enforceable against Borrower and any such Subsidiary in accordance with their terms, except as such enforceability may be limited by bankruptcy, insolvency, moratorium, reorganization or similar laws or equitable principles relating to or limiting creditors' rights generally.
- 2.6 MechTronics Acquisition. The MechTronics Acquisition has been, or will be, consummated in accordance with all applicable state and federal laws.
- 2.7 DUC Acquisition Corp. Without limiting the foregoing, the representations and warranties set forth in Section 4.4 of the Loan Agreement are true and correct as of the date hereof as each applies to DUC Acquisition Corp. and the outstanding shares of capital stock of DUC Acquisition Corp.

CONDITIONS PRECEDENT.

The effectiveness of this First Amendment is subject to the satisfaction of each of the following conditions precedent:

- 3.1 Documentation. Borrower shall have delivered or caused to be delivered to Bank, at Borrower's sole cost and expense, the following, each of which shall be in form and substance satisfactory to Bank:
 - a. two counterpart executed originals of this First Amendment;
 - b. an executed original Instrument of Joinder from DUC Acquisition Corp. with respect to the Guaranty;
 - c. an executed original Second Amended and Restated Security Agreement (Subsidiaries);
 - d. an executed original Second Amendment and Reaffirmation of Security Agreement (Borrower) executed by Borrower, together with the stock certificate(s) and a stock power signed in blank, covering all the issued and outstanding shares of DUC Acquisition Corp.;
 - e. an Amended and Restated Borrower and Subsidiary Patent and Trademark Security Agreement;

- f. two executed original UCC-1 Financing Statements for debtor DUC Acquisition Corp. of Arizona, Inc. (for filing with the Arizona Secretary of State and the County Recorder of Maricopa County, Arizona);
- g. two executed original UCC-1 Financing Statements for debtor MechTronics of Arizona, Inc. (for filing with the Arizona Secretary of State and the County Recorder of Maricopa County, Arizona);
- $\ensuremath{\text{h.}}$ an original executed Consent and Reaffirmation of Guarantors;
- i. an original Landlord's Consent and Waiver executed by the Landlord of the Property;
- j. evidence that the execution, delivery and performance by the Borrower (and any guarantor) of this First Amendment and any instrument or agreement required under this First Amendment have been duly authorized;
- k. a completed form of the Bank's Environmental Questionnaire & Disclosure Statement (the "EQDS Form") for MechTronics; and
- l. an executed original UCC-2 Financing Statement termination of that certain UCC-1 Financing Statement #820806 filed by Bank One on February 23, 1995 with the Arizona Secretary of State.
- 3.2 MechTronics Acquisition Agreement. Bank and Bank's counsel shall have reviewed the MechTronics Purchase Agreement, and such document shall have been deemed to be acceptable to the Bank and its counsel.
- 3.3 EQDS. Bank's Environmental Services Group ("ESG") shall have received and reviewed the Bank's Environmental Questionnaire & Disclosure Form for the MechTronics facility located at 1601 East Broadway Road, Phoenix, Arizona (the "Arizona Property"), and deemed the information therein to be acceptable to the Bank.
- 3.4 Phase I. ESG shall have received from a consultant acceptable to Bank a verbal report of the results of the most current existing Phase I Environmental Report on the Arizona Property or any such report required by the Bank and deemed the information therein to be acceptable to the Bank.
- $\,$ 3.5 Leases. Bank's counsel shall have reviewed the lease between MechTronics and DUC Acquisition Corp. relating to the Arizona Property and deemed such lease to be acceptable to the Bank.

3.6 Material Contracts. Bank shall have reviewed a list of all of MechTronic's material contracts and deemed such contracts to be acceptable to the Bank.

- 3.7 No Defaults. All of Borrower's representations and warranties contained herein shall be true and correct on and as of the date of execution hereof and no Default or Event of Default shall have occurred and be continuing under any of the Loan Documents, as modified hereby.
- 3.8 Good Standing Certificate. Bank shall have received a Certificate of Good Standing issued by the Arizona Secretary of State confirming the good standing of MechTronics and DUC Acquisition Corp.
- 3.9 No Liens. Bank shall have received evidence that the assets of MechTronics are free and clear of all Liens and Rights of Others.
- 3.10 No Adverse Change. There shall have occurred no material adverse change in the condition of the Borrower or its Subsidiaries (financial or otherwise).
- 3.11 Fees. Borrower shall have paid the facility fee required pursuant to Section 4.2 of this First Amendment.

4. MISCELLANEOUS.

- 4.1 Costs and Expenses. Borrower agrees to pay all costs, expenses, attorneys' fees, search fees, filing and recordation fees and all other charges and expenses incurred by the Bank in connection with (1) the negotiation, preparation, delivery and execution of this First Amendment and the First Amendment Documents, including without limitation, the Bank's (i) attorneys' fees and costs (including allocated costs of in-house counsel) and (ii) out-of-pocket audit and appraisal costs (including allocated costs of in-house auditors), and (2) carrying out the terms of this First Amendment and the First Amendment Documents, whether incurred before or after the effective date hereof.
- 4.2 Facility Fee. Borrower shall pay to Bank a non-refundable Line B facility fee in the amount of \$25,000.00, together with an additional \$5,070 Line A facility fee, all of which shall be earned in full upon the execution of this First Amendment.
- 4.3 Effect of Amendment. Except as provided in this First Amendment, all of the terms and conditions of the Loan Agreement shall remain in full force and effect.

4.4 Counterparts. This First Amendment may be executed in counterparts and any party may execute any counterpart, each of which shall be deemed to be an original and all of which, taken together, shall be deemed to be one and the same document. The execution hereby by any party shall not become effective until this First Amendment is executed by all parties hereto.

4.5 Prior Agreements. This First Amendment contains the entire agreement between Bank and Borrower with respect to the subject matters hereof, and all prior negotiations, understandings and agreements with respect thereto are superseded by this First Amendment.

 $\,$ $\,$ $\,$ This First Amendment is executed as of the date stated at the beginning of this First Amendment.

BANK OF AMERICA NATIONAL TRUST AND SAVINGS ASSOCIATION

By /s/ J. Thomas Fagan, Vice President

J. Thomas Fagan, Vice President

[Printed Name and Title]

DUCOMMUN INCORPORATED

By /s/ Joseph C. Berenato, President

Joseph C. Berenato, President

[Printed Name and Title]

```
6-MOS
       DEC-31-1996
          JAN-01-1996
            JUN-29-1996
                           104
                       0
                 19,135
                   246
21,285
              47,165
                        56,314
               30,980
97,119
        25,077
                            0
              0
                        0
                          73
                        52
 97,119
                       52,661
             52,661
                         35,038
                 47,043
0
                   0
               697
               4,883
1,367
           3,516
                   0
                    0
                   3,516
0.55
                   0.51
```